

TERMS & CONDITIONS

PART A - DEFINITIONS AND INTERPRETATION

1. Definitions

In this Agreement, unless the context requires otherwise, the following terms and expressions shall have the meanings set out below:

"Account" includes the Settlement Account, the Custodian Account, the Commodity Investment Account, the Digital Token Account and any current, savings, deposit, securities, investment or other account opened and maintained with us, DBS Private Bank, whether singly, in joint name or in trust.

"Account Application" means the DBS Private Bank Account Application.

"Account Statement" means any written document (including facsimile or other electronic means from which it is possible to produce a hard copy) including periodic records and statement of accounts (a) confirming and setting out any and all Investment and Transaction entered into by you or on your behalf and/or (b) recording any other activity in relation to your Accounts, and containing such information as we consider appropriate.

"Account Terms" means the General Terms and Conditions Governing Accounts and Services, set out in Part B of the Terms and Conditions.

"Affiliate" means, in relation to us or the Bank, (a) any entity controlled, directly or indirectly by the Bank, (b) any entity that controls, directly or indirectly the Bank or (c) any entity, directly or indirectly, under common control with the Bank; and "control" of any entity or person means ownership of a majority of the voting power of the entity or person.

"Agent" means any agent, correspondent, broker, dealer, counterparty, adviser, bank, attorney, custodian, sub-custodian, Depository, Depository Agent, manager, assayer, refiner, service provider or Nominee selected or used by us, whether in Singapore or elsewhere, and may include any of our Affiliates.

"Aggrieved Parties" has the meaning given to such term in Clause 2.8(a) of the Gold Bullion Terms.

"Agreement" means the Private Banking and Investment Services Agreement, comprising the Account Application and the Terms and Conditions.

"Allocated Basis" means, in relation to any Gold Bullion held by us for and on your behalf, the Gold Bullion which has been segregated and details of which we notify in writing to you, and which shall include any Gold Bullion of equivalent quality, value, fineness, and weight (but not necessarily equivalent shape, dimensions or marks), which we may substitute from time to time in our absolute discretion and notify you in writing.

"Alternate Currency" means, in relation to a Currency Linked Investment, the currency (other than the Base Currency) in which the principal amount could be repaid.

"Applicable Agreement" means any document, application, agreement, contract, security document or terms and conditions relating to any Account, Service, Investment or Transaction (including any advice, Transaction Record, term sheet and/or any additional terms and conditions as may be prescribed by us from time to time but excluding, for the avoidance of doubt, this Agreement) as may be signed by you (whether in writing or by electronic means as prescribed by the Bank), or binding on you or agreed between us from time to time.

"Applicable Laws" means all relevant laws (including statutory enactments and judgments of any legal court or tribunal, and any form of unwritten or uncodified laws) of Singapore or any other applicable jurisdictions, rules, regulations, orders, rulings, directives, notices, circulars, decrees, bye-laws, policy statements, guidelines, practice notes, interpretations, standards, consultation papers, requirements, custom, usage or general commercial and regular banking practices (whether or not having the force of law) of any governmental or regulatory authority or agency, self-regulatory organisation, market, Exchange, clearing house, trade repository and electronic trading platform or depository system (whether in Singapore or elsewhere) having supervisory jurisdiction or relevance to this Agreement, any Applicable Agreement, Account, Service, Investment or Transaction. Such Applicable Laws may be amended from time to time notwithstanding any summary or description set out herein.

DBS Bank Ltd., Singapore 1 Co. Reg. No. 196800306E

"Assaying Agreement" has the meaning given to such term in Clause 5.1(b) of the Gold Bullion Terms.

"Assaying Documentation" has the meaning given to such term in Clause 5.2(a) of the Gold Bullion Terms.

"Assets" means Cash, Investments, and any of your assets or property as may be delivered and transferred by you to us or to our order whether by way of security or for management, safe custody or any purpose whatsoever. In cases where you are trustees opening and maintaining an Account for the purposes of a trust, as expressly made known to and acknowledged by us, references to Assets shall be construed to include references to the Assets of such trust, notwithstanding that beneficial ownership of those Assets may vest in persons other than the trustees.

"Authorised Agent" means a person authorised by you under the Account Application, any power of attorney or other letter, document or instrument, to give Instructions with respect to the operation of any of your Accounts, the use of any Service and/or any Investment or Transaction, in such form and substance as may be acceptable to us, which has been validly executed by you, received by us and updated in our records and in respect of whom we have not received from you any written notice of revocation or termination of such person's appointment, powers or authority.

"Bank" or "we" or "us" or "our" means, unless the context suggests otherwise, DBS Bank Ltd., Singapore, including DBS Private Bank, its successors and assignees.

"Bank Account" mean any Account, Branch Account and other account that may be opened by you with the Bank

"Banking Act" means the Banking Act 1970 of Singapore.

"Base Currency" means, in relation to a Currency Linked Investment, the currency in which the principal amount is placed.

"Blockchain" means the private blockchain managed and operated by the DBS Group.

"Branch" means any branch of DBS Bank Ltd., Singapore wherever located.

"Branch Account" means any current, savings, deposit, securities, custodian, investment or other account opened by you at any Branch, whether singly, in joint name or in trust, but excluding, for the avoidance of doubt, any Account.

"Beneficiary Institution" has the meaning given to such term in Clause 24.3 of the Account Terms.

"Business Day" means any day on which banks in Singapore are opened for business, but excluding Saturdays, Sundays and any gazetted public holidays.

"Cash" means all monies standing from time to time to the credit of your Bank Accounts, including cash accounts and Deposits, and all rights, benefits and proceeds of such Bank Accounts and any renewal or redesignation thereof.

"China Connect Terms" means the Specific Terms and Conditions Governing Transactions on the China Stock Connect, set out in Part I of the Terms and Conditions.

"CLI Fixing Date" means, in relation to a Currency Linked Investment, the date on which we determine whether to pay the principal amount and interest accrued thereon in the Base Currency or the Alternate Currency.

"CLI Fixing Time" means, in relation to a Currency Linked Investment, the time on the CLI Fixing Date at which the prevailing exchange rate between the Base Currency and the Alternate Currency will be used to determine whether to pay the principal amount and interest accrued thereon in the Base Currency or the Alternate Currency.

"CLI Maturity Date" means, in relation to a Currency Linked Investment, the date of maturity of such Currency Linked Investment, as agreed between you and us and if such day does not fall on a Business Day, the CLI Maturity Date shall be the first Business Day immediately following such day unless we determine otherwise.

- "CLI Value Date" means, in relation to a Currency Linked Investment, the date from which interest on a Currency Linked Investment accrues.
- "Commodity" means gold, silver or any other precious metal or commodity which we may, in our absolute discretion, from time to time stipulate as a Commodity for the purpose of any Commodity Investment.
- "Commodity Investment" means any Transaction in relation to Commodities effected through your Commodity Investment Account that we may, at your request, enter into with you, including any sale or purchase of Commodities, but excluding, for the avoidance of doubt, any Gold Bullion Transaction.
- "Commodity Investment Account" has the meaning given to such term in Clause 7.1 of the Product Terms.
- "Communication" means any notice, demand or other communication that may be sent by us to you relating to your Investments, Transactions, Accounts and/or any of the Service provided to you under this Agreement and/or any Applicable Agreement including Account Statements and Transaction Records.
- "Contracts (Rights of Third Parties) Act" means the Contracts (Rights of Third Parties) Act 2001 of Singapore.
- "Corresponding Purchase Transaction" means, in relation to the Gold Trading Services, a corresponding transaction between us and our counterparty whereby we agree to purchase Gold Bullion to sell the same to you pursuant to that Purchase Confirmation.
- "Corresponding Sale Transaction" means, in relation to the Gold Trading Services, a corresponding transaction between us and our counterparty whereby we agree to sell Gold Bullion to that counterparty, of which is to be purchased from you pursuant to that Sale Confirmation.
- "Credit Facilities" means any overdraft, credit, loan, guarantee, trading or other facility and accommodation in its widest sense which we may extend to you from time to time.
- "Currency Linked Investment" means a currency linked investment in which a placement of funds is accepted in one currency, the Base Currency, and which may be repayable in another currency, the Alternate Currency.
- "Custodian Account" has the meaning given to such term in Clause 3 of the Custody Terms.
- "Custody Terms" means the Specific Terms and Conditions Governing Custodial and Nominee Services, set out in Part D of the Terms and Conditions.
- "Customer Data" means all information (including personal data as defined in the Personal Data Protection Act) relating to you (and where applicable, your directors, partners, Authorised Agents, shareholders and beneficial owners), your Investments, Transactions, Accounts, the Services utilised by you and your account or dealing relationship with us, including Customer Information as defined in the Banking Act.
- "DBS Group" means DBS Group Holdings Ltd. and its subsidiaries.
- "DBS Privacy Policy" means such data policies, notices and/or other communications issued by us from time to time in relation to personal data provided by you or otherwise collected by us.
- "DBS Private Bank" includes DBS Private Bank and/or DBS Treasures Private Client (as the case may be).
- "Deposit" means any time deposit, fixed deposit or any fixed investment instrument in which a specific sum or sums of money is deposited with us or the Bank for a specified period of time at an agreed rate of interest but excluding Structured Deposits and any other deposit of Gold Bullion under the Gold Bullion Terms.
- "Deposit Date" means, in relation to a Deposit, the date agreed between you and us as being the first Business Day on which such Deposit shall commence.
- "Deposit Insurance and Policy Owners' Protection Schemes Act" means the Deposit Insurance and Policy Owners' Protection Schemes Act 2011 of Singapore.

"Deposit Maturity Date" means, in relation to a Deposit, the final day of the Deposit Period and if such day does not fall on a Business Day, the Deposit Maturity Date shall be the first Business Day immediately following such day unless we determine otherwise.

"Deposit Period" means, in relation to a Deposit, the period commencing on the Deposit Date and ending on the Deposit Maturity Date (both dates inclusive).

"Depository" means any central depository, depository, clearing house or clearing system in relation to any Asset or any Depository Agent, agent, depository, sub-delegate, share registrar or other institution or body authorised by any relevant Exchange in accordance with its constitution, by-laws, rules and regulations and all Applicable Laws.

"Depository Agent" means such entity identified as a depository agent of any Depository and as may be nominated by us from time to time.

"Derivative Transaction" has the meaning given to such term in the ISDA Master Agreement.

"Designated Facility" has the meaning given to such term in Clause 3.2(b) of the Gold Bullion Terms.

"Digital Exchange" means DBS Digital Exchange Pte. Ltd.

"Digital Payment Tokens" refers to "digital payment tokens" as defined in the Payment Services Act.

"Digital Token" includes Digital Payment Tokens and Security Tokens.

"Digital Token Account" has the meaning given to such term in Clause 3.1 of the Digital Token Terms.

"Digital Token Custody Services" means the custody of Digital Tokens for and on your behalf, pursuant to the Digital Token Terms.

"Digital Token Order" means any offer to enter into a Digital Token Transaction, or any request, application, order or instruction (in whatever form and howsoever sent, given or transmitted) sent, given or transmitted to the Bank by you or which the Bank or an officer of the Bank reasonably believes to be your offer, request, application, order or instruction and includes any request or instruction to revoke, ignore or vary any previous request or instruction.

"Digital Token Services" means the Digital Token Trading Services and the Digital Token Custody Services.

"Digital Token Settlement Date" has the meaning given to such term in Clause 2.3(b)(iii) of the Digital Token Terms.

"Digital Token Terms" means the Specific Terms and Conditions Governing Transactions in Digital Tokens, set out in Part J of the Terms and Conditions.

"Digital Token Trading Services" means the purchase and sale of Digital Tokens for and on your behalf, pursuant to the Digital Token Terms.

"Digital Token Transaction" means any transactions as we may from time to time permit you to carry out involving the trading of Digital Tokens, pursuant to the Digital Token Terms.

"Early Termination Date" has the meaning given to such term in Clause 4.7(b) of the Gold Bullion Terms.

"Error Trade" has the meaning given to such term in Clause 1.8(a)(v) of the Digital Token Terms.

"Event of Default" means any one of the events set out in Clause 25 of the Account Terms.

"Event of Force Majeure" includes any one or more of the following:

(a) any act of God;

- (b) any act of any sovereign;
- (c) any act of foreign law, judgment, order, decree, embargo, blockade, labour dispute;
- (d) acts, restrictions, rules, directions, regulations, edicts, mandates, by-laws, governmental or regulatory orders, refusals to grant any licenses or permissions, changes in policy or prohibitions or measures of any kind on the part of any government or regulatory authority;
- (e) interruptions, calamity, war, invasion, acts of a foreign enemy, hostilities, terrorism, natural disasters, aircraft or aerial objects, fire, explosions, floods, lightning or other adverse weather conditions, strikes, industrial actions, civil war or strife, rebellion, revolution, insurrection, or failure of utility service;
- (f) breakdown or failure of transmission or communication of data or computer facilities;
- (g) actual, probable or suspected occurrence of disease, illness, epidemic, pandemic;
- (h) sabotage;
- (i) a material adverse change in the monetary, political, financial (including conditions in any financial market) or economic conditions or exchange or capital controls or other restrictions, whether in Singapore or elsewhere; and
- (j) any other matter or cause beyond our reasonable control.

"*Exchange*" means any recognised exchange or quotation system on which Securities are traded and shall include the Digital Exchange.

"Extraordinary Event" includes the following:

- (a) any event which we, in good faith, believe to have a material adverse effect on any Investment, Transaction, Account and/or Service;
- (b) an Event of Force Majeure (where applicable);
- (c) any form of exchange control restriction or requirement of whatsoever nature affecting the availability, convertibility, credit or transfer of currencies, commodities, Securities, financial instruments or funds, any form of debt or other moratorium in any jurisdiction;
- (d) any devaluation, redenomination or demonetisation of the underlying currencies, commodities, Securities, Digital Tokens or instruments of any Investment or Transaction;
- (e) any restriction or requirement which, in our opinion, adversely alters or changes the rights or obligations which we undertook at the time the Investment or Transaction was entered into, the Account was established or the Service was made available to you (as the case may be); and
- (f) any other event beyond our control (including any default on the part of our counterparty to hedging arrangements effected by us in connection with any of your Investments or Transactions) which makes it impracticable, illegal or impossible for us to perform our obligations under this Agreement or any Applicable Agreement or to effectively hedge our obligations thereunder.

"Fiduciary Placement" has the meaning given to such term in Clause 8.1 of the Product Terms.

"Financial Advisers Act" means the Financial Advisers Act 2001 of Singapore.

"foreign currency" refers to any currency other than Singapore Dollars.

"**Fund**" means any investment company, partnership, unit trust, private equity fund, hedge fund, offshore fund, mutual fund or other collective investment scheme.

"Fund Investment" means any Investment involving any interest, unit, sub-unit, share, stock or other securities in any Fund and, where the context so requires, any instrument evidencing ownership thereof or representing rights to receive, purchase or subscribe for the same, or evidencing or representing any other rights and interest therein.

"Fund Investments Terms" means the Specific Terms and Conditions Governing Investments and Transactions in Funds, set out in Part G of the Terms and Conditions.

"Fund Offering Documents" means, in relation to any Fund, the prospectus, offering memorandum, private placement memorandum, information memorandum, explanatory memorandum, term sheet, constituent documents, application form, subscription agreement and any other offering documents of the Fund.

"Fund Representative" means the manager, issuer, adviser or agent of any Fund.

"General Risk Disclosure Statement" includes the general risk disclosure statement, set out in Part C of the Terms and Conditions and any other risk disclosures set out in this Agreement and/or any other Applicable Agreement.

"Gold Assaying Services" means the process in which we assist you or facilitate the delivery of Precious Metals to such service provider as we may appoint from time to time, for the purposes of carrying out or arranging to be carried out the melting, moulding, hallmarking and packing of the Precious Metals into Gold Bullion.

"**Gold Bullion**" means gold bars which meet the London Bullion Market Association's standard of "Good Delivery" and fineness from time to time in effect.

"Gold Bullion Services" means the Gold Custody Services, the Gold Trading Services, and the Gold Assaying Services.

"Gold Bullion Service Fees" has the meaning given to such term in Clause 2.1(a) of the Gold Bullion Terms.

"Gold Bullion Service Provider" has the meaning given to such term in Clause 5.1(b) of the Gold Bullion Terms.

"Gold Bullion Terms" means the Specific Terms and Conditions Governing Gold Bullion Services, set out in Part H of the Terms and Conditions.

"Gold Bullion Trade Date" means the date of the purchase or sale of the Gold Bullion, as set out in the relevant Purchase Confirmation or Sale Confirmation (as the case may be).

"Gold Bullion Transactions" means any Transaction we effect with you or on your behalf in relation to the Gold Bullion, including any Gold Trading Transaction.

"Gold Custody Services" means the storage of Gold Bullion for and on your behalf, under the Gold Bullion Terms.

"Gold Trading Services" means the purchase and sale of Gold Bullion for and on your behalf, under the Gold Bullion Terms.

"**Gold Trading Transaction**" means any Transaction involving the trading of Gold Bullion in accordance with Clause 4 of the Gold Bullion Terms.

"Hold Mail Service" has the meaning given to such term in Clause 19.1 of the Account Terms.

"Instruction" means any request, application, authorisation, order or instruction from you or any Authorised Agent in relation to any Account, Service, Investment and/or Transaction, in whatever form, substance and manner as may be required or otherwise acceptable to us and howsoever sent, given or transmitted (whether by mail, telecommunications, computer or other electronic terminal, equipment or system or otherwise as long as it is an acceptable mode of transmission or communication to us) at any time whatsoever, or which the Bank reasonably believes to be your request, application, authorisation, order or instruction.

"Investments" means Securities, Funds, structured notes, Digital Tokens, Deposits, partnership interests (including limited partnerships), instruments entitling the holder to Securities (such as warrants), certificates representing Securities, futures, contracts for differences, foreign exchange contracts of any kind, derivatives of any kind (including swaps), long term insurance contracts, certificates of deposits and commodities (including Gold Bullion), together with such other rights and interests purchased and/or procured by you through us or deposited with us or as are expressly or impliedly made the subject of this Agreement.

"ISDA Master Agreement" means the Master Agreement, published by the International Swaps and Derivatives Association, Inc., and the Schedule thereto, in the form agreed by us.

"Joint Account" means any Account opened in the joint names of two or more persons.

"Joint Account Holder" means a person in whose joint name a Joint Account is opened.

"Liability" means at any time (a) all sums (whether principal, interest, fees, costs, charges, expenses, commissions or otherwise) which are or at any time may be or become due from or owing by you to us, or which you have agreed to pay or discharge, whether actually or contingently, under or in connection with this Agreement and/or any Applicable Agreement; and (b) all other liabilities and monies which now are or at any time may be or become due from or owing by, or be incurred by, you to us, in whatever currency those sums, liabilities or monies shall be denominated or owing, whether alone or jointly with any other person and on any account whatsoever, whether current or otherwise, and whether present, future, actual or contingent and whether as principal debtor, guarantor, surety or otherwise howsoever.

"Losses" means all and any losses, whether direct or consequential (including physical loss, wrong delivery or destruction of, or damage to any Gold Bullion), damages, costs, expenses (including all duties, taxes or other levies and legal fees on a full indemnity basis), charges, penalties, actions, suits, proceedings, claims, demands and all other liabilities of whatsoever nature or description howsoever arising.

"Mailing Address" means the mailing address indicated in the Account Application or last notified in writing by you to us.

"MAS" means the Monetary Authority of Singapore and/or its successors.

"Matched Trade" has the meaning given to such term in Clause 2.3(b)(ii) of the Digital Token Terms.

"Minor" means a person under the age of 21.

"Network Transfer-Out Instruction" has the meaning given to such term in Clause 3.9(d)(i) of the Digital Token Terms.

"Nominee" means DBS Nominees (Pte) Ltd or any other nominees appointed by us from time to time.

"Notification Period" has the meaning given to such term in Clauses 3.10(b) or 3.10(c) of the Gold Bullion Terms, as the case may be.

"Offering Documents" has the meaning given to such term in Clause 7.2 of the Securities Terms, and/or the relevant documents in relation to any Investment or Transaction.

"Participating Merchants" has the meaning given to such term in Clause 21.1 of the Account Terms.

"Payment Services Act" means the Payment Services Act 2019 of Singapore.

"Personal Data Protection Act" means the Personal Data Protection Act 2012 of Singapore.

"Personnel" means any of our directors, officers, employees, servants, agents and representatives.

"Pre-Agreed Exchange Rate" means, in relation to a Currency Linked Investment, the agreed exchange rate at which the principal amount and interest accrued thereon is converted into the Alternate Currency when repayment is to be made in the Alternate Currency on the CLI Maturity Date.

"Precious Metals" means gold bullion, and may be in the form of a bar, any artefact, or any other form said to be gold, or other similar precious metal.

"Premature Termination" means:

- (a) in relation to any Deposit, any withdrawal of the whole or any part of such Deposit before the Deposit Maturity Date; or
- (b) in relation to any investment product, any termination of such investment product prior to its maturity.

"Premature Termination Costs" means all such administrative and other charges as we may, in our absolute discretion, impose in connection with any Premature Termination and all Losses (including any Losses incurred by us in unwinding corresponding or hedge positions and all legal costs on a full indemnity basis) incurred by us by reason of or in connection with any Premature Termination by you.

"Privileged Card" has the meaning given to such term in Clause 21.1 of the Account Terms.

"Privileged Services" has the meaning given to such term in Clause 21.1 of the Account Terms.

"**Product Terms**" means the Specific Terms and Conditions Governing Products and Services, set out in Part E of the Terms and Conditions.

"Purchase Confirmation" has the meaning given to such term in Clause 4.2(f) of the Gold Bullion Terms.

"Purchase Request" has the meaning given to such term in Clause 4.2(a) of the Gold Bullion Terms.

"Remittances" has the meaning given to such term in Clause 11.5 of the Account Terms.

"Sale of Goods Act" means the Sale of Goods Act 1979 of Singapore.

"Sale of Goods (United Nations Convention) Act" means the Sale of Goods (United Nations Convention) Act 1995 of Singapore.

"Sale Confirmation" has the meaning given to such term in Clause 4.3(g) of the Gold Bullion Terms.

"Sale Request" has the meaning given to such term in Clause 4.3(a) of the Gold Bullion Terms.

"Securities" means any asset of whatever nature of a type commonly referred to as securities, including stocks, shares, notes, options, warrants, debentures, book-entry securities, bonds, certificates of deposit instruments, and other equity or debt instruments, rights and obligations over any of the same and any analogous instrument, right or derivative product related thereto, together with all interest, dividends, distributions, bonuses, allotments, accretions, entitlements and other rights and benefits arising therefrom or attaching thereto.

"Securities and Futures Act" means the Securities and Futures Act 2001 of Singapore.

"Securities Terms" means the Specific Terms and Conditions Governing Securities Transactions, set out in Part F of the Terms and Conditions.

"Security Tokens" are typically in the form of a digital token in a blockchain environment representing an interest in an underlying asset. It may represent assets including equity, investment funds, real estate, debt or future cash flow or other value, but does not include Digital Payment Tokens.

"Securities Transaction" means any Transaction pursuant to which Securities are acquired, purchased and/or sold.

"Security Provider" means you and/or any other person or entity furnishing security for any Credit Facilities and/or any party to any security document.

"Services" means all services, financial and non-financial, made, or to be made available by us, DBS Private Bank, which includes any core service such as financial banking or other service including electronic services, products, information, functions and facilities which may be offered by us from time to time, and also includes the Credit Facilities, Gold Bullion Services and Transactions entered into under this Agreement and/or any Applicable Agreement and "Service" means any one of them.

- "Settlement Account" means an Account managed by DBS Private Bank, whether in Singapore Dollars or any other currency, for purpose of effecting settlement in relation to any Investment, Transaction and/or Service made available by us.
- "Singapore Dollars" means the lawful currency of Singapore.
- "SMS" means short message service.
- "Specific Terms" means collectively, the Custody Terms, the Securities Terms, the Product Terms, the Fund Investments Terms and the Gold Bullion Terms and any other specific terms covering any Investment, Transaction and/or Service made available by us from time to time.
- "Structured Deposit" means a deposit under which any interest is payable, or is at risk, in accordance with a formula which is based on the performance of an Underlying.
- "Substituted Portion" has the meaning given to such term in Clause 3.4(d) of the Gold Bullion Terms.
- "Supported Cryptocurrencies" means Digital Payment Tokens which are from time to time supported by the Bank for the purposes of providing the Digital Token Services (which may be amended from time to time at the Bank's absolute discretion without prior notice to you).
- "Supported Currencies" means currencies which are from time to time supported by the Bank in connection with Digital Token Transactions (which may be amended from time to time at the Bank's absolute discretion without prior notice to you).
- "Supported Digital Tokens" includes Supported Cryptocurrencies and Supported Security Tokens.
- "Supported Security Tokens" means Security Tokens which are from time to time supported by the Bank for the purposes of providing the Digital Token Services (which may be amended from time to time at the Bank's absolute discretion without prior notice to you).
- "Terminated Transaction" has the meaning given to such term in Clause 4.7(b) of the Gold Bullion Terms.
- "Termination Notice" has the meaning given to such term in Clause 4.7(b) of the Gold Bullion Terms.
- "**Terms and Conditions**" means these terms and conditions, comprising these definitions and interpretation, the Account Terms, the General Risk Disclosure Statement and the Specific Terms.
- "Third Party Claim" has the meaning given to such term in Clause 2.8(c) of the Gold Bullion Terms.
- "Third Party Institution" has the meaning given to such term in Clause 8.1 of the Product Terms.
- "Trading System" means the trading system operated by or on behalf of the Digital Exchange which allows for the trading of certain Digital Tokens and which is made accessible in such form and manner as may be prescribed by the Digital Exchange from time to time.
- "Transactions" means any transactions as we may from time to time permit you to carry out pursuant to or in connection with any Accounts, Services and/or Investments, including Securities Transactions, Derivative Transactions, Gold Bullion Transactions, Digital Token Transactions, foreign exchange transactions, spot or futures transactions, Commodities Investments, traded option transactions or combination of one or more of any of the foregoing.
- "Transaction Record" means a written document or notice (including notices issued electronically and/or by fax) containing the specific terms and conditions of an Investment or Transaction entered into by you, whether or not it refers to this Agreement, including transaction advices.
- "Transfer-In Instruction" has the meaning given to such term in Clause 3.9(b)(i) of the Digital Token Terms.
- "Transfer-Out Instruction" has the meaning given to such term in Clause 3.9(c)(i) of the Digital Token Terms.
- "Transferred Portion" has the meaning given to such term in Clause 3.4(d) of the Gold Bullion Terms.

"Trust Deed" has the meaning given to such term in Clause 8.1(a) of the Account Terms.

"Underlying" includes (and is not limited to) any one or more currencies, foreign exchange forwards, equities, equity indices, bonds, interest rates, interest rate futures, index futures, commodities, funds / collective investment schemes, any other type of capital market products / securities, any other money market or financial instruments or a combination of any or two more financial instrument, and digital assets.

"US Dollars" means the lawful currency of the United States of America.

"U.S. Person" means: (i) an individual that is located in, or resident of, the United States; (ii) an entity that is incorporated or located in the United States; (iii) an entity that is incorporated or located outside the United States, but (a) whose trading activity is arranged, directed or controlled by a U.S. Person, or (b) whose trading strategies are substantially identical to, or coordinated with, any U.S. affiliate; or (iv) a "U.S. Person" as defined under Rule 902 of Regulation S of the Securities Act of 1933, as may be amended from time to time.

"Withdrawn Digital Tokens" has the meaning given to such term in Clause 3.10 of the Digital Token Terms.

"you" or "your" or "yourself" means you in your capacity as our customer and the person in whose name any Account is maintained or to whom any Service is provided, and where the context so permits, includes the successor(s), legal representative(s) (in the case where you are insane or mentally incapacitated), trustee(s) in bankruptcy, personal representative(s) (in the case where you pass away) or your Authorised Agent.

2. Interpretation

In this Agreement, unless the context requires otherwise:

- (a) words are not gender-specific;
- (b) words defined which are expressed in the singular shall, where the context allows, include the plural and vice versa;
- (c) references to a "*Clause*" or "*Part*" shall be construed as a reference to a Clause or Part in this Agreement;
- (d) headings are included for ease of reference only and shall not be taken into account in construing any provision of this Agreement;
- (e) all references to a time of day shall, in the absence of any express statement to the contrary, be references to Singapore time;
- (f) references to statutes and other legislations shall be construed as a reference to such statute or legislation as amended, replaced or re-enacted from time to time;
- (g) references to "*this Agreement*" or other documents (howsoever called) shall include variations, modifications and/or replacements thereof and supplements thereto;
- (h) references to "person" or "party" shall include, where the context so permits, any individual, company, body corporate, firm, partnership, limited liability partnership and any other business concern, statutory body and agency and government authority, whether local or foreign; and
- (i) references to "include", "includes" and "including" as they appear in this Agreement are not limiting and are deemed in each instance to be followed by the words "without limitation" or "but not limited to".

PART B – GENERAL TERMS AND CONDITIONS GOVERNING ACCOUNTS AND SERVICES

1. Introduction

- 1.1 This Agreement is applicable to each and every Investment and Transaction entered into by you, your Accounts and all Services offered and/or made available by us to you from time to time. Any offer and/or availability of Accounts and Services by us under this Agreement are at our absolute discretion.
- 1.2 You may, in addition to your Account, open and maintain one or more Branch Accounts with any Branch. Unless otherwise agreed, your utilisation of all such Branch Account and/or any of the services generally offered by the Bank in connection therewith shall be governed by the terms and conditions governing such Branch Accounts and/or services (available at www.dbs.com.sg/corporate/list/cash-list.page (for non-individual Branch Accounts) and www.dbs.com.sg/personal/deposit (for individual Branch Accounts)) and not by this Agreement.
- 1.3 All of your Investments, Transactions, Accounts and Services shall be subject to all Applicable Laws. You hereby authorise us to do all such things as we deem necessary to comply with such Applicable Laws.
- 1.4 In this Agreement, in the event of any conflict or inconsistency between these Account Terms and any of the Specific Terms, the latter shall prevail.
- 1.5 In the event of conflict or inconsistency between this Agreement and the provisions of any Applicable Agreement, the latter shall prevail in respect of the Service made available pursuant to that Applicable Agreement.
- 1.6 For the avoidance of doubt, our rights as set out in this Agreement shall always be in addition to and without prejudice to any other rights or remedies to which we may be entitled under any other agreement and/or document entered into with or by you (and conversely, our rights in any other agreement and/or document entered into with or by you shall be in addition to and without prejudice to any other rights or remedies as set out in this Agreement).

2. Accredited Investor Status

- DBS Private Bank will only provide Services relating to Investments under this Agreement to clients who qualify and agree to be treated by us as an accredited investor, as defined in the Securities and Futures Act.
- 2.2 You will promptly notify us if, at any time, you do not qualify as an accredited investor or do not wish for us to treat you as an accredited investor. In such event, you acknowledge that we will no longer be able to make available to you any new Investments and certain Services may be discontinued. However, we will continue to serve you as an accredited investor in respect of any Investments undertaken by you prior to such event.

3. Communications and Instructions

3.1 Opening of Accounts

You authorise us, in our absolute discretion, to open and/or maintain an Account in your name and at any time subsequently to open such further Account of whatever nature in your name as you may direct or as we may in our absolute discretion deem fit. Unless otherwise agreed by us in writing, this Agreement shall apply to each such Account.

3.2 Mandate and Instructions

- (a) You agree that all Instructions in relation to the matters dealt with under this Agreement and/or any Applicable Agreement may only be given by you and/or your Authorised Agent strictly in accordance with the mandate for the time being in force and all such Instructions are valid and binding on you.
- (b) You agree to perform and ratify any Applicable Agreement, Investment and Transaction entered into or action taken by us as a result of such Instructions.
- (c) Until we receive notification in writing of the revocation of the appointment of your Authorised Agent, we are authorised (but shall not be obliged) to accept Instructions from such Authorised Agent provided always that we may, in our absolute discretion, require any such Instruction to be confirmed by you prior to executing the same.

- (d) You authorise us to accept, without verification, any Instruction which we reasonably believe to originate from you and/or your Authorised Agent, regardless of the circumstances prevailing at the time the Instructions were given or the nature or amount of any Transaction effected pursuant thereto and notwithstanding any error, misunderstanding, error in transmission, fraud, forgery or lack of clarity in the terms of such Instructions or lack of authority in relation to the Instructions. For the avoidance of doubt, we shall be under no duty to inquire into the authenticity of any Instruction or the identity or authority of the person giving or purporting to give any Instruction and you acknowledge and agree that you are under an express duty to prevent any fraudulent, forged or unauthorised Instruction from being given to us.
- (e) You authorise us to accept, rely and act upon any Instruction given or purported to be given by you or your Authorised Agent:
 - (i) orally, whether by telephone or otherwise;
 - (ii) which may be transmitted to us by facsimile, via email, electronic platforms or other electronic means;
 - (iii) in such manner as we may determine and/or deem fit to accept from time to time.
- (f) Without limitation to the generality of Clause 31, you acknowledge that you are aware of the risks involved in the use of postal services, telephone, facsimile, email, electronic platforms, SMS, electronic communication or similar means of non-electronic and electronic transmission, which may include errors in transmission, technical defect, oversight, power or system failure, fraud, forgery, misunderstanding, theft or loss of mobile or other devices, unintended disclosure, unauthorised interception, manipulation by third parties or any Event of Force Majeure. By authorising us to accept Instructions and/or effect Communications through such means, you agree that you shall bear all such risks.
- (g) If you or your Authorised Agent transmit to us Instructions via email, electronic platforms or other electronic means, you acknowledge and confirm that any Instruction shall not be deemed received by us until actual receipt by us and our confirmation of such receipt and you further acknowledge and accept the risks associated with the time lag between such Instruction being sent and actual receipt and confirmation by us of such receipt.
- (h) All Instructions must be in the English language and shall sufficiently identify the Account through which any Investment or Transaction should be undertaken. If your Instructions are given in a language other than English, we may, in our absolute discretion (but shall not be obliged to), choose to accept those Instructions but in such case, our interpretation of such Instructions shall be final and binding on you. Further, all Communications shall be in the English language. If you require the Communications to be given in a language other than English, we may, in our absolute discretion (but shall not be obliged to), choose to provide a translation of the Communication purely for informational purposes and in the event of inconsistency or different interpretation between the English language and the language other than English, the English language version will prevail.
- (I) Notwithstanding any provision to the contrary in this Agreement and/or any Applicable Agreement, we shall be entitled to, without notice and/or without disclosing any reason whatsoever, refuse to accept or act on any Instruction, or require you to provide to us the original Instruction in writing (where the Instruction had been transmitted to us via facsimile, email, SMS, electronic platforms or other electronic means of communications) or confirm any Instruction prior to acting on the same, including in circumstances where:
 - (i) there is ambiguity or conflict in the Instructions given to us and/or the Instructions are not in accordance with the mandate for the time being in force and/or we have any reasonable doubt on the authenticity, clarity or completeness of the Instructions given to us;
 - (ii) you are or appear to be unable or unwilling to provide us with such information or documents or assistance as we may reasonably require;
 - (iii) we reasonably believe that you are not capable, whether mentally or otherwise, of managing yourself, your Accounts, Investments and Transactions; and/or
 - (iv) the Instructions given to us are inconsistent with and/or contrary to any Applicable Law.

- (j) All Instructions given pursuant to this Agreement and/or any Applicable Agreement may not be revoked or amended unless we agree otherwise.
- (k) Any standing Instruction in respect of the operation of your Accounts shall cease to have effect upon receipt by us of notice of your death, incapacity, bankruptcy, winding up or insolvency.

3.3 Operation of Accounts

You authorise us to and we may (but shall not be obliged to):

- honour and comply with all cheques, drafts, orders to pay, bills of exchange, promissory notes and all other orders for payment whatsoever expressed to be drawn, signed, accepted, endorsed, made or given by you or on your behalf drawn upon or addressed to or made payable by us, whether your Account is in credit or debit or may become overdrawn as a result thereof (but always without prejudice to our right to refuse any overdraft or increase of overdraft beyond any limit as may be prescribed by us from time to time);
- (b) honour and comply with all Instructions given or purported to be given by you or your Authorised Agent, whether to withdraw monies from any of your Accounts, deliver, dispose of or deal with any Asset, purchase, sell or otherwise deal in any Investment or undertake any other Transaction in relation to your Accounts; and
- (c) make available to you any Credit Facilities and, by way of security, accept as duly signed or executed on your behalf (whether in writing or by electronic means as prescribed by the Bank) any document creating or evidencing any charge, mortgage, pledge or other security interest whatsoever over or in respect of any Asset, provided that each such document shall have been signed by or, in our determination, appears to have been signed by, you or your Authorised Agent (whether in writing or by electronic means as prescribed by the Bank).

3.4 Business Days

We shall accept and execute your Instructions during banking hours on Business Days.

- 3.5 Consent to Audio or Electronic Recording and/or Written Records
 - You authorise us to record any oral Instruction from you and/or your Authorised Agent and any communications between you and/or your Authorised Agent and us (including telephone and face-to-face conversations) by audio or electronic recording devices and/or in writing. Any such record made by us shall be final and conclusive as to the fact and contents of such oral Instructions and communications and shall be binding on you.
 - (b) Any such record shall be our property and you agree you are not entitled to any such record.
- 3.6 Information on Services and Products
 - (a) We may, from time to time, provide you with information on Services and products (including research, commentaries and other publications).
 - (b) Such Communications may be sent via email, electronic platforms or means, SMS, telephone calls or in such other manner as we may deem appropriate.

4. Constraints in Execution of Instructions

- 4.1 We will endeavour to execute all Instructions as accepted by us within a reasonable period of time. However, you acknowledge that due to various constraints including:
 - (a) volatility in Securities and commodities prices or exchange rates on most Exchanges; and/or
 - (b) any requirement under Applicable Laws, including those pertaining to the prevention of fraud, money-laundering, terrorist financing and the provision of financial or other services to any person or entity which may be subjected to sanctions,

we may not always be able to trade at the prices or rates quoted to you at any specific time and/or there may be delay and/or failure by us in processing and/or effecting any Instruction and/or in performing any of our duties or obligations under this Agreement and/or any Applicable Agreement. Notwithstanding the foregoing, you agree to accept and be bound by the outcome of any Transaction entered into by you.

5. Joint Account

- 5.1 Each Joint Account Holder jointly and severally agrees to all the terms in this Agreement and/or any Applicable Agreement and shall be jointly and severally liable for all obligations and Liabilities incurred on or in respect of such Joint Account and pursuant to this Agreement and/or any Applicable Agreement. The Liability of each Joint Account Holder shall not be discharged or affected in any way by the death or incapacity of any other Joint Account Holder.
- 5.2 In relation to a Joint Account opened with a joint alternate authority, any Instruction given and/or any Applicable Agreement and/or other documents duly signed by any one Joint Account Holder (whether in writing, orally or by electronic means as prescribed by the Bank) in accordance with the signing requirements specified in the Account Application will be accepted and be binding on each and every Joint Account Holder.
- 5.3 Where the Joint Account is operated with joint authority, any written Instruction may be given by the Joint Account Holders in one or more counterparts, all of which when taken together shall constitute one and the same Instruction and will be binding on all the Joint Account Holders. Notwithstanding the foregoing and/ or any signing requirement specified in the Account Application, you agree that any Joint Account Holder may give Instructions (orally or by electronic means as prescribed by the Bank) to us singly and we shall be entitled (but not obliged) to act on such Instructions and all the Joint Account Holders shall be liable for any such Instruction as if they had given such Instructions jointly.
- We may (but shall not be obliged to) credit the Joint Account with funds received in favour of any Joint Account Holder without prior notice unless specific Instructions to the contrary are given to us.
- 5.5 You agree to dispense with giving each Joint Account Holder a separate set of Communications and/or any other correspondence in relation to the Joint Account and that only one set of Communications and/or correspondence shall be sent to the Mailing Address. Notwithstanding any other provision to the contrary in this Agreement, any Communications and/or other correspondence sent, despatched or delivered by us to any one Joint Account Holder shall be deemed to have been sent and received by all the Joint Account Holders.
- 5.6 (a) Upon receipt of notice of the death of any Joint Account Holder, we may, subject to the production and/or execution of such documents (including any grant of probate or letters of administration) and/or indemnities as we may require, hold the Assets in the Joint Account to the order of the surviving Joint Account Holder(s). This shall be without prejudice to any right (whether present, future, actual, contingent or otherwise) that we may have in respect of any Asset in the Joint Account arising out of any lien, charge, pledge, set-off, counterclaim or otherwise or to any step which we may consider desirable to take in respect of the Joint Account to establish the legality of such holding or in view of any claim by any person other than the surviving Joint Account Holder(s).
 - (b) Any transfer made by us in our absolute discretion of the Assets in the Joint Account to the surviving Joint Account Holder(s) shall constitute a complete discharge of our obligations under this Agreement and/or any Applicable Agreement and we shall be released from all demands, claims, suits and actions whatsoever by the heirs, beneficiaries, executors and/or administrators of the deceased Joint Account Holder.
 - (c) Without limitation to the generality of Clause 31, each surviving Joint Account Holder and/or the personal representatives of any deceased Joint Account Holder shall indemnify and hold us harmless from and against any claim, demand, action and proceeding that may be brought against us and any Losses which we may incur or suffer arising out of or in connection with:
 - (i) the closure of the Joint Account and, if applicable, the opening of a new bank account and/or the transfer of any Asset from the Joint Account to such new account;
 - (ii) any transfer of the Assets in the Joint Account to the surviving Joint Account Holder(s);
 - (iii) us acting upon or carrying out any Instruction given or purported to be given by the surviving Joint Account Holder(s) and/or the personal representatives of any deceased Joint Account Holder;
 - (iv) any suspension and/or freezing of the operation of the Joint Account and the retention of the Assets in the Joint Account; and
 - (v) any dispute between any surviving Joint Account Holder and any personal representative of the deceased Joint Account Holder.

6. Accounts with Minors

- 6.1 We may (but shall not be obliged to) allow your application to operate a Joint Account on a joint basis with a Minor, subject always to the following:
 - (a) we may impose such conditions and/or restrictions on the operation of the Joint Account and/or the availability of the Services as we may in our absolute discretion consider appropriate;
 - (b) we shall have the right to (i) decline to execute any Instruction given by the Minor without assigning any reason thereof or (ii) require the parent or legal guardian of the Minor to consent to any such Instruction prior to us acting on the same; and
 - (c) you agree that you shall at all times be responsible for all Instructions given by the Minor in relation to such Joint Account and for ensuring the Minor's compliance with this Agreement and/or any Applicable Agreement.
- 6.2 You further undertake to indemnify and to hold us harmless against any claim, demand, action and proceeding that may be brought against us, and any Losses which we may incur or suffer, arising out of or in connection with us acting upon or carrying out any Instruction given by the Minor.

7. Partnership Account

- 7.1 If you are, and your Account is opened in the name of, a partnership (other than a partnership having separate legal personality), any change in:
 - (a) the name of the partnership;
 - (b) the partners of the partnership as a result of death, retirement or introduction of new partner; or
 - (c) the constitution of the partnership,

shall not affect your Liabilities which shall continue and be binding on you and all partners from time to time constituting the partnership. We shall be entitled to debit your Account at any time in respect of any sum howsoever due or owed to us by any partner from time to time constituting the partnership.

8. Trustee Accounts

- 8.1 Without prejudice to any provision in this Agreement and/or any Applicable Agreement, where you are acting as trustee of a trust, you undertake, represent and warrant to us that:
 - (a) you have full capacity, power and authority, in your capacity as trustee of the trust, in accordance with the terms of the document constituting the trust (the "**Trust Deed**") and all Applicable Laws, to enter into and to perform and deliver this Agreement and any Applicable Agreement (including any Applicable Agreement relating to Credit Facilities made available to you); and
 - (b) you shall ensure that all Instructions given and all Investments and Transactions undertaken by you, in your capacity as trustee of the trust, are in accordance with the terms of the Trust Deed and all Applicable Laws.
- You acknowledge and agree that we will not, nor shall we be under any duty or obligation to, verify and/or monitor any of the matter referred to in Clause 8.1 and/or review or retain a copy of the Trust Deed.
- 8.3 Where you are an individual trustee, upon receipt of notice of your death or incapacity (whether mental or otherwise), we may, subject to the production and/or execution of such documents (including any grant of probate or letter of administration) and/or indemnities as we may require, do any of the following:
 - (a) if your Account is in the name of more than one trustee, we may hold the Assets in your Account to the order of the surviving or other trustee(s); or
 - (b) if there is no surviving or other trustee:
 - (i) if the beneficiary is a corporation or if none of the beneficiaries (being individuals) is a Minor, we may close your Account and transfer the Assets therein to the beneficiary or each of the beneficiaries in equal shares; or
 - (ii) if any of the beneficiaries is a Minor, we may close your Account and transfer the Assets therein to your legal or personal representatives to be held in trust for the beneficiaries.

Notwithstanding the foregoing, we may, in our absolute discretion suspend and/or freeze the operation of your Account until we are able to determine, to our satisfaction, the person(s) entitled to ownership, management or control of the Assets therein.

9. Death or Incapacity of Individuals

Subject to Clause 5, upon our receipt of notice of your death or incapacity, we shall not deal with and/or accept Instruction from any person other than your legal or personal representatives (including your administrators or executors), provided always that we may, in our absolute discretion, require your legal or personal representatives to produce and/or execute such documents (including a court order, grant of probate or letters of administration) and/or indemnities as we may stipulate. At no point in time shall we be obliged to enquire into the propriety of the appointment of and/or any Instruction from your legal or personal representatives. Accordingly, we shall not be liable for any Losses suffered by any person resulting from us acting reasonably on the Instructions of your legal or personal representatives.

10. Deposits

- 10.1 Any deposit that you make shall be in the manner and/or in the currency (where applicable) as we may prescribe and/or agree to from time to time and shall be subject to our right to levy and/or charge any prevailing prescribed commission and/or service charge. Notwithstanding the foregoing, we may, in our absolute discretion, at any time, without disclosing any reason, refuse to accept any deposit for your Account, limit the amount that may be deposited and/or return all or any part of the deposit.
- 10.2 Unless otherwise agreed by us, cash deposits will not be accepted into your Settlement Account.
- 10.3 If you instruct us to credit your Account with the equivalent in the currency of that Account of any deposit denominated in a foreign currency, we have the right to use such rate of exchange as we may conclusively determine.
- All cheques and/or other financial instruments deposited with us for collection are received by us solely as agent for collection with no responsibility for non-payment or any Losses that you may suffer or incur resulting from any inability on our part to collect under any circumstance or cause whether beyond our control or otherwise.
- 10.5 We may (but are not obliged to) give immediate credit for cheques and/or other financial instruments deposited into your Account but such deposits shall not be available for withdrawal or transfer until such cheques and/or other financial instruments have been cleared and the proceeds have in fact been received by us. Notwithstanding the foregoing, where we agree, in our absolute discretion, to accept foreign currency financial instruments for clearing, we shall only credit your Account with the proceeds after adjustments have been made for the difference in, where applicable, exchange rates, bank commissions, stamp duty and other charges upon clearance and all risks connected with the clearing of such foreign currency financial instruments, including exchange rates movement, shall be borne by you.
- 10.6 We reserve the right to debit, earmark and/or revise any amount credited into your Account without prior notice to or consent from you if:
 - (a) the cheques and/or other financial instruments presented for deposit are subsequently dishonoured;
 - (b) the payment received by us on any cheque and/or other financial instrument presented for deposit is less than the amount credited into your Account;
 - (c) the correspondent paying bank, financial institution or any agent and sub-agent in relation to any cheque, draft, money order, telegraphic transfer or other payment instruction or financial instrument claims a refund or repayment from us of the monies credited to your Account on any grounds including fraud, illegality and/or invalidity (whether or not disputed) and we shall be under no obligation to dispute any ground raised or the claim for refund or repayment;
 - (d) we are notified or become aware that such amount had been credited into your Account due to a mistake, error or omission, and/or
 - (e) we (i) are notified of or in good faith suspect that your Account has been used for unlawful purposes or that there has been any misuse of your Account; or (ii) deem it necessary or appropriate in good faith to comply with applicable laws, regulations, notices, requests or directives issued by any government or regulatory body.

11. Withdrawals

- 11.1 Withdrawal (where the context so permits, includes transfer of funds, telegraphic transfers and demand drafts) from your Account shall be up to the limit that we may prescribe and in the manner and on such terms and conditions as we may prescribe from time to time. Notwithstanding the foregoing, we may, in our absolute discretion, at any time, without disclosing any reason, refuse to allow any withdrawal from your Account and/or limit the amount that may be withdrawn.
- 11.2 Unless otherwise agreed by us, cash withdrawals may not be made from your Settlement Account.
- 11.3 You shall maintain sufficient funds in your Account to meet all payments and withdrawals and you agree that we may impose fees for any unsuccessful debit Instructions.
- In the event that we receive any withdrawal Instruction for payments or in relation to any Investment, Transaction and/or Services, the amount of which, in the aggregate, would exceed the credit balance on your Account or the authorised limit of any Credit Facilities made available to you, we may decline to act on such Instruction provided that if we should elect to act on such Instruction, we shall have the right to process such payments and/or withdrawals in any order or priority as shall be determined by us, in our absolute discretion.
- 11.5 Where withdrawals are made in the form of cashier's orders, demand drafts and/or other forms of remittances (collectively, the "**Remittances**"), the following terms and conditions shall apply:
 - (a) the Remittance is subject to all Applicable Laws;
 - (b) we may remit funds on your behalf by any means, including by mail, telex, cable or SWIFT and may make use of any Agent and we shall not be liable for any mutilation, interruption, failure, error or delay occurring in the wire, cable or mails, or any act, omission, negligence or default on the part of any postal authority, telegraph, cable or wireless company or Agent;
 - (c) in the event a refund from or repurchase by us of the amount of any Remittance is desired, such refund or repurchase will be made, subject to such conditions as we may prescribe, at our prevailing buying rate for the currency in question less all costs, charges, expenses and interest (where applicable), provided that we are in possession of the funds which are subject of the Remittance, free from any exchange or other restriction; in the case of a cashier's order or demand draft, such cashier's order or demand draft must be duly endorsed by you and returned to us; and
 - (d) in the event a cashier's order or demand draft is lost, stolen, or destroyed, you may request for payment on such cashier's order or demand draft to be stopped, and for a replacement to be issued or for a refund of the amount thereof, and such request shall be subject to our consent and upon such conditions as we may in our absolute discretion stipulate, including the execution of an indemnity in our favour and the payment of such fees or charges as we may impose.

12. Interest On Accounts

- 12.1 We will pay interest (or charge negative interest, where applicable) at our prevailing interest rate for any interest bearing Account and where interest is payable to you, such interest shall be calculated in such manner as prescribed by us from time to time.
 - (a) Where interest is payable to you, such interest will be credited to your Account on a monthly basis or at such other intervals as we may determine from time to time. For the avoidance of doubt, no interest will accrue or be paid by us on any balances in a dormant Account.
 - (b) Where negative interest is chargeable on your Account, such interest will be debited from your Account at such intervals as we may determine from time to time.
- 12.2 Notwithstanding any provision to the contrary in this Agreement and/or any Applicable Agreement, we reserve the right to do any of the following from time to time and under any circumstances as we may reasonably deem fit (including but not limited to, where market rates (based on the currency in which your deposit is denominated) fall below zero:
 - (a) reduce, suspend or stop the payment of interest on any credit balance in your Account;
 - (b) impose zero or negative interest rates on your Account, and you shall pay us the relevant fees and charges and any amount owing to us (including negative interest, service fees, if any);

- (c) impose and debit such fees or charges from your Account; and/or
- (d) in the case of Deposits, revise the Deposit Period.

13. Overdrafts

- 13.1 You shall ensure that none of your Accounts becomes overdrawn or exceeds any overdraft limit specified by us.
- 13.2 Without prejudice to Clause 13.1 above, if any Instruction given or any Investment or Transaction undertaken by you results or may result in your Account becoming overdrawn or exceeding any overdraft limit specified by us, we may take such action as we deem fit, including:
 - (a) permitting your Account to be overdrawn whereby interest and bank charges shall be computed on such overdrawn amount at such rate and on such basis as we may in our absolute discretion determine. The overdrawn amount and all accrued charges and interest shall be repayable by you to us on demand:
 - (b) terminating any outstanding Investments or Transactions; and/or
 - (c) transferring such amounts as may be necessary from any of your Bank Accounts to restore your Account to credit.

14. Foreign Currency Transactions

Unless otherwise agreed to in writing between you and us, when we effect your Instructions in relation to any Investment and/or any Transaction denominated in a foreign currency (whether on any Exchange or otherwise), we shall be entitled, without prior notice to or consent from you, to convert the monies in your Account into and from such foreign currency (notwithstanding that such foreign currency may not be freely convertible and/or transferable) at a rate of exchange to be determined by us, in our absolute discretion, based on the then prevailing market rates, and you shall indemnify us for all Losses arising from or in connection from such currency conversion.

15. Your Obligations for Interests, Charges, Commissions & Taxes

- 15.1 All amounts due from you to us (including any overdrawn amount) shall bear interest (including default interest) at such rate, and shall be calculated on such basis, as we may, in our absolute discretion, determine. Without prejudice to the foregoing, the interest on any amount due and owing to us (including capitalised interest) shall be capitalised at such interval as we may determine and added to the principal sum then due and owing and shall bear interest at the rate stipulated by us and be secured, if security has been provided, and payable accordingly until full payment is received by us.
- 15.2 Interest shall be payable to us on all amounts due and owing by you to us before as well as after any demand made or judgment obtained by us and shall continue to accrue notwithstanding the termination of this Agreement and/or closure of your Account.
- 15.3 We shall be entitled, in our absolute discretion, to impose charges, fees or commissions for all or any of your Accounts and the Services made available by us or for any action taken by us in carrying out any of your Instructions in relation to the matters dealt with under this Agreement and/or any Applicable Agreement.
- 15.4 All stamp duties, disbursements, taxes, charges, costs and expenses and any other Liabilities of any nature whether in Singapore or in any other jurisdiction in respect of any Investment, Transaction, Account or Service shall be borne by you.
- We reserve the right, without prior notice to you, to modify at any time the rates and/or amount of interest (including default interest), charges, fees or commissions payable by you in relation to your Account, any Service and/or in relation to any of your Liabilities (including any overdrawn amount).
- 15.6 A certificate duly signed by our authorised officer as to any of your Liabilities, the applicable interest rate and/or the applicable exchange rate shall, in the absence of manifest error or fraud, be conclusive and binding on you.

- 15.7 In the event that any goods and services tax (or such other tax of a similar nature) or any other tax, stamp duty, disbursement, charge, cost and expense and/or any liability of any nature is incurred by us and/or levied on your Account, Investments and/or Transactions and is required to be paid by us, the same shall be borne and paid for by you and you shall indemnify and/or reimburse us for such payments made on your behalf in the respective currencies in which these expenses were incurred.
- 15.8 We shall have the right, from time to time, to debit your Bank Accounts at any time without prior notice to you for all amounts due and payable to us, whether in respect of any Bank Account, or in connection with any Investment, Transaction, Service or otherwise, including charges, commissions, fees, taxes, stamp duty, disbursements, costs and expenses, interest and default interest, and, if necessary, to make the currency conversions at such rates as we may determine notwithstanding that any such debiting may result in your Bank Account becoming overdrawn. Clause 13 or the Applicable Agreement governing such overdrawn amounts shall apply accordingly.

16. Payments

- All sums due to us may first be rounded up (whereby you are liable to pay the rounded up amount(s)) and must be paid promptly to us or to our order in the currency in which they are due (unless otherwise required by us), in immediately available and freely transferable funds without any set-off, counterclaim, withholding, deduction or condition of any kind (except to the extent required by any Applicable Law). Rounded up differences/ amounts are not refundable and will be dealt with in our sole discretion. If any amount is so required to be deducted or withheld by virtue of any Applicable Law, the amount payable to us shall be increased so that the net amount received by us will be equal to the full amount we would have received had no such withholding or deduction been required.
- Where any amount received or recovered by us from you is in a currency (other than the relevant currency in which such amount payable by you is denominated in, the amount received or recovered by us from you shall only constitute a discharge by you to the extent of the amount converted by us at such rate of exchange as we may conclusively determine into that relevant currency (less any cost and expense incurred by us as a result of any currency conversion).
- 16.3 Unless otherwise agreed or stated herein, all amounts due to us under this Agreement and/or any Applicable Agreement shall be repayable on demand.
- 16.4 We may (but shall not be obliged to) earmark such amount of Cash in any of your Bank Accounts as we may deem necessary for the purpose of settling any of your payment obligations to us or any third party in connection with or arising from any Investment, Transaction or Service under this Agreement and/or any Applicable Agreement, and you may not, without our prior written consent, withdraw or transfer any part of the earmarked amount until such payment obligations have been fully discharged. If at any time the credit balances in your Bank Accounts fall below the earmarked amount, you shall immediately deposit into your Bank Accounts an amount equivalent to such shortfall.

17. Discharge of Bank's Payment Obligations

- 17.1 Any sum that may be payable by us to you shall be subject to all Applicable Laws, including any withholding tax requirement, foreign exchange restriction or control. You agree and acknowledge that we may withhold, retain and/or deposit any monies payable to you into a sundry or other account pending determination of the applicability of such withholding tax requirement, foreign exchange restriction or control and we shall not be liable for any Losses that may be incurred by reason of such withholding, retention or deposit.
- 17.2 All payments from us to you shall be net of all applicable duties, disbursements, taxes, charges, fees, commissions, costs or other expenses,
- Unless otherwise specified in any Applicable Agreement, any sum payable by us to you shall be paid on a Business Day (or if the payment date is not a Business Day, on the next Business Day immediately following such payment date). Unless otherwise instructed by you, such payment may be made in cash, banker's draft, cashier's order, by credit into any of your Bank Accounts, or by such other manner as we deem fit.

18. Account Statements and Transaction Records

We will send you Account Statements setting out any transaction and/or the details of any activity on your Account on a monthly basis or at such intervals as we may deem appropriate. In addition, we may send you a Transaction Record following the execution of any Investment or Transaction. You shall notify us in writing if you do not receive any Account Statement or Transaction Record within seven (7) days of their expected date of receipt.

- 18.2 You shall carefully examine all Account Statements and Transaction Records received from us and promptly report any irregularity, discrepancy, inaccuracy, omission, incorrect entry, error and/or unauthorised transaction within five (5) Business Days (in the case of Transaction Records) or fourteen (14) Business Days (in the case of Account Statements) from the date of receipt of such documents.
- 18.3 Unless we receive any objection from you within the time period stated in Clause 18.2, you shall be treated as having conclusively authorised and accepted as valid and correct in all respects, all matters in such Account Statements and Transactions Records which shall be conclusive and binding against you.
- 18.4 Without prejudice to the provisions of this Clause 18, you agree that we have the right (but not the obligation), without prior notice to you, to add, alter, rectify and/or correct the particulars set out in any Account Statement and/or Transaction Record to rectify any error therein.
- 18.5 You acknowledge that in relation to the market values of any Investment or Transaction that may be set out in any Account Statement and/or Transaction Record:
 - (a) such values are only indicative values as at the applicable value dates indicated in the Account Statement;
 - (b) the provision by us of such market values does not imply that an actual trading market exists for that Investment or Transaction or that it is appropriate to assume (for accounting or other purposes) that such a trading market exists;
 - (c) such market values may differ significantly from the actual trading prices (if any) for entering into, purchasing, redeeming or terminating any Investment or Transaction;
 - (d) such market values may differ from the valuations adopted by us for purposes of determining the security value of any Asset provided by you by way of security; and
 - (e) you should consult your own advisers as to the appropriateness of any particular use of such market values, whether in connection with the preparation of your financial statements, fulfilling reporting obligations or otherwise.
- Market values of bonds that trade over the counter are traded via networks of brokers and dealers. Prices reflected in our trading systems (including and not limited to online trading systems utilised by you) and statements are based on inputs and quotes from these networks and market data providers. Price sources may depend on the markets in which the bonds trade, and are indicative only. There may be a disparity between the indicative prices and the actual tradable bond prices and this is due to various factors such as market conditions, currency fluctuations, volume or liquidity of the relevant market, as well as the frequency of inputs and updates by the aforementioned networks of brokers and dealers.

18.7 You agree that:

- (a) our electronic records (if any) of any Account Statement and Transaction Record are final and conclusive as to the fact and contents of such Account Statement or, as the case may be, Transaction Record, and shall be binding on you; and
- (b) such electronic records may be referred to in any proceeding and that you shall not dispute the accuracy or the authenticity of such electronic records.

19. Hold Mail Service

- 19.1 We may (but shall not be obliged to), if so requested and authorised by you, retain on your behalf all Communications (such service being the "**Hold Mail Service**").
- 19.2 Notwithstanding your request for Hold Mail Service, we are entitled, in our absolute discretion, to forward any Communication as we consider necessary or appropriate to the Mailing Address or other address as indicated in the Account Application or last notified by you to us. You further acknowledge and agree that the Hold Mail Service shall not apply to any notice of demand from us.
- 19.3 You shall collect or give written Instructions for the dispatch and/or disposal of all Communications retained by us pursuant to the Hold Mail Service at least once every twelve (12) months. Unless otherwise instructed in writing, we may in our absolute discretion dispose of or destroy all or any such Communication in any manner as we deem appropriate eighteen (18) months from the date shown on such Communication.

19.4 You acknowledge that:

- (a) we are under no obligation to inform you of any new Communication, open any Communication, peruse its contents and/or communicate the contents thereof to you;
- (b) notwithstanding any provision to the contrary in this Clause 19, all Communications retained by us pursuant to the Hold Mail Service shall be deemed to have been duly delivered and received by you on the date as shown on such Communication and you shall be deemed to have notice thereof;
- (c) by choosing to utilise the Hold Mail Service, you agree to accept and take full responsibility for all risks and Losses associated with or arising out of the use of such Hold Mail Service, including the following:
 - (i) any fraud, omission, discrepancy or error occurring or arising in connection with your Investments, Transactions, Accounts and the Services utilised by you which are best prevented by your timely receipt and review of all Communications from us. Your use of the Hold Mail Service may lead to a delay in detecting, and you may be precluded from raising any objection to us in relation to, any such fraud, omission, discrepancy or error; and
 - (ii) you may not be aware of any amendment to, or revision of, this Agreement and/or any Applicable Agreement, or to any information whatsoever in connection with your Investments, Transactions, Account or the Services utilised by you, including amendments or revisions to fees, charges or interest rates (including negative interest, where applicable).
- 19.5 We are authorised to debit any of your Accounts for charges relating to the Hold Mail Service. If you fail to make any such payment, we shall be entitled to terminate such Service immediately and our obligations hereunder shall be fully discharged by us destroying or sending to you all Communications then retained by us pursuant to the Hold Mail Service.

20. Safekeeping Services

- 20.1 Unless otherwise agreed to by us, we may (but shall not be obliged to), at your request, accept monies or other items (approved by us) for temporary safekeeping for no more than one (1) Business Day (or such other period as we may agree), on such terms and conditions as we may determine. Without prejudice to the foregoing, we reserve the right, at any time, to refuse to accept any item for safekeeping without assigning any reason for such refusal.
- 20.2 In the absence of any Instruction from you as to the subsequent transfer or disposal of such monies or items, you authorise us to open a safekeeping account or safe deposit box at any of our Branches or with any of our Affiliates or Nominees, on your behalf, and to deposit such monies or items in such safekeeping account or safe deposit box. In such event, you agree to comply with the terms and conditions governing such safekeeping account or safe deposit box, as the case may be.
- 20.3 Where your monies or items are deposited with any of our Affiliates or Nominees, you accept that while we will exercise reasonable care in the selection of such Affiliate or Nominee, such monies or items are placed with such Affiliate or Nominee at your sole risk and we shall not be liable for any act, omission, neglect or default of such Affiliate or Nominee.
- 20.4 You may request and we may require a withdrawal of all or part of your monies or items being safekept with us or our Affiliate or Nominee at any time. Delivery of such monies or items shall be carried out at your expense, on such terms and conditions as we may stipulate, and you shall in all circumstances accept full responsibility for any Losses arising out of or in connection with such delivery.

21. DBS Privileged Service

21.1 We may (but shall not be obliged to) issue you with one or more card(s) and/or any other electronic or computerised token, device or gadget (collectively, the "**Privileged Card**") to enable you to access various privileged services (the "**Privileged Services**") that may be offered by us and/or other third party participating merchants (collectively, "**Participating Merchants**"). The Privileged Services shall be determined by us from time to time in our absolute discretion.

- Any Privileged Card issued to you remains our property and must be returned to us immediately upon our request. You may not transfer and/or assign the Privileged Card to any other person.
- 21.3 Your utilisation of the Privileged Services is subject to the Applicable Agreement and, where applicable, any terms and conditions that may be imposed by the Participating Merchants, and you agree to be bound by and to comply with the same.
- 21.4 We may, at any time, vary the eligibility criteria for the Privileged Card, suspend or withdraw all or any part of the Privileged Services, vary the frequency and manner of use of the Privileged Services and the Branches or other locations where such Privilege Services would be made available or add or remove Participating Merchants, without prior notice to or consent from you.

22. Appointment of Agents / Outsourcing

- 22.1 You agree that we shall have the right to outsource or sub-contract any part of our banking operations in connection with your Account and the Services offered by us to any third party (whether in Singapore or elsewhere) on such terms as we deem fit in our absolute discretion, without prior notice to or consent from you. Such operations shall include cheque clearing, creation, maintenance and archiving of documents and records, card production and mailing and insertion of security and user identification codes. You further agree that we may employ or use the services of any Agent for, and delegate to any such Agent, the performance of our duties and exercise of our rights in connection with your Investments, Transactions, Accounts and/or any of the Services offered to and utilised by you, upon such terms and conditions as we deem fit and you shall be bound by the same.
- We shall not be responsible for any act, omission, neglect or default of any such third party or Agent provided that such third party or Agent was selected and appointed by us in good faith.

23. Your Undertakings, Representations and Warranties

- 23.1 You hereby undertake, represent and warrant to us that:
 - (a) unless otherwise agreed to by us in writing, you do not and will not rely on us as an adviser or fiduciary for any purpose whatsoever; accordingly, you agree that any advice or recommendation that may be given by us to you at your request shall be given or made without any responsibility on our part and on the basis that you shall not rely on any such advice or recommendation for any purpose but shall make your own assessment and rely on your own judgment;
 - (b) before entering into any Investment or Transaction, you shall (and you are deemed to have done so should you enter into any Investment or Transaction):
 - (i) ensure that you fully understand the nature of such Investment or Transaction, the terms and conditions thereof, and the risks associated therewith;
 - (ii) seek independent legal, tax and financial advice if required; and
 - (iii) consider your specific financial needs and investment objectives and make your own assessment and rely on your own judgement as to whether such Investment or Transaction is suitable for you;
 - (c) unless we acknowledge otherwise in writing, you are entering into each and every Investment or Transaction as principal for your own account and not as agent for any person; accordingly, you shall be liable to us as principal for all obligations in respect of each Investment or Transaction;
 - (d) (where you are a corporation) you are a person, duly incorporated or otherwise properly constituted and validly existing under the laws of your jurisdiction of incorporation/constitution;
 - (e) you have full capacity, power and authority to enter into, perform and deliver, and have taken all necessary action to authorise your entry into, performance and delivery of, this Agreement and any Applicable Agreement;
 - (f) your execution and delivery of, and performance of your obligations under this Agreement and any Applicable Agreement does not and will not violate, contravene or, conflict with or constitute a default under any provision of your constitutional documents (where you are a corporation), Applicable Laws or any instrument binding on you or any of the Assets;

- (g) in entering into this Agreement, any Applicable Agreement and any Investment or Transaction, you have obtained and made, and you shall maintain in effect, all necessary authorisations, consents or approvals, exemptions, licenses, notifications and filings and you shall comply with all terms and Applicable Laws binding on you;
- (h) you are responsible for your own tax affairs and for ensuring compliance with your own tax obligations;
 - (ii) you have complied with all Applicable Laws on tax binding on you;
 - (iii) you have not committed or been convicted of any serious tax crimes or been subject to any investigation or criminal proceedings, whether in Singapore or elsewhere, in relation to tax matters; and
 - (iv) none of your Assets constitute proceeds from serious tax crimes;
- (i) your purpose for opening your Account is not illegitimate; you shall not use your Account as a platform for illegal tax activities and you are aware of Singapore's firm stance against illegal or illicit tax activities;
- (j) the obligations expressed to be assumed by you in or under this Agreement and any Applicable Agreement are legal, valid, binding and enforceable obligations;
- (k) except as otherwise notified to us in writing, you are the legal and beneficial owner of the Assets, free and clear of any lien, charge or other encumbrance and/or restriction as to title and transferability and no person other than you has, or will have or acquire any beneficial or other interest in or security or other rights over your Account or any of the Assets without our prior written consent;
- (l) you shall not, without our prior written consent, create or permit the existence of any lien, charge, mortgage or other encumbrance, security interest and/or restriction over any Asset except in favour of us or as imposed by Applicable Laws;
- (m) you shall promptly provide and/or execute such documents and do such acts or deeds, at your own costs, as may be required by us at any time in connection with this Agreement, any Applicable Agreement or any Investment or Transaction;
- (n) you shall promptly notify us in writing of any change:
 - (i) in the beneficial ownership of the Assets;
 - (ii) (where applicable) in your constitution, shareholders, partners, directors or company secretary, or the nature of your business;
 - (iii) in your particulars, circumstances, status or of any material information submitted to us, including any change in citizenship, residence, tax residency, address(es) on record, telephone and facsimile numbers and email addresses; and
 - (iv) to your Authorised Agent, your signature or signing requirements in respect of your Account;
- (o) you shall cooperate fully in respect of any enquiry that we may make for the purposes of compliance with any Applicable Law (including the United States Foreign Account Tax Compliance Act and/or any other reporting and/or withholding tax requirements of any government); without prejudice to the generality of the forgoing, you shall provide all relevant information, details and/or documents as may be necessary to enable us to comply with any such Applicable Law;
- (p) there is no legal or other proceedings initiated or threatened against you, and no meeting has been convened for your bankruptcy, dissolution, liquidation, winding-up, termination of existence or reorganisation, or for the appointment of a receiver, manager (judicial or otherwise), trustee or similar officer of you or over any part of your assets and you have not made or proposed to make any arrangement, composition with, or any assignment for the benefit of your creditors;
- (q) there is no pending action, suit or proceedings at law or in equity (whether in Singapore or elsewhere) before any court, tribunal, governmental body, agency, official or any arbitrator that is likely to affect the legality, validity or enforceability against you of or your ability to perform the obligations under this Agreement or any Applicable Agreement;

- (r) you shall ensure that your obligations and indebtedness to us are direct, unconditional and unsubordinated and will at all times rank at least pari passu with all your other present and future unsecured and unsubordinated obligations (except for such obligations mandatorily preferred by law);
- (s) you shall pay on demand all sums that may be due from you to us howsoever incurred;
- (t) you shall promptly notify us of the occurrence of any Event of Default or any event which may potentially constitute an Event of Default;
- (u) you have taken all necessary precautions to ensure that all bills, cheques and/or financial instruments which you have presented or may present to us for any purpose whatsoever are authentic and all endorsements in relation thereto are regular;
- (v) all documents, records, statements and information submitted by you to us for any purpose whatsoever are true and accurate and we are entitled to rely on the same in our private banking relationship with you and you shall provide us with such other information or documents or assistance as we may reasonably require from time to time in connection with this Agreement and/or any Applicable Agreement and/or to comply with any Applicable Law; and
- (w) you are subject to civil and commercial law and to legal proceedings; and you shall not claim any immunity or privilege from any set-off, suit, judgment, execution, attachment or any other legal process.
- Each of the undertakings, warranties and representations set out in this Clause 23 shall continue to have full force and effect for so long as this Agreement and/or any Applicable Agreement remains in force and shall be deemed to be repeated by you each time you open an Account, utilise any Service or undertake any Investment or Transaction.

24. Collection, Processing, Use and Disclosure of Customer Data

- 24.1 (a) The DBS Privacy Policy (available at www.dbs.com/privacy) is incorporated by reference into and forms part of this Agreement. The DBS Privacy Policy shall apply to all Customer Data provided by you or otherwise collected by us from any other sources or in the course of your relationship with us or any of our Affiliates and you hereby consent to the collection, processing, use and disclosure of Customer Data in accordance therewith.
 - (b) If you provide us with Customer Data of another individual (including, where applicable, your directors, partners, Authorised Agents, shareholders and beneficial owners), you undertake, represent and warrant to us that you have obtained such individual's consent for, and hereby consent on behalf of such individual to, the collection, processing, use and disclosure of his/her Customer Data by us in accordance with the DBS Privacy Policy.
 - (c) In the event of any conflict or inconsistency between this Agreement and the DBS Privacy Policy, the former shall prevail.
- 24.2 Without prejudice to any provision in the DBS Privacy Policy, you agree that we and our Personnel may disclose any Customer Data to any of the following persons (whether in Singapore or elsewhere) for the purpose of establishing, maintaining and/or operating your Account, providing any Service to you, effecting your Instructions, including in relation to any Investment or Transaction, managing your banking relationship with us or any of our Affiliates and/or any other purpose connected or relevant to our business (including operating internal controls and complying with Applicable Laws):
 - (a) any Branch or any other office, representative office and related company of the Bank (including our Affiliates), whether in Singapore or elsewhere;
 - (b) any Agent, our insurers, professional advisers (including auditors and legal advisers), any person providing services to us or any other third party provider of services (including debt collection, printing, or mailing of cheque books, reports, newsletters or other documents, professional, management, administrative, data management, electronic, telecommunications, computer, payment, collection, security, investigation, clearing and credit reference or checking services) or to whom we have outsourced or sub-contracted any part of our banking operations, their subcontractors and any persons further engaged by any subcontractor to perform such Services or operations, and any person providing services to any of them;

- (c) any Exchange, clearing house or trade repository in connection with your Investments, Transactions, Accounts and/or the Services utilised by you;
- (d) the issuer and where applicable, the manager and trustee of any of your Investments;
- (e) any person with (or through whom) we enter into (or may potentially enter into) any transaction in connection with the purchase or sale of any credit insurance or any other contractual protection or hedging with respect to your obligations under any of your Investments, Transactions, Accounts or the Services utilised by you;
- (f) any court or tribunal, government, quasi-government, regulatory, fiscal, monetary or other authority, agency, body or person, whether in Singapore or elsewhere, where such disclosure is required by Applicable Laws (including Applicable Laws on anti money-laundering or which impose any reporting and/or withholding obligations on us, such as the United States Foreign Account Tax Compliance Act) or pursuant to any order of court or tribunal or any code or guideline not having the force of law but with which we generally comply;
- (g) any credit bureau approved by the relevant authorities and any member or subscriber of such credit bureau;
- (h) any person who provides introducing or referral services to us or to whom we provide introductions or referrals;
- (i) any person to whom any fee, commission or other amount may be payable, for the exclusive purpose of determining the quantum of such fee, commission or other amount;
- (j) any insurer, guarantor and/or any Security Provider in relation to any of your Investments, Transactions, Accounts or any of the Services utilised by you;
- (k) any actual or potential assignee or transferee of any of our rights and obligations or other actual or potential participant or sub-participant of any of our rights and/or obligations under or relating to any of your Investments, Transactions, Accounts or any of the Services utilised by you;
- (l) any person in connection with the marketing or promotion of any Service or product offered by us or such person or investigating any complaint or dealing with any query relating to the marketing or promotion of any such Service or product;
- (m) any person whom we believe in good faith to be you or your Authorised Agent;
- (n) any of your auditors and where applicable, your directors, shareholders and partners;
- (o) where you are acting as trustee of a trust, the beneficiary(ies) and/or settlor(s) of the trust;
- (p) the other Joint Account Holder(s) of a Joint Account;
- (q) where you are a Minor, your parent or legal guardian;
- (r) any person to whom, in our opinion, disclosure is required for the proper administration and operation of your Account or the execution of any Investment or Transaction or the provision of any Service;
- (s) any person as you or your legal or personal representatives may, from time to time, expressly authorise us in writing and in such case, such authorisations shall remain valid and in effect until we receive written revocation of such authorisation from you or your legal or personal representatives;
- (t) any person or organisation to clarify or correct any wrongful or erroneous belief, representation or allegation to any third party, whether made by you or on your behalf, both in public and in private, regarding any of the Bank's dealings with you or otherwise in relation to the Bank's products, processes or policies, regardless of the form of media or platform which may include but is not limited to, published articles, posts, complaints or petitions; and
- (u) any person to whom we are under a duty to disclose or we consider in good faith is in our interest to make such disclosure.

- 24.3 In the event that you instruct us to send funds by wire transfer to a bank or financial institution (the "Beneficiary Institution"), whether in Singapore or elsewhere, you acknowledge that we may be required to disclose certain Customer Data, including your name, Account number, address, unique identification number and the date and place of birth, to the Beneficiary Institution and you consent to such disclosure.
- 24.4 You acknowledge and accept that any Customer Data disclosed by us may be subject to further disclosure by the recipient to other parties whether in accordance with the laws of the country/region in which the recipient is located or otherwise. Such laws may be wider in scope and implemented under less restrictive terms than would otherwise be the case in Singapore. You agree that we shall not be liable for any Losses sustained and/or incurred by you by reason of or in connection with such further disclosure by the recipient.
- 24.5 For the avoidance of doubt, (a) any consent given by you in relation to the collection, use, processing and disclosure of Customer Data shall continue notwithstanding your death or incapacity, the termination of this Agreement or any Applicable Agreement or the closure of any of your Accounts and (b) the rights conferred on us in this Clause 24 and/or the DBS Privacy Policy are in addition to, and shall not prejudice, any other rights that we may have under Applicable Laws or any Applicable Agreement.

25. Events of Default and Rights Upon Events of Default

- 25.1 The occurrence of any of the following shall constitute an Event of Default:
 - (a) you fail to pay any amount which is due to us on the due date of payment or, if payable on demand, on demand by us;
 - (b) you fail to perform or comply with any of your obligations under this Agreement and/or any Applicable Agreement;
 - (c) any representation and/or warranty made by you to us, whether contained in this Agreement and/or any Applicable Agreement or otherwise, shall prove to be false, incorrect or misleading in any material aspect;
 - (d) as the case may be, you become bankrupt or insolvent or unable to pay your debts as they become due or any bankruptcy application, judicial management application, winding up application or other insolvency application has been presented against you or a resolution has been passed for you to be wound up or placed under judicial management;
 - (e) any attachment, sequestration, distress, execution or other legal process is issued or levied or an administrator, receiver, judicial manager, trustee-in-bankruptcy, custodian or other similar person has been appointed (or an application for the appointment of any such person has been presented) in respect of you or any of your assets;
 - (f) you enter into or propose or make any arrangement, composition with, or any assignment for the benefit of your creditors;
 - (g) we are given notice of a garnishee order and/or injunction and/or similar order in respect of your Account and/or any of your Assets and/or any information which we reasonably believe would likely adversely affect your private banking relationship with us, whether in Singapore or elsewhere;
 - (h) any legal proceeding, suit or action of any kind whatsoever (whether criminal or civil) is instituted against you and we are of the opinion that it will or could materially and adversely affect your ability to perform and observe your obligations under this Agreement and/or any Applicable Agreement;
 - (i) as the case may be, you pass away or become, in our view, incapable of managing your affairs, whether by reason of mental incapacity or otherwise;
 - (j) you are or become, or are or become associated with, or any Asset is or becomes associated with, an individual and/or entity named in any lists (including and not limited to the Specially Designated Nationals and Blocked Persons List administered by the United States office of Foreign Assets Control) or falls within the scope of any sanctions, freezing, anti-terrorism or other programs enforced and administered by the relevant regulatory authorities or bodies, whether in Singapore or elsewhere, and whether those are targeted or general;
 - (k) at any time and in our absolute discretion, we consider that the continuation of any Investment, Transaction, Account and/or Service would not be in the Bank's interest or would be inconsistent with prudent banking practice whether in Singapore or elsewhere;

- (l) any consent, approval or authorisation referred to in this Agreement and/or any Applicable Agreement is not granted or ceases to be in full force and effect;
- (m) any event occurs, which in our opinion, may adversely affect your financial condition or standing;
- (n) if circumstances arise or continue (or which we deem in our reasonable opinion and discretion to be the case) which render it unlawful for either us or you to perform or comply with any of the obligations under this Agreement, any Applicable Agreement, and/or any agreement between us and an Agent, and/or place or may place us in jeopardy; or
- (o) an event of default (howsoever described) occurs under any of the Specific Terms or any Applicable Agreement.
- 25.2 If an Event of Default occurs, we may, with or without notice to you and without prejudice to any other claim or right whatsoever which we may have, take any of the actions set out below (which may result in Losses to you):
 - (a) close, suspend and/or terminate any Account and/or Service;
 - (b) declare all or any Liabilities immediately due and payable whereupon such Liabilities shall become due and payable in the currency in which they are denominated;
 - (c) cease to comply with all or any Instructions;
 - (d) take such action as we think necessary or appropriate to cancel, settle, redeem or terminate any outstanding Transactions between us as principal or with any third party entered into by us as your agent;
 - (e) sell, realise, assign, transfer or otherwise dispose of any of your Investments or Assets held by us in such manner and on such terms as we may think fit and to apply the net proceeds thereof (after deduction of any expense incurred in connection therewith) in or towards satisfaction of your Liabilities to us; and/or
 - (f) exercise our rights of appropriation, debit, set-off and/or consolidation of accounts in accordance with Clause 28.
- 25.3 Notwithstanding and without prejudice to Clause 25.2 above, upon the occurrence of an Event of Default, we shall be deemed to have opened a new Account in your name and all payments made by you or any third party on your behalf to us shall be treated as having been credited to the new Account and shall not operate to reduce any amount due and to be recovered from you under this Agreement and/or any Applicable Agreement.
- 25.4 You irrevocably and unconditionally appoint us to be your attorney (with full powers of substitution), in your name or otherwise on your behalf, and as your act and deed, to sign, seal, execute and deliver all deeds, instruments, agreements and any other documents and to do all acts and things which may be required or which we shall consider expedient for the purpose of exercising any of our rights and powers under this Clause 25.

26. Suspension of Accounts and Services

- We may, in our absolute discretion, without assigning any reason whatsoever suspend and/or freeze the operations of any or all of your Accounts and provision of any Service at any time and for such duration as we deem fit, which may result in Losses to you, for any reason whatsoever including situations where:
 - (a) an Event of Force Majeure has occurred;
 - (b) an Event of Default has occurred;
 - (c) we received Instructions from you, your Authorised Agent and/or one or more Joint Account Holders (as the case may be), to suspend the operation of any of your Accounts and/or the provision of any Service offered to you;
 - (d) we receive contradictory Instructions from you, one or more Joint Account Holders, your directors (in the case of corporations), partners (in the case of partnerships) and/or your Authorised Agent;
 - (e) we are notified or become aware of any dispute between you, one or more of the Joint Account Holders, your directors (in the case of corporations), partners (in the case of partnerships) and/or your Authorised Agent;

- (f) we have reasonable grounds to believe, based on information gathered or received by us, that any of your Accounts, Investments or Transactions is or was conducted for an illegal purpose or if we are of the opinion that it is advisable to freeze and/or suspend the operations of any of your Accounts and/or the provision of any Service made available to you to safeguard our interest; and/or
- (g) any Applicable Law so requires.
- 26.2 Without prejudice to Clause 26.1 above, we reserve the right to impose such conditions in relation to the further operation of your Account and/or continued utilisation of any Service as we may determine in our absolute discretion.

27. Closure or Transfer of Accounts, Termination of Services and Agreement

- 27.1 Subject to Clause 3.2, the provisions of the Specific Terms and/or any Applicable Agreement, you may request to close your Account and/or terminate any Service at any time by giving us Instructions in accordance with this Agreement. Where we accept your instructions to close your Account, you agree to do all such things and/or execute all such documents as we may require to facilitate such closure and/or termination, including giving Instructions for the termination, closing out and/or liquidation of your Investments and Transactions and making payment of all Liabilities in respect of such Account or Service, and we are entitled to effect the closure of your Account within such timeframe as we may deem fit.
- Subject to Clause 25 and the provisions of the Specific Terms and/or any Applicable Agreement, we may at any time, without disclosing any reason, close any of your Accounts or transfer such Account to DBS Treasures Private Client, DBS Treasures or other business segments (as shall be determined by us) and/or terminate any Service made available to you by giving not less than seven (7) days written notice to you or, where we determine that it is not practicable to give seven (7) days prior written notice, immediately upon issuing a notice to you. In the absence of any Instruction from you in due time, you agree and authorise us, at your cost and expense, to do all such things as are necessary to facilitate the closure or transfer of such Account and/or termination of the Service, including terminating, closing out and/or liquidating your Investments and Transactions, debiting from your Account all Liabilities in respect of such Account or Service and converting the Assets into monies in such manner as we consider appropriate.
- 27.3 Without prejudice to Clause 27.2, if:
 - (a) the balance in any of your Accounts falls below any minimum amount prescribed by us;
 - (b) there is no activity on any of your Accounts for an extended period (the duration of which shall be determined by us from time to time in our absolute discretion); or
 - (c) you do not qualify as an accredited investor or do not wish for us to treat you as an accredited investor,

we may close such Account and/or terminate any Service made available to you without further notice to you. For the purpose of this Clause 27.3, "activity" excludes interest crediting, enquiries and debiting of any charge, fee and interest.

- 27.4 Upon the closure of any of your Accounts,
 - (a) all Services linked to that Account will be terminated automatically; and
 - (b) you shall return to us all property belonging to us, including the Privileged Card.
- 27.5 Without limitation to the generality of the foregoing, we may, upon the closure of any of your Accounts, discharge our entire liability with respect to such Account by issuing to you a draft or cheque in the currency of the Account without recourse to us as drawer, payable to your order, in the amount of the credit balance in the Account as at the relevant date, together with such other documents (if any) as may be necessary to transfer to you such claims as we may have on such funds.
- 27.6 For the avoidance of doubt, the closure or transfer of your Account and/or the termination of any Service are without prejudice to any right, remedy or obligation which have accrued or are still accruing and shall not affect the continued operation, validity, enforceability and/or applicability of this Agreement.
- 27.7 This Agreement shall terminate automatically when all of your Accounts have been closed, all Services made available to you have been terminated and all Liabilities have been paid in full. Termination of this Agreement shall not affect the provisions relating to indemnities, disclosure of information and our rights, powers and benefits as set out herein.

28. Rights of Appropriation, Debit, Set-off and/or Consolidation

- In addition and without prejudice to any charge, lien or right to which we may be entitled by law or otherwise, we may, in our discretion, at any time and without prior notice to you:
 - (a) appropriate, set-off or debit all or part of the credit balances in any Bank Account or to which you are beneficially entitled (notwithstanding that any Deposit has not matured) and apply the same towards payment and satisfaction of all or any part of your Liabilities to us;
 - (b) (where the amount of a Liability is unascertained) in good faith estimate such amount and exercise our rights as set out in Clause 28.1(a) in respect of such estimate;
 - (c) for the purpose of Clause 28.1(a) above, sell by public or private sale any of your Assets and set off the proceeds against all or part of your Liabilities to us; and
 - (d) (where any of your Liabilities is in a different currency from the credit balances in any of your Bank Accounts or to which you are beneficially entitled) make the necessary conversion at such currency exchange rate as we may determine in order to exercise our rights as set out in Clause 28.1(a).
- 28.2 Our rights pursuant to this Clause 28 subsist for as long as any Liabilities are due from you to us.
- 28.3 Any exercise of our rights in accordance with this Clause 28 shall not be treated as a payment of the amount due (except to the extent of any amount standing to the credit of any of your Bank Accounts) or a waiver of any Event of Default.
- 28.4 In addition and without prejudice to our general right of set-off under law, under this Agreement or otherwise, we are deemed to have exercised our right of set-off upon the occurrence of the following events:
 - (a) upon the crystallisation of any floating charge created by you over your property, assets or undertakings, the presentation of a bankruptcy, winding up or other insolvency application, an application for the appointment of a judicial manager or similar officer against you, or other similar process or the passing of a resolution to effect the same; and
 - (b) the issue of any execution or levy of any execution on any of your Bank Accounts whether by us or other parties.
- 28.5 Without prejudice to the foregoing, we may, in our absolute discretion, aggregate the amount payable to or from you under this Agreement and/or any Applicable Agreement on any date and only the net amount payable by or to you shall be made.
- 28.6 If any security or payment to us is avoided or reduced by virtue of any Applicable Law relating to insolvency for the time being in force, any settlement, assignment, payment, release or discharge between us shall be wholly void and we shall be entitled to exercise all of our rights against you as if such settlement, assignment, payment, release or discharge had never been granted, given or made.

29. Lien

- 29.1 In addition and without prejudice to any other rights to which we may be entitled by law or otherwise, we shall, for so long as any of your Liabilities remain outstanding, have a right of lien over all Assets held from time to time by us, our Agents or Nominees, in your name, or on your behalf, including Assets in transit to us, our Agents or Nominees. Immediately upon default by you, we shall be entitled, without further notice to you, to dispose, either by enforced sale or in the open market, any such Assets and to apply the proceeds towards payment and satisfaction of all or any part of your Liabilities to us (which shall, for the avoidance of doubt, include all costs, expenses and charges incurred in connection with such disposal).
- 29.2 For so long as any of your Liabilities remain outstanding, we reserve the right to decline any request by you to withdraw any of the Assets held with us, our Agents or Nominees.
- 29.3 Without prejudice to any provision in this Agreement, you agree to perform all such acts and/or execute all such documents as may be necessary for the purpose of maintaining, protecting or perfecting our right of lien over the Assets.
- 29.4 You agree and acknowledge that our Agents and Nominees may also claim a lien, right of retention or sale over any of the Assets held by it.

30. Charge

- 30.1 In consideration of us agreeing to or continuing to
 - (a) make available Credit Facilities or other Services or accommodation whatsoever, including to giving time and indulgence, to you; and/or
 - (b) enter into Transactions with you, you hereby charge, pledge, mortgage, assign and otherwise create a first fixed charge to and in favour of us over all your Assets as a continuing security for the payment and discharge of your Liabilities.

The charge created hereunder is in addition to, and independent of, any charge, guarantee or other security or right or remedy now or at any time hereafter held by or available to us.

Without prejudice to any provision in this Agreement, you agree to perform all such acts and/or execute all such documents as may be necessary for the purpose of perfecting the charge created under this Clause 30.

31. Indemnity and Exclusion of Liability

- 31.1 Save for Losses arising directly from our gross negligence, fraud or wilful default, we shall not be responsible or liable to you for any Losses suffered or incurred by you however caused in connection with any of your Investment, Transaction, Account and/or pursuant to the Services made available to you, including any Losses arising from or in connection with:
 - (a) your default or negligence;
 - (b) your or your Authorised Agent's failure to comply with or fulfil any of your obligations and/or warranties, to the extent that such breach or failure interferes directly or indirectly with our performance of our obligations under this Agreement and/or any Applicable Agreement;
 - (c) any liability for tax or similar charges payable in connection with any Investment or Transaction or arising from the utilisation of any of the Services offered by us to you;
 - (d) us acting upon any financial instrument, notice, resolution, request, certificate, report, or other document which we, in good faith, believed to be genuine and properly executed or any Instruction from you and/or your Authorised Agent;
 - (e) any falsification of identity or faulty identification which could not have been detected despite us exercising due care in verifying the identity of the party we are dealing with;
 - (f) our refusal to accept or act on any of your Instructions in accordance with this Agreement;
 - (g) any loss or destruction of cheques and/or other financial instruments or delay in presentation thereof;
 - (h) any debits or revisions of the amount credited to your Account;
 - (i) any delay or failure in performing any of our duties or other obligations resulting wholly or partly from any action taken by us for the purpose of complying with Applicable Laws;
 - (j) any irregularity, inaccuracy, omission, incorrect entry, error and/or unauthorised transaction in any Account Statement or Transaction Record or any loss of, destruction of or error in any other records save as expressly provided for in this Agreement;
 - (k) the suspension, closure or termination of any of your Accounts or any Service made available to you or your Authorised Agent;
 - (l) the use of postal services, facsimile, telegraph, telephone, telex, SMS, email, electronic platforms or any other electronic and non-electronic means of communications with you.
 - (m) any delay, interception, loss or failure in the delivery, transmission or dispatch of any Communication to you or if any Communication is not sent in accordance with this Agreement or if the content of any Communication is disclosed to any third party during transit or any matter arising from the use of the Hold Mail Service;
 - (n) our disclosure of Customer Data in accordance with this Agreement, the DBS Privacy Policy, any Applicable Agreement and/or Applicable Law;

- (o) any information, advice or opinion given by us or our Personnel to you, whether or not provided at your request or relied upon;
- (p) any mutilation, interruption, omission, failure, error or delay in the issue or remittance of drafts or other financial instruments due to any reason whatsoever and whether arising in Singapore or elsewhere;
- (q) the exercise or non-exercise of any power or discretion conferred upon us under this Agreement and/or any Applicable Agreement;
- (r) any Event of Force Majeure;
- (s) any constraint which we may face (through no fault of ours) in executing any of your Instructions;
- (t) any action taken or omission by us pending the completion of our updates of our records of your particulars and/or information in accordance with this Agreement, including processing your Instructions in accordance with the mandate for the time being in force or sending any Communication to your Mailing Address last known address, facsimile or email address in our records as you had directed for any Communication to be sent prior to our receipt of your written notice of change; and
- (u) our enforcing or attempting to enforce or protect any right, power or remedy which we may have against you pursuant to this Agreement and/or any Applicable Agreement.
- 31.2 Without prejudice to the foregoing, you agree to indemnify and to hold us, our Personnel, Affiliates, Nominees and Agents harmless from and against any Losses, save for Losses arising directly from our or their gross negligence, fraud or wilful default, which we may suffer or incur in connection with any of your Investment, Transaction, Account, and/or our provision of any Service to you and/or your utilisation of the same, including any Losses arising from or in connection with:
 - (a) us acting upon or carrying out, in good faith, any Instruction purportedly given by you or your Authorised Agent, notwithstanding that these Instructions may not be authorised, accurate or complete;
 - (b) the operation, maintenance or closure of any of your Accounts;
 - (c) our Communications with you by any mode of transmission;
 - (d) (i) the collection of any cheque, bill, note, draft, dividend, warrant or other instrument presented by you for collection, (ii) the guaranteeing of any endorsement or discharge of the same and/or (iii) our acting in reliance on your guarantee of the regularity of all endorsements and authenticity of all signatures on all bills and cheques and/or financial instruments which you may present to us;
 - (e) the use of any system or electronic and non-electronic means of transmission, communication, transportation or otherwise in carrying out your Instructions (including, any loss, delay, misunderstanding, mistake, distortion or duplication arising therefrom or in connection therewith);
 - (f) our involvement (directly or otherwise) in any proceeding (whether in or out of Singapore) of whatever nature in connection with your Investments, Transactions, Account and/or any Service offered to you;
 - (g) the preparation of any document or agreement necessary to facilitate any Investment or Transaction, or providing any Service as may be requested by you from time to time or as we deem advisable in our absolute discretion;
 - (h) our disclosure of Customer Data in accordance with this Agreement, any Applicable Agreement and/or Applicable Law;
 - (i) any exercise of our rights of appropriation, debit, set-off and/or consolidation of accounts in accordance with Clause 28;
 - (j) any Event of Force Majeure;
 - (k) where you are acting as trustee of a trust, any dispute between the beneficiaries and yourself;
 - (l) any breach by you of any of the terms in this Agreement and/or any Applicable Agreement; and

- (m) enforcement or in contemplation of the enforcement or protection of any of our rights or resolution of any dispute (whether by judicial proceedings or otherwise) relating to the matters covered under this Agreement and/or any Applicable Agreement.
- 31.3 You acknowledge and agree that we may, from time to time, be required to make certain representations and/or warranties and/or accept sole and principal responsibility or liability to other third parties in respect of any Investment or Transaction entered into by us on your behalf. In such cases, you shall indemnify and hold us harmless against any and all actions taken by us (which we determine to be necessary, desirable or expedient) to ensure that we will not be in breach of our said principal responsibility or liability.
- 31.4 The indemnities in this Clause 31 constitute a separate and independent obligation from the other obligations in this Agreement, shall give rise to a separate and independent cause of action, shall apply irrespective of any indulgence granted by us and shall continue in full force and effect despite any judgment, order, claim or proof for a liquidated amount in respect of any sum due under this Agreement or any judgment or order. For the avoidance of doubt, the indemnities in this Clause 31 shall not in any way be prejudiced or affected by the closure of any of your Accounts and/or the termination of this Agreement.

32. Extraordinary Event

Without prejudice to any provision in this Agreement and/or any Applicable Agreement, upon the occurrence of any Extraordinary Event, we may in our absolute discretion determine any adjustment or action necessary in relation to any of your Investments, Transactions, Accounts and/or Services. Such adjustments or actions may include altering or varying the quantities of currencies, Securities or commodities or instruments or the exchange rates or specifications of currencies, Securities or commodities or instruments bought or sold pursuant to any Transaction, terminating any Transaction, or making payments, or converting the currency of your Accounts, in or to a currency which is a freely transferable currency at that time, at such rate of exchange as we may, in good faith, deem appropriate, and you agree to be bound by such adjustment or action.

33. Event of Force Majeure

Upon the occurrence of any Event of Force Majeure, all our obligations under this Agreement and the performance thereof shall be excused and/or suspended to the extent that the discharge and fulfilment of such obligations are prevented, frustrated, hindered or impeded as a consequence of any such Event of Force Majeure. The occurrence of any Event of Force Majeure shall not affect any right and obligation that have accrued or are accruing under this Agreement.

34. Change in Status Etc.

Your obligations and Liabilities under this Agreement and/or any Applicable Agreement shall continue to be in full force and effect and be binding on you notwithstanding:

- (a) (where you are an individual) your death, incapacity (whether mental incapacity or otherwise), bankruptcy or other legal disability;
- (b) (where you are a corporation) your winding up, insolvency, dissolution or other legal disability;
- (c) (where you are a corporation) any change by amalgamation, reconstruction or otherwise which may be made to your constitution; and
- (d) any change by amalgamation, reconstruction or otherwise which may be made to the constitution of the Bank and any sale of all or any part of our undertaking and assets to another person, and all our rights and interests under this Agreement and/or any Applicable Agreement may then be enforced by any such amalgamated or reconstructed company or person as if it had been named herein instead of us.

35. Entire Agreement

This Agreement constitutes the entire agreement between us in relation to the matters covered or dealt with under this Agreement or any part thereof. Any previous agreement and understanding in relation to the matters covered or dealt with in this Agreement are hereby superseded.

36. Waiver

36.1 Save as otherwise provided, time shall be of the essence in the performance of all obligations under this Agreement.

- Without prejudice to Clause 36.1 above, we may (but shall not be obliged to) grant time or other indulgence to you or any other person provided always that:
 - (a) any delay or omission by us in exercising or enforcing our rights, powers or remedies under this Agreement and/or any Applicable Law shall not operate as a waiver thereof or prejudice, affect or impair the exercise or enforcement of our rights, powers or remedies at any time thereafter; and
 - (b) any single or partial exercise or enforcement of any right, remedy or power shall not preclude the subsequent exercise or enforcement thereof or the exercise of any other right, remedy or power.
- Without limiting the foregoing, no waiver by us of any breach of any provision of this Agreement, whether unconditionally or otherwise, shall be deemed to be a waiver of any subsequent breach of that or any other provision of this Agreement.
- The rights, powers and remedies provided under this Agreement are cumulative and not exclusive of any right or remedy which we may have under any other agreements and/or provided by Applicable Laws.
- Without prejudice to the generality of the foregoing, our rights under this Agreement are in addition to and do not prejudice or adversely affect and are not prejudiced or adversely affected by any collateral or other security or right or any judgment or order obtained by us for all or any part of your Liabilities.
- We shall not be obliged to resort or seek to enforce any security or personal guarantee or liability of any other person and we shall have full power, in our absolute discretion, to give time for payment or grant any other indulgence to or make any other arrangement with any person without prejudice to your Liability under this Agreement and/or any Applicable Agreement.

37. Notices

- Without prejudice to our rights to utilise any other effective mode of communications, we may send any Communication to you by post, facsimile, electronic mail or other electronic transmission, hand-delivery or such other manner as we deem fit to your Mailing Address, facsimile numbers, email address or other relevant numbers in our records.
- 37.2 Such Communications shall be deemed to have been received by you:
 - (a) on the day it was delivered personally;
 - (b) on the day it was transmitted by facsimile, telex, cable or electronic transmission whether or not it was actually received by you;
 - (c) two (2) days or seven (7) days after the day of posting it to your Mailing Address in Singapore or outside Singapore respectively, notwithstanding that the Communications may be returned undelivered.

and shall be effective on and from the date of deemed receipt, or the date (if any) as specified in the Communications.

37.3 In respect of any Communication which is returned undelivered, we reserve the right (but shall not be obliged) to dispose of such Communication within one (1) month after it is returned to us and to stop sending you any Communication until you update your particulars and information with us.

38. Illegality or Unenforceability of Provisions

- Notwithstanding any provision to the contrary in this Agreement, if it shall become (or it shall appear to us that it has or will become) unlawful or otherwise prohibited for us to maintain or give effect to any of our obligations and/or enforce any of our rights under this Agreement, we shall thereupon notify you to that effect and we shall be entitled to exercise all or any of our rights under Clause 27.
- If at any time, any provision in this Agreement is or becomes invalid, unlawful or unenforceable under the laws of any jurisdiction, neither the validity, legality and enforceability of (a) the same provision under the laws of any other jurisdiction or (b) the remaining provisions in this Agreement shall in any way be affected or impaired.

39. Amendment of Terms and Conditions

We may, by notice in writing, supplement, vary and/or modify the terms of this Agreement at any time and such supplement, variation and/or modification shall take effect from the date specified by us in the notice

- (which shall be binding upon receipt or deemed receipt by you). Such notice may be given to you through, or by publication of the supplement, variation and/or modification on, our website at www.dbs.com.sg.
- If you do not accept any such supplement, variation and/or modification, you shall immediately discontinue operating your Account and/or utilising the Services provided by us and promptly close your Account and terminate this Agreement. If you continue to operate your Account and/or utilise the Services provided by us after such written notification, you are deemed to have agreed to such supplement, deletion, variation and/or modification without reservation.

40. Assignment/Transfer

- 40.1 This Agreement is binding and enures to our mutual benefit and that of our respective successors and assigns.
- 40.2 In the absence of our prior written consent, you shall not charge, assign or transfer to any third party whether by security or otherwise your rights, benefits or obligations under this Agreement.
- 40.3 You further agree that we may, at any time, assign or transfer all or any part of our rights and obligations under this Agreement and may deliver all or any of the Assets held by us as security to our transferees, who shall then become vested with all the powers and rights in this Agreement and/or the Assets so transferred, and we shall thereafter be relieved and fully discharged from any liability or responsibility thereto.

41. The Contracts (Rights of Third Parties) Act

- 41.1 Unless expressly provided to the contrary in this Agreement, a person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act to enforce any term set out in this Agreement.
- 41.2 Notwithstanding Clause 40.1 above, it shall be the intention of parties that all defences and limitations in this Agreement shall be enforceable by all of the Bank's Affiliates, Branches, entities into or with which the Bank may merge or consolidate, any entity formed as a result of acquisition by or of the Bank, as well as our Personnel, all of whom or which shall be deemed third parties under the Contracts (Rights of Third Parties) Act. For the purposes of the Contracts (Rights of Third Parties) Act, consent of any of such third parties is not required for any variation, rescission or termination of this Agreement.

42. Applicable Laws and Jurisdiction

- 42.1 This Agreement and all relations between you and us shall be governed by and construed in all respects in accordance with the laws of Singapore and you irrevocably submit to the non-exclusive jurisdiction of the courts of Singapore.
- 42.2 Notwithstanding Clause 41.1 above, you agree that we shall be at liberty to initiate and take any legal action or proceeding in connection with the matters referred to in this Agreement or any dispute arising in relation thereto in any other court of competent jurisdiction and the initiation and taking of legal action or proceedings in any one or more jurisdictions shall not preclude us from taking legal action or proceedings in any other jurisdiction whether concurrently or not. You further agree to waive any objection that any legal action or proceeding commenced in the courts of any jurisdiction have been brought in an inconvenient forum and/or that such courts do not have jurisdiction over such legal action or proceeding and/or other legal actions or proceedings have been brought in another forum.
- 42.3 You further agree that the service of any legal process (including a writ of summons, summons or any originating process) on you may be effected by leaving the same at, or sending the same by registered mail to the Mailing Address. Such service of legal process or originating process shall be deemed to be good and effectual service of legal process on you if sent by post, on the date immediately following the date of posting and if served by personal delivery, on the date of leaving or delivery at the Mailing Address. Such service shall be deemed to be proper service of the legal process even though the legal process is returned undelivered. Notwithstanding the foregoing, we shall be entitled to effect service of legal process in any other manner permitted by law.
- 42.4 You further undertake that where you do not have an address in Singapore, you shall, if required by us, nominate a process agent, at your expense, with an address in Singapore to accept service of any legal process in Singapore on your behalf. The process agent shall acknowledge its appointment as such agent to us and service of legal process on such process agent shall be deemed to constitute good and effectual service of legal process on you.

PART C - GENERAL RISK DISCLOSURE STATEMENT

- 1. This General Risk Disclosure Statement forms an integral part of this Agreement. Please note that this is an important document which you should read carefully. Unless otherwise defined in this General Risk Disclosure Statement, terms and references defined in Part A of this Agreement shall have the same meanings when used in this General Risk Disclosure Statement.
- 2. The objective of this General Risk Disclosure Statement is to explain to you briefly the nature and risks of the Investments and Transactions that you may undertake with us. In particular, you must be aware that the risk of loss in any Investment or Transaction can be substantial.
- 3. However, this General Risk Disclosure Statement does not purport to advise you of the suitability of any particular Investment or Transaction for your purpose and/or disclose or discuss all the risks and other significant aspects of any Investment or Transaction. You should carefully consider whether entering into any particular Investment or Transaction is suitable for you, your operations, business and organisation in light of your financial resources, experience, objectives for engaging in the Investments or Transactions, ability to bear risk and other relevant circumstances. It is your sole responsibility to make your own independent appraisal and investigation into the risks associated with the desired Investment or Transaction. Prior to entering into any particular Investment or Transaction, you should consult your own legal, tax, financial and other relevant professional advisers and ensure that you have sufficient investment knowledge and experience to make your own evaluation of the merits and risks of such Investment or Transaction.

Contractual Terms and Obligations

- 4. You agree that we are not obliged to give advice or make recommendations on any Investment or Transaction and notwithstanding that we may do so on your request or otherwise, any such advice or recommendation are given or made without any responsibility on our part and on the basis that you will nevertheless make your own assessment and rely on your own judgment.
- 5. Before entering into any Investment or Transaction, you should ensure that you understand fully:
 - (a) the nature and fundamentals of each Investment or Transaction and the market in which such Investment or Transaction (where applicable) is traded in;
 - (b) the legal terms and conditions and all of your rights and obligations, as set out in the Applicable Agreement for each Investment or Transaction, including:
 - (i) the terms as to price, tenor, expiration dates, restrictions on exercising an option (as the case may be) and other terms material to the Investment or Transaction;
 - (ii) any term describing risk factors, such as volatility, liquidity, inability to exit the Investment or Transaction before its scheduled maturity or expiry date; and
 - (iii) the circumstances under which you may become obliged to make or take delivery of the underlying interest of the Investment or Transaction;
 - (c) the legal risks surrounding the Investment or Transaction, including the circumstances under which the Investment or Transaction may be illegal, resulting in it being void and unenforceable;
 - (d) the extent of the economic and associated risks to which you are exposed to as a result of such Investment or Transaction (and to determine that such risk is suitable for you in light of your own specific experience in relation to that specific Investment or Transaction and your financial objectives, circumstances and resources);
 - (e) the regulatory and tax treatment of the Investment or Transaction (which can be complex); and
 - (f) the nature and scope of the relationship between yourself and us in respect of each Investment or Transaction undertaken by you.
- 6. We may (but shall not be obliged to) furnish you with term sheets or other materials describing the Investment or Transaction and setting out the material terms thereof or such other information as we may consider relevant in relation thereto. We may (but shall not be obliged to) provide you with a sensitivity analysis

illustration of your potential exposure to market movements. Any such sensitivity analysis is for illustration only and is not to be treated as our view on current or future market movements. You are advised to carry out your own analysis.

7. Investors are to understand fully the nature of each Investment or Transaction which would be set out in the various materials, including the issuer's Offering Documents.

DBS Private Bank may not necessarily be (and typically would not be) the issuer, and in most instances be merely an intermediary (for example, as a broker, distributor and/or seller of these products issued by third party issuers or by another division of the Bank which is acting in the capacity of an issuer). Investors should always refer to the issuer's Offering Documents for information on the terms that they deem material or key to their investment purpose. The Bank may also provide to you other documents such as term sheets, summaries and/or post-trade contract notes.

Investors are to request and obtain the Offering Documents prior to investing and assess holistically the risks as well as the nature and structure of the Transaction to be in line with the investor's personal and financial objectives.

Investors should also note that there may be differences in the Bank's presentation of certain terms or the use of terminology in the Bank's other documents as compared with the issuer's Offering Documents. Documentation issued by the Bank would typically have to be general in respect of an entire product category or to cater for the majority of products custodised / sold by the Bank and not necessarily specific to any particular issuer.

For purposes of illustration, certain products and/or investment structures may be deemed by third party issuers to be Shariah-compliant, and/or stated in the issuer documentation to be approved by relevant esteemed Shariah councils/advisors. The issuer's terms relating to these products would hence likely be Shariah specific, in accordance with Shariah principles, unlike the Bank's documents (such as transaction advices, summaries or commentaries) which would be general and unlikely to deploy specific Shariah terms. An investor would need to assess, where relevant, that the nature of the transaction, based on the terms and/or structure, would not detract from their individual standards of adherence to Shariah principles, and should not rely on the Bank's general documents.

- 8. Unless otherwise indicated by us, your Investments are not bank deposits and are not obligations of, or guaranteed by us or any of our Affiliates and are subject to all investment risks and possible loss of the principal amount invested by you. Past performance of the same or of similar investments is not an indication of, nor a guarantee of future performance of your Investment.
- 9. You should always ensure that you have in place proper and sufficient means of monitoring the various types of risks associated with any Investment or Transaction (which can be complex and substantial). We will not be undertaking such monitoring for you in any circumstance whatsoever.
- 10. The Bank is subject to laws, regulations, rules or guidelines. Such rules and guidelines include and are not limited to or in relation to dealing restrictions, which would lead to the Bank instituting internal processes such as blackout periods or grey-listed (restricted) counters disallowing the Bank from entering into transactions related thereto (whether as a principal, agent, nominee, originator or otherwise).

We apply a risk-based approach in assessing the rules. The way in which the Bank complies with such regulations, rules or guidelines may be viewed to be more prudent or stringent as compared to other institutions or compared with a prima facie interpretation (as each institution has its risk thresholds and compliance or operational reasons and policies). This may result in the Bank's inability to carry out certain transactions or to transact during certain periods of time for purposes of such compliance.

11. We are entitled to (but shall not be obliged to) act upon your Instructions in relation to any Investment or Transaction and you should not assume that we will warn you if your Instructions are ill-timed or inadvisable for any reason or if such Instructions are likely to result in Losses to you. We will not be liable for any advice or opinion given by any of our employees or agents with respect to any Investment or Transaction and you are taken to be acting in reliance upon your own judgment as to the merits and risk of entering into any Investment or Transaction.

12. You should be aware that we are not responsible for any Losses that may be suffered by you, arising from or in connection with movement in prices or exchange rates, errors or delays in the transmission of any Instruction from or to you or changes in any Applicable Law.

Potential Conflict of Interest

- 13. Regardless of whether you or any third party make a profit or loss from any Investment or Transaction, we may make a profit from any Investment or Transaction entered into with you or on your behalf.
- 14. You understand that we act simultaneously for a large number of customers as well as for our own account. Accordingly, conflicts of interest cannot be completely avoided.

You acknowledge that we and/or our Affiliates (including any of our internal divisions or segments) may, for example:

- (a) be the issuer of any Investment,
- (b) combine your orders with our own orders or the orders of other customers,
- (c) make Investments or effect Transactions for you through the agency of and/or with a counterparty which is a related organisation or a person otherwise associated with us,
- (d) have a position or a direct or indirect interest in any Investment or Transaction even if such position or interest is opposite to that taken by you,
- (e) have bought or sold any investment or entered into any transaction as principal or for other customers,
- (f) have other banking, advisory or other corporate relationships with companies whose investments are held for your account or are purchased and sold for you and our Personnel may be officers and directors of such companies; or
- (g) have other banking, advisory or other relationships with you or any of your other accounts where we may hence take on multiple roles in the various Transactions, Investments, Services and/or for your Account(s), and receive benefits for our roles.

Subject to any Applicable Law, we and/or our Affiliates shall not be liable to account or specifically disclose to you any profit, charge or remuneration made or received from any such Investment or Transaction or other connected investments or transactions.

- 15. Any "all-in" price provided by us for Investments or Transactions entered into with you as principal may include a sales and trading mark-up over the interbank price. Such sales and trading mark-up will not exceed the maximum dollar amount or percentage range detailed in our published fee and benefits schedule in force at the relevant time, unless you agree otherwise. If we are able to execute the Investment or Transaction at a better interbank price, you agree that we may retain the benefit from such price improvement provided that the sales and trading mark-up do not exceed the maximum amount or percentage range published in our fees and benefits schedule, or otherwise agreed with you.
- 16. Solely for purposes of illustration, clients may authorise us to provide discretionary portfolio management services for their account(s). Pursuant thereto, clients would grant to us discretionary powers to invest, and we may for various reasons choose to invest for your account in an Investment or Transaction that is arranged, distributed by and/or issued by another DBS entity/affiliate or division within DBS (including and not limited to DBS flagship products).

For certain of these Investments or Transactions, we may also be a distributor for such Investments to you and other clients. We may also be the investment adviser to the issuer (or such other external party) in some of these Investments or Transactions.

Concurrently, another division (or affiliate) of DBS may also be the issuer or manager or arranger. The rights and responsibilities of any parties including DBS would depend on and be subject to the relevant role undertaken in each respective transaction.

The list of activities and banking, advisory or corporate relationships is not exhaustive and DBS receives various fees and benefits for its roles.

- 17. You acknowledge that we may pay to, or receive from, any Agent fees, charges or other quantifiable benefits (such as rebates, commissions and retrocessions) in any form in respect of, or may profit or gain (whether in the form of a sales and trading mark-up or otherwise) from (a) any Investment or Transaction effected for or with you or (b) any hedge effected by us in connection with any Investment or Transaction or (c) any Service provided to you in our capacity as principal, trustee or agent. We will disclose such quantifiable benefits, profit or gain received by us in accordance with Applicable Laws and to the extent permitted by Applicable Laws, you agree that we shall not be liable to account to you for, and may retain, the same for our own account and benefit.
- 18. The Services provided by us to you are non-exclusive and we are under no obligation to account to you for any benefit received by providing any Service to other customers or to disclose to you any fact or thing which may come to our notice in the course of providing any Service to other customers or in any other capacity or in any manner whatsoever otherwise than in the course of providing Services to you under this Agreement and/or any Applicable Agreement.
- 19. You should be aware that we are engaged in certain customer driven and proprietary activities in many markets. These general activities, as well as our hedging activities which are or may be related to certain Investments or Transactions entered into with you, may adversely affect the value of such Investments or Transactions.

Market Forces and Related Risks

- 20. Your payments or receipts under any Investment or Transaction will be dependent on changes in the particular financial market to which the Investment or Transaction is linked, and you will be exposed to price, currency exchange, interest rate or other volatility in that market. Such market movements cannot be predicted accurately. You acknowledge and accept that you may sustain substantial Losses on your Investment or Transaction (such as sustaining a total Loss in excess of the invested amount and any collateral held by us) if the market conditions move against your positions. It is in your interest to fully understand the impact of market movements, in particular the extent of profit or loss that you would be exposed to when there is an upward or downward movement in the relevant rates and the extent of loss if you have to liquidate a position if market conditions move against you. Your positions may be liquidated at a loss and you will be liable for any resulting deficit.
- 21. You should also be aware that there is a general risk of market failure or collapse which may arise from any political or financial development or any unpredictable event that may immediately result in sharp price movements, volatile market conditions and strained market liquidity.
- 22. Under certain market conditions, it may be difficult or impossible to liquidate a position, to assess a fair price or assess risk exposure. This can happen, for example, where the market for an Investment or Transaction is illiquid, where there is simply no market traders for such Investment or Transaction, where there is a failure in electronic or telecommunications systems or where there is the occurrence of an Event of Force Majeure (which includes any form of restriction, moratorium or suspension on trading imposed by an Exchange, market or other authority regulating trading in the Investments or Transactions). Such events will increase the risk of loss to you.
- 23. On certain Exchanges, the performance of a Transaction by a broker (or any third party with whom he is dealing on your behalf) may be "guaranteed" by that Exchange. However, any such guarantee is unlikely in most circumstances to provide full cover and may not protect you completely if the broker or the third party defaults on its obligations to you.
- 24. As the prices of over-the-counter Transactions are individually negotiated and there is no central source for obtaining prices, there are or may be inefficiencies in transaction pricing. We consequently cannot and do not warrant that our prices or the prices that we secure for you are or will at any time be the best price available to you. We may make a profit from an over-the-counter Transaction undertaken by you regardless of the outcome of that over-the-counter Transaction from your point of view. Further, since there is no central market, over-the-counter Transactions may only be unwound with the agreement of the counterparty. Over-the-counter Transactions entail more risks than exchange-traded Transactions in terms of liquidity risk, credit and counterparty risk and pricing transparency.
- 25. Placing contingent orders (such as "stop-loss" or "stop-limit" orders, which are intended to limit Losses to certain amounts) may not necessarily limit your Losses to the intended amounts, as it may be difficult or impossible to execute such orders either in accordance with your Instructions, or at all, under certain market conditions. At times, it is also difficult or impossible to liquidate a position without incurring substantial

Losses. Accordingly, you accept and bear the risk of, and hereby release and discharge us from, all liability arising out of the execution or the non-execution of a "stop-loss" or "stop-limit" order and pursuant to such acceptance authorise us, should any such circumstances occur, to execute any order at such rate and in such manner as we may deem appropriate. Strategies using combinations of positions, such as "spread" and "straddle" positions, may be as risky as taking simple "long" or "short" positions.

Securities Trading

The prices of Securities fluctuate, sometimes dramatically. The prices of Securities may move up or down. In certain circumstances, the Securities may become valueless. There is therefore an inherent risk that Losses rather than profits may be incurred as a result of buying and selling Securities.

Exchange Traded Instruments, Trading Facilities and Electronic Trading

- 27. In respect of Investments or Transactions involving underlying contracts or instruments which are traded on an Exchange, market conditions of the Exchange (such as liquidity) and/or the operation of the rules of such Exchange (such as any discretion on the part of the Exchange to suspend or limit trading of any contract or instrument because of price limits or "circuit breakers") may increase the risk of loss by making it difficult or impossible to effect any transaction (including closing out any Investment or Transaction) or liquidate or offset any position. Under certain circumstances, the specifications of outstanding contracts (including the exercise price of an option) may also be modified by the Exchange or clearing house to reflect changes in the underlying interest.
- 28. Most open-outcry and electronic trading facilities are supported by computer-based component systems for the order-routing, execution, matching, registration or clearing of trades. As with all facilities and systems, they are vulnerable to temporary disruption or failure. Your ability to recover Losses may be subject to limits on liability imposed by one or more parties, including the system provider, the Exchange, clearing house or member firms and such limits may vary. You should obtain details in this respect from the relevant parties.
- 29. If you trade through or on an electronic trading system, you will be exposed to the risks of any defect, deficiency or malfunction in, and/or any breakdown, disruption or failure of, any telecommunications, computer or other electronic equipment or system associated with such electronic system. This could result in a disruption in the trading activities at the Exchange or an unavailability of reference prices for the relevant Investment or Transaction. In such circumstances, the Investment or Transaction may not be executed according to your Instructions or may not be executed at all, which may lead to Losses to you.

Alternative Stock Markets

- 30. Alternative stock markets (such as the Growth Enterprise Market in Hong Kong) are often established as a market designed to accommodate companies with higher investment risk. In particular, companies may list on an alternative stock market with neither a track record of profitability nor any obligation to forecast future profitability. There may be risks arising out of the emerging nature of companies listed on an alternative stock market and the business sectors or countries in which such companies operate. Securities listed on such alternative stock markets may be susceptible to higher market volatility and/or a lack of liquidity, as compared to main board listed Securities. The higher risk profile and characteristics of an alternate stock market mean that it is a market more suited to professional and other sophisticated investors.
- 31. You should also be aware that companies listed on an alternative stock market are generally not required to issue announcements in gazetted newspapers. As such, the principal means of information dissemination on such markets is through publication on an internet website. Accordingly, you should ensure that you have access to up-to-date information on the companies listed on an alternative stock market as published on the relevant internet website.

Investments or Transactions in Other Jurisdictions

32. Investments or Transactions in other jurisdictions may expose you to additional risk. Your investment may entail trading on foreign exchanges or in foreign markets where the issuer or issuer group of companies are located or governed in. Such markets can have different rules and may be subject to regulations that may offer different or diminished investor protection or varied financial/creditor hierarchy frameworks. As an illustration, there may be recovery and resolution regimes in certain markets which may empower resolution authorities to intervene if a financial institution or its parent company becomes non-viable. Such intervention may take the form of a bail-in, through the writing down of certain claims of unsecured creditors of the failing institution and/or conversion of unsecured debt claims to equity, impacting the ranking and priority of investors. Other forms of intervention can also include (and are not limited to) powers of authorities to

exercise powers to stay certain contractual rights that may have earlier been agreed upon by the contracting parties such as and including the right to terminate the Investment contract/ Transaction. Should the Issuer (and/or relevant parties such as any guarantors (as applicable) and/or any of their affiliates) become subject to such bail-in or resolution powers, this could have an adverse impact on the payments under and returns on the Notes and investors may lose some or all of their investment. Investors can face difficulties in instituting any action as a creditor or investor.

Before you trade, you should enquire about any rule relevant to your particular Investment or Transaction and make an independent assessment and investigation. Your local regulatory authority may be unable to compel the enforcement of rules of the regulatory authorities or markets in other jurisdictions where your Investments or Transactions have been effected. You should enquire about the details relating to the types of redress available in both your home jurisdiction and the other relevant jurisdictions before you trade.

Emerging Markets

- 33. Emerging markets typically refer to markets in countries with moderate to low per capita national income. While Investments or Transactions involving emerging markets financial instruments may yield large gains, they can also be highly risky and unpredictable. There may be inadequate regulations and safeguards available to investors. Besides the risks inherent in all investments, those associated with emerging markets include country/region risk where government intervention in markets, perhaps in the form of exchange control laws or restrictions in the repatriation of profits, which may affect the value of any Investment or Transaction or your ability to enjoy its benefits. In addition, events (for instance, natural disasters, fluctuations in commodity prices and/or exchange rates and political upheavals) which may have a minor or limited effect in more mature markets could affect emerging markets profoundly.
- 34. Any Investment or Transaction involving emerging markets financial instruments or referencing an emerging market needs careful and independent assessment of the risks in relation thereto (including sovereign risk, issuer risk, price risk, political risk and liquidity risk).

Indicative Values of Derivative Transactions and Non-Listed Instruments

- 35. In respect of Investments or Transactions in financial derivatives and non-listed financial instruments, in particular in "combined" or "structured" transactions, the absence of a "market" or "common" reference price may make it impossible for us to provide the precise value of that Investment or Transaction. You should therefore be aware that the price indications provided by us are always based on the latest available market prices (if any) of the underlying instrument or have been derived from sources believed to be reliable. Consequently, price indications provided by us may not reflect the actual price at which an Investment or Transaction may be terminated or unwound, if this is possible at all. We do not make any representation as to the accuracy or completeness of price indications for any Investment or Transactions and do not accept liability for any Losses arising from the use thereof.
- 35A. The "net open positions", "market" price or value of Investments or Transactions in financial derivatives and non-listed financial instruments are (where possible and subject to the circumstances stated here) typically marked-to-market from time to time by the Bank and may take reference to the rates or prices prevailing at the relevant time in the relevant market obtained from various sources such as and not limited to: screen based sources, benchmark administrators, other market participants, dealers in the relevant markets, information vendors, or sourced from internally, or by employing valuation methods that the Bank deems to be commonly accepted or applicable. Valuation information from third parties may be part of these third parties' proprietary systems which the Bank has no oversight of. The valuations may therefore not correspond with valuations given by another market participant and the Bank shall have no liability in respect of any error or omission arising from the valuations given or from use of the valuations or reliance placed on them.

The valuation of the Investments or Transactions would also be duly discounted by the Bank in accordance with its prevailing policies and practices in certain circumstances such as and not limited to where these constitute collateral in favour of the Bank or for purposes of setting off your Liabilities. For certain Investments or Transactions, mark-to-market gains may also not be factored into the valuation by the Bank. The Bank may include its cost of funding, hedging or other requirements to the extent it deems required to adjust any quotations and valuations and the client accepts the valuation from the Bank on an "as-is" basis and which may be adjusted at any point in time.

Off-Exchange Investments or Transactions

36. Off- exchange Investments or Transactions may be less regulated or subject to a separate regulatory regime and as such, the risk is correspondingly higher. Additionally, such off-exchange Investments or Transactions may be non-transferrable and therefore, it may be difficult or impossible for you to close out or liquidate an open position. Situations may also arise where no market traders are prepared to deal in such off-exchange Investments or Transactions or no proper information may be available to determine the value or the fair price of such Investments or Transactions or to assess the exposure to risk. Before entering into any such off-exchange Investments or Transactions, you should fully familiarise yourself with all applicable rules and the attendant risks.

Leveraged Transactions / Margin Trading

- 37. Leveraging may be undertaken by way of a loan, utilisation of margin trading facilities or may be embedded within an instrument such as a structured note. The high degree of leverage resulting from a relatively small margin requirement in respect of any Investment or Transaction can work against you as well as for you due to fluctuating market conditions and you acknowledge that the use of leverage can lead to large Losses in excess of the original invested amount, as well as gains.
- 38. When you undertake any Investment or Transaction using any margin trading facilities extended to you, please be aware of the following:
 - (a) You must ensure that you have sufficient margin in respect of any approved limit. The required amount of margin varies, depending on the Investments or Transactions undertaken by you and is determined by us from time to time, in our absolute discretion, and may be changed at any time.
 - (b) We retain full discretion to determine the Assets that are acceptable as margin and the valuation of such Assets.
 - (c) The risk of loss in financing an Investment or Transaction through the use of the margin trading facilities is significant. You may sustain Losses in excess of the margin deposited with us as a relatively small market movement may have a large impact on the value of the margin. In some cases, while the amount of the initial margin deposited with us may be small relative to the value of the Investment or Transaction, a relatively small market movement would have a proportionately larger impact on the margin deposited or to be deposited with us; this may work against you as well as for you.
 - (d) If the market moves against you, you may not only sustain a total loss of your initial margin deposit and any additional margin deposited with us to maintain your position, you may also incur further Liability to us or sustain further or additional Losses.
- 39. Subject to the provisions of any Applicable Agreement, if we, in our absolute discretion, determine that the value of the margin deposited with us is insufficient to support your exposure in all or any of the Investments or Transactions, we may take such action as we deem fit to make up the shortfall, including:
 - (a) calling upon you to deposit additional margin at short notice in order to maintain your position; if you fail to comply with the request for additional margin within the specified period, we may liquidate all or any of your Investment or Transaction at a loss and you will be liable for any resulting deficit in your Account;
 - (b) realising such part or all of the margin or other collateral as we deem necessary to satisfy your Liabilities, without notice to or consent from you; and/or
 - (c) closing out, liquidating, setting off, realising or otherwise dealing with any or all of your outstanding Investments or Transactions (notwithstanding that any such Investment or Transaction has not yet matured and regardless of whether you may suffer Loss as a result thereof) at such time and by such means and in such manner as we, in our absolute discretion, deem appropriate, without notice to or consent from you.
- 40. You must be prepared to pay interest cost for the use of any margin trading facilities. Any such facilities may also be terminated or varied by us in accordance with the terms of the Applicable Agreement.
- 41. You should therefore carefully consider whether carrying out leveraged Investments or Transactions is suitable for you, in light of your own financial position and investment objective, and ensure that you do not commit yourself to any Investment or Transaction which is beyond your means.

Counterparty Risk and Credit Risk

- 42. If we are your counterparty to any Investment or Transaction, you are taking on and are subject to the credit risk of DBS Bank Ltd., Singapore. You understand that we deal with you at arms' length as your counterparty. We are not your fiduciary, nor do we accept any fiduciary obligation owed to you. You acknowledge that any dealing, trading or engagement or transaction with us by you could result in Losses to you and a gain to us. Subject to any Applicable Law, we are not obliged to account to you for the gain that we enjoy.
- 43. We do not and will not give you any advice, whether written or oral, other than the representations which will be expressly set forth in any Applicable Agreement and any confirmation which may be signed or executed by you after negotiations with us as your counterparty.
- 44. We may not always be your contractual counterparty or the issuer of certain Investments or Transactions. Where we are not your contractual counterparty or the issuer, your contractual counterparty or the third party issuer, and not us, will be liable to you under that Investment or Transaction. Accordingly, in considering whether to enter into any Investment or Transaction, you should take into account all risks associated with such counterparty or third party issuer, including the counterparty's or third party issuer's financial standing.
- 45. Often, you will be purchasing an unsecured obligation of such counterparty (as opposed to an obligation of a central clearing corporation as would be the case with exchange traded futures and options) and you should evaluate the comparative credit risk. In addition, there is a risk of counterparty or issuer default which may arise from, *inter alia*, insolvency factors.
- 46. You agree that any Investment or Transaction entered into on your behalf with any counterparty and/or broker is subject to the prevailing terms and conditions as may be stipulated by such counterparty and/or broker and is dependent on the performance, settlement, payment or delivery by such counterparty and/or broker, including their respective agents (notwithstanding that between you and us, we act as principal in such Investment or Transaction). You shall hold us, our Personnel, Affiliates, Nominees and Agents harmless from any liability in connection with the failure of these parties to meet their obligations/ responsibilities and that of any other external parties involved in the said Investment or Transaction. Any insolvency or default of such counterparty and/or broker may result in Losses to you or lead to your positions being liquidated or closed out without prior notice to or consent from you. In certain circumstances, you may not even get back the actual Assets which you have deposited as collateral with us and you may have to accept any available payment in cash.
- 47. Certain transactions also involve the assumption by you of other credit risks and you should ensure that you are able to evaluate all such risks.

Deposited Assets and Cash

- 48. When we hold in custody any Asset, we may/ likely will use the services of an Agent. Assets of clients are held on an omnibus basis and depending on the policies and location (among others) of our Agents, our Agents may decide to impose requirements on the entire omnibus holding custodised by the Bank, if they deem that certain laws or regulatory directions warrant these. It is not expected that this would occur on a frequent basis, and it might be temporary, however, there may be restrictions on you and your dealings in existing (or new) investments/ holdings. Whilst these restrictions are in place, you may not be able to deal in your Assets at all nor would you be able to receive any benefits (such as distributions) or exercise any rights in respect of them. Your Assets would continue to be subject to market fluctuations, changes and other implications that in the worst case scenario, could result in Losses to you.
- 49. You should familiarise yourself with the protections accorded to any Cash and other Assets deposited by you in connection with any Investment or Transaction, particularly in the event of our insolvency, or the insolvency of an issuer, counterparty, custodian or intermediary. The extent to which you may recover your Cash or other Assets will be governed by Applicable Laws and you may in certain cases fail to recover all of such Cash or Assets. In some jurisdictions, property which had been specifically identified as your own will be pro-rated in the same manner as cash for the purposes of distribution in the event of a shortfall.

Transaction Costs and Tax

50. Before entering into any Investment or Transaction, you should request for a clear explanation of all commissions, fees and other charges for which you will be liable. Your net returns from any Investment or Transaction would be affected by any such commissions, fees and other charges, as well as any relevant tax liability (such as income tax). The tax implications of any Investment or Transaction are dependent upon the nature of your business activities and the Investment or Transaction in question and you should therefore consult your tax adviser to understand the relevant tax considerations. These transaction costs must be considered in any risk assessment undertaken by you.

51. You should be aware that the interest payable by you under any Credit Facilities (where applicable), foreign exchange risks and any negative gearing are variables that add to the risks of any Investment or Transaction.

Liquidity and Marketability Risks

- 52. You should be aware that at certain times, or under certain market conditions, it may be difficult or impossible to liquidate a position, to assess value or to determine a fair price of any Investment or Transaction. Certain Securities and money market instruments, in particular, structured notes or products, may not be readily realisable or marketable. There can be no certainty that market traders will be prepared to deal in them.
- 53. Liquidity risks decrease for near term Investments or Transactions and increase for Investments or Transactions which have longer maturity periods or are linked to emerging markets and instruments with lower credit ratings. Reversing any Investment or Transaction at short notice can be difficult or even, impossible, especially for complicated structures. Unexpected and sudden erosion of liquidity can also arise from sharp price movements and volatile market conditions, resulting in illiquid markets.
- The benefits of customisation of any Investment or Transaction to achieve your particular financial and risk management objectives may be offset by significant liquidity risks.

Currency Risks

- 55. Fluctuations in exchange rates can have an impact on your profit or Loss from any Investment or Transaction if that Investment or Transaction is denominated or settled in a different currency from the currency used to carry out the Investment or Transaction, for your ordinary business or to keep your accounts.
- 56. When you trade in a foreign jurisdiction, you should also take into account the applicable tax and exchange controls, including whether profits may be repatriated by you.

Interest Rate Risk

57. Interest rate fluctuations may have an adverse impact on the value of certain Investments, in particular, debt instruments, such as bonds or money market instruments.

Non-Transferability And Non-Marketability

58. Generally, an Investment or Transaction cannot be assigned or transferred without the consent of the counterparty. In this regard, we are not obliged to terminate, unwind or repurchase any Investment or Transaction from you. If your Investments or Transactions are customised and not fungible, engaging in a transaction with another dealer to offset a Transaction you have entered into with us, whether on your behalf or otherwise, will not automatically close out those positions (as would be true in the case of equivalent exchange-traded futures and options) and will not necessarily function as a perfect hedge and may increase the risk to you. It may also be difficult or impossible to liquidate an existing position, assess the value thereof, determine a fair price or assess the exposure to risk.

Risks Associated With Specific Investments

Fixed Income Investments

- 59. Although Investments in fixed income instruments are perceived to be conservative investments and more predictable than equities, they are not without risks. In particular, in purchasing any fixed income instruments, you should be aware of the following:
 - (a) You take on the risk that the issuer of the fixed income instrument or the counterparty may not honour its obligations to pay principal and/or interest, resulting in Losses to you. In this regard, published ratings of any issuer of fixed income instrument should be supplemented by your own credit analysis of the issuer's credit risk as changes in the ratings of any issuer of fixed income instrument may lag behind changes in financial conditions. You should perform periodic independent analysis to determine the credit risk of the issuer of any fixed income instrument and evaluate the merits and risks of such fixed income instrument.
 - (b) You are exposed to liquidity risk as there may be no market for a fixed income instrument and you may not be able to sell the fixed income instrument at the desired time or price. Even when a market exists, there may be a substantial difference between the offer and purchase price for a fixed income instrument.

- (c) You are exposed to the risk of interest rate fluctuations as the value of fixed income instruments will fluctuate with changes in interest rates. The degree of interest rate sensitivity depends on the maturity, coupon and call provisions of the fixed income instrument. Floating rate fixed income instruments lessen your interest rate risk to the extent that the interest rate adjustments are responsive to market rate movements. If the issuer of the fixed income instruments has the right to redeem the fixed income instruments before maturity, this can adversely affect your exposure.
- (d) Where the fixed income instrument is held by the Bank (or its nominee) on your behalf, the Bank will provide custody services in respect of such holdings in accordance with the Custody Terms. Should a credit event occur, the Bank will take reasonable steps to forward to you any notice or other communication received in respect of such fixed instrument. As the Bank may not be privy to the debt restructuring plans and/or other negotiations between the issuer and its creditors (due to the confidential nature of such discussions), the availability and flow of information may be greatly diminished in such circumstances.

Options

- 60. Options are essentially contracts whereby the owner of the options has the right, but not the obligation, to purchase or sell an asset at a fixed price at or by a specific date. Investments or Transactions involving options carry a high degree of risk and are not suitable for many members of the public. You should only enter into such Investments or Transactions after you have read, understood and familiarised yourself with the type of options, style of exercise, the nature and extent of rights and obligations and all associated risks. If in doubt, you should seek independent advice so as to understand and be familiar with the risks involved.
- 61. If you purchase an option, you should be aware of the following:
 - (a) Under certain adverse market conditions when the market moves against an option position, the purchased option can expire worthless and you will suffer a total loss of the original investment which would consist of the option premium and the transaction costs. If you are contemplating purchasing deep-out-of-the-money options, you should be aware that ordinarily, the chance of such options becoming profitable is remote.
 - (b) In order to realise any value from the option, it will be necessary either to offset the option position or to exercise the option.
 - (c) Some option contracts may provide only a limited period of time for the exercise of the option, and some option contracts may provide for the exercise of the option on a specified or stipulated date. For barrier options, the exercise rights will only arise when the market value of the underlying instrument reaches the barrier (in the case of knock-in options) or will expire irrevocably when that barrier is reached (in the case of knock-out options).
 - (d) Exercising an option results either in a cash settlement, or the acquisition or delivery of the underlying instrument.
- 62. Buying options involves less risk than selling or writing options because if the price of the underlying instrument moves against you, you can simply allow the option to lapse and your maximum Loss is limited to the premium, plus any commission or other charges. However, if you buy a call option on an underlying instrument and later exercise the option, you will acquire the underlying instrument and therefore expose yourself to the risks on the underlying instrument.
- 63. The risks associated with selling or writing an option is generally greater than purchasing an option. Although the premium received by the seller is fixed, the seller may sustain a Loss well in excess of the amount of premium received. The seller may also be required to deposit additional margin to maintain the position if the market moves unfavourably. If the purchaser exercises the option, the seller would be required to either settle the option in cash, or acquire or deliver the underlying instrument. If the option is "covered" by the seller holding a corresponding position in the underlying instrument or another option, the risk may be reduced. An option is described as "covered" if the option seller already has a corresponding quantity of the relevant underlying instrument at its disposal. Conversely, if the option is not covered, the possible Loss will be unlimited.
- 64. You should carefully calculate the price and the exchange rate (where applicable) which the underlying instrument would have to reach for the option position to become profitable. This would include amounts by which the underlying instrument or the extent at which the exchange rate would have to rise above or fall below the strike price to cover the sum of the premium and all other costs incurred in entering into and exercising or closing the option position.

- 65. Certain Exchanges in some jurisdictions permit deferred payment of the option premium, limiting the liability of the purchaser of an option to margin payments not exceeding the amount of the premium. Nonetheless, the purchaser of an option is still subject to the risk of losing the premium and transaction costs. When the option is exercised or expires, the purchaser is responsible for any unpaid premium outstanding at that time.
- Apart from plain vanilla put and call options, there are other types of options, including non-deliverable foreign exchange options, acquisitions of two or more options commonly known as a combination, and exotic options. There is no limit to the structures, types and terms of such options. Investments or Transactions involving such options are very complex and high risk and you should seek independent advice before entering into any such Investment or Transaction.

Forwards and Futures

- 67. Forwards and futures entail the obligation to deliver or take delivery on a specified expiration date of a defined quantity of an underlying asset at a price agreed. Futures are standardised contracts traded on an Exchange while forwards are traded over-the-counter. Forwards and futures may involve high degree of risks
- 68. When buying or selling an underlying asset by way of a forward or futures contract, a specified initial margin must often be supplied at the beginning of the Investment or Transaction. We may require additional margin to be provided periodically or at any time during the life of the forward or futures contract if we determine that the margin provided by you has fallen below the amount required by us. This usually corresponds to the mark-to-market loss arising from a decline in value of the Investment or Transaction or the underlying assets.
- 69. For forward sales, the underlying asset must be delivered at the strike price agreed even if its market value has risen since the date the Investment or Transaction was entered into. The seller thus does not benefit from the increase in the market value above the agreed strike price.
- 70. For forward purchases, the buyer must take delivery of the underlying asset at the strike price agreed even if its market value has fallen since the date the Investment or Transaction was entered into. The buyer's potential Loss would be the difference between the agreed strike price and the market value of the underlying assets. The maximum Loss corresponds to the strike price. Notwithstanding, potential Losses can substantially exceed any margin requirements.

Structured Products

- 71. Structured products are formed by combining two or more financial instruments, including one or more derivatives. Structured products may carry a high degree of risk and are not suitable for all investors. These products can be extremely complicated and of a high risk nature and may involve risks associated with financial instruments that may be interconnected.
- 72. The net outcome of structured products will depend on the performance of the Underlying. The effect of a market movement (however slight) or event could lead to substantial Losses and may even involve the loss of the entire amount initially invested. You should therefore ensure that you fully understand the risks associated with each Underlying, as well as the structured product as a whole, and satisfy yourself that you are willing to accept all such risks. You should also be aware that each structured product has its own risk profile and given the unlimited number of possible combinations, it is not possible to detail in this General Risk Disclosure Statement all the risks which may arise in any particular case. In this regard, prior to entering into any Investment or Transaction involving structured products, you should obtain independent advice so as to understand and be familiar with the risks involved.
- 73. You should note that with structured products, the buyer can only assert their rights against the issuer; hence particular attention should be paid to issuer risk. You should also be aware that a total loss of your investment is possible if the issuer or its counterparty should default.
- As structured products are usually executed over-the-counter, it may accordingly be difficult to liquidate an existing position, assess the value of, determine a fair price for or assess your exposure to risks under such Investment or Transaction. This uncertainty should be factored in by you in the overall consideration of the potential impact of your Investment or Transaction.

75. Any forecast on the economy, stock market, bond market and economic trends of the markets provided to you is not necessarily indicative of the future or likely performance of any structured products. Where any past performance of a structured product, or that of its Underlying, is provided to you to illustrate possible returns of such structured product, such past performance is also not necessarily indicative of future performance of such structured product.

Structured Deposits

- 76. Unlike traditional Deposits, Structured Deposits have an investment element. The returns on such Structured Deposits are variable and not guaranteed, and are usually contingent on the performance of one or more relevant Underlying. Structured Deposits are also not insured deposits for the purposes of the Deposit Insurance and Policy Owners' Protection Schemes Act. Structured Deposits cover a wide range of investment products and may include range deposits and such like products. Prior to entering into any Investment or Transaction involving Structured Deposits, you should read and understand all the relevant terms and conditions of, and the risks associated with, such Structured Deposits and ensure that such investments are suitable for you.
- 77. Structured Deposits are generally held for longer tenors than traditional fixed or time Deposits. You should therefore ensure that you have sufficient funds and/or the necessary liquidity in order to hold each Structured Deposit until maturity. Unless we otherwise agree, the principal amount of a Structured Deposit cannot be withdrawn, whether partially or in whole, prior to its maturity. If we allow for Premature Termination, we shall be entitled to deduct from the principal amount, any Premature Termination Costs. Certain Structured Deposits may be subject to early termination before maturity by us in accordance with the Applicable Agreement. In such instances, you may receive less than the principal amount invested in the Structured Deposit or be exposed to the potential loss of the principal sum invested in the Structured Deposit if the Structured Deposit is not held to maturity.

Currency Linked Investment

- 78. A Currency Linked Investment involves a currency option which confers on us, as the deposit-taking institution, the right to repay the principal amount and interest accrued thereon on the CLI Maturity Date in either the Base Currency or the Alternate Currency. All or part of the interest received on the Currency Linked Investment represents the premium on the currency option. If you enter into a Currency Linked Investment, you must be prepared to receive the principal amount and accrued interest in the Alternate Currency, converted at the Pre-Agreed Exchange Rate on the CLI Maturity Date.
- 79. Whether the principal amount and interest accrued thereon is to be repaid in the Base Currency or the Alternate Currency will depend on the level of the prevailing exchange rate between the Base Currency and the Alternate Currency against the Pre-Agreed Exchange Rate at the CLI Fixing Time. If repayment is to be made in the Alternate Currency, the amount payable will be converted from the Base Currency into the Alternate Currency at the Pre-Agreed Exchange Rate. This may result in you receiving less than the principal amount initially invested, when such amount in the Alternate Currency is converted back to the Base Currency.
- 80. Currency Linked Investments are subject to exchange rate fluctuations which may affect the returns on such investment. A wide range of factors, including national and international financial and economic conditions and political and natural events, may affect exchange rates. The effect of normal market forces may at times be discounted by interventions of central banks and other bodies, including foreign exchange controls being imposed. At times, the exchange rates may rise or fall sharply. If the exchange rate moves against your favour, you may incur a loss on your principal sum in comparison with the amount initially invested.
- 81. Unless we otherwise agree, the principal amount cannot be withdrawn, whether partially or in whole, prior to its maturity. If we allow for Premature Termination, we shall be entitled to deduct from the amount payable to you any Premature Termination Costs. Any payment you may receive upon such Premature Termination may be substantially less than the principal amount initially invested.
- 82. A Currency Linked Investment may be subject to early termination by us before maturity upon the occurrence of an Extraordinary Event (including, the imposition of exchange control restrictions or any devaluation, redenomination or demonetisation of the Base Currency or the Alternate Currency). In such instances, you may receive less than the principal amount initially invested.

Swaps

- 83. A swap transaction involves an exchange of future payment streams, and occasionally, also an exchange of principal at the start of the transaction or on maturity. Examples of swap transactions include interest rate swaps, currency swaps and total return swaps.
- An investor who enters into an interest rate swap will be subject to interest rate risk. Interest rates movements will affect the swap's cash flow and mark-to-market value. If the swap involves payments in different currencies, your position will also be affected by fluctuations in exchange rates. As movements in interest and exchange rates may be influenced by a variety of factors, such as inflationary fears and a weakening currency, it is often difficult to anticipate such movements.
- 85. A party to a swap transaction also runs the risk that the counterparty will default or otherwise fail to perform its obligations.
- 86. A swap agreement may also be combined with an option and is known as a "swaption". As a hybrid, a swaption generates two important exposures: (a) the probability of exercise and (b) the credit risk emerging from the swap. The credit risk of any swaption is the cost to one counterparty of replacing the swaption in the event that the other counterparty is unable to perform. You should also be aware that liquidity risk is high for swaption with long-dated option components.

Credit Derivatives

- 87. Credit derivatives are financial instruments that permit one party (the beneficiary) to transfer the credit risk of a reference asset (such as traded sovereign and corporate debt instruments or syndicated bank loans), which it typically owns, to another party (the guarantor) without actually selling the reference asset. Credit derivatives can take the form of swaps, options or hybrid financial instruments.
- 88. Credit derivatives involve a liquidity risk. Often, such instruments cannot be sold before maturity as there is no market for such instruments. Investors will additionally be exposed to the credit risk on each of the reference assets, as well as the issuer of the reference assets. As Investments or Transactions involving credit derivatives carry a high degree of risk, you are advised to seek independent professional advice should you decide to enter into such Investments or Transactions.

Unit Trusts

- 89. Unit trusts are not bank deposits or obligations of or guaranteed by us, or any of our Affiliates. A unit trust is a pool of money managed collectively by a fund manager who invests in a portfolio of assets to achieve certain investment objectives. Unit trusts are naturally subject to investment risks including the possible loss of the principal amount invested. The value of units and the income from any unit trust may fall as well as rise and cannot be guaranteed. Past performance of a unit trust is not necessarily indicative of the future performance of that unit trust.
- 90. Any forecast or opinion provided to you by the fund manager are as at the date of the document and is subject to change from time to time. Such forecasts or opinions should not be regarded as a guarantee of future or likely performance of the unit trust. You should also note that there are necessarily limitations whenever performance is stated or comparison is made to another unit trust for a period of less than three years and that there are limitations and difficulties in using any graph, chart, formula or other device to determine whether or not and when to make an investment in any unit trust.

Exchange Traded Funds

- 91. Exchange traded funds are open-ended investment funds listed and traded intra-day on an Exchange. Exchange traded funds are not principal protected and you may not get back your original investment. You should also be aware that exchange traded funds may not make any dividend distributions, even if the Securities it holds do so.
- 92. An exchange traded fund is exposed to the liquidity and market risks of the Securities it holds and may incur substantial Losses due to the inability to dispose of its holdings of any affected Securities. Further, an exchange traded fund may concentrate its investments in issuers of one or more particular industries or geographical regions. If the particular industry or geographical location performs poorly, this will magnify the negative impact on the value of the exchange traded funds.
- 93. You should also be aware that most exchange traded funds are not actively managed. Accordingly, exchange traded funds may be adversely affected by a decline in the market segments relating to its Underlying.

- 94. A number of factors may also affect an exchange traded fund's ability to achieve a high correlation with its Underlying and there can be no guarantee that an exchange traded fund will achieve a high degree of correlation. A failure to achieve a high degree of correlation may prevent an exchange traded fund from achieving its investment objectives.
- 95. There is a risk that the exchange traded fund manager's strategy, the implementation of which is subject to a number of constraints, may not produce the intended results. This risk is especially pertinent when the exchange traded fund does not replicate its Underlying, but instead holds non-index Securities.
- 96. You should be aware that the value of an exchange traded fund may decline when the counterparty with whom the exchange traded fund purchases financial instruments from and/or enter into agreements with, becomes insolvent or otherwise fail to perform their obligations for any reason.
- 97. You should take note of the following additional risks (which are not exhaustive) if you invest in non-traditional exchange traded funds:
 - (a) Non-traditional exchange traded funds may employ the technique of short selling to achieve an investment exposure consistent with its investment objective. The use of such short selling technique may involve additional transaction costs and other expenses. As a result, the cost of maintaining a short position may exceed the return on the position, which may cause the non-traditional exchange traded fund to lose money. Under certain market condition, short selling can increase the volatility and decrease the liquidity of certain Securities and may lower the return or result in Losses to the non-traditional exchange traded fund.
 - (b) Non-traditional exchange traded funds may seek to provide a return which is either a multiple and/or an inverse of the daily performance of its Underlying. A non-traditional exchange traded fund rebalances its portfolio on a daily basis, increasing exposure in response to that day's gains or reducing exposure in response to that day's losses and there is a risk of a near complete loss of the value of the non-traditional exchange traded fund. Non-traditional exchange traded funds are designed as short term trading vehicles for investors who intend to actively monitor and manage their portfolios. They are not intended and/or suitable for investors who do not intend to actively monitor and manage their portfolio.
 - (c) For Investments or Transactions involving leveraged non-traditional exchange traded funds, you will be exposed to the risk that any adverse daily performance of that exchange traded fund's Underlying will be leveraged.
 - (d) Inverse exchange traded funds, a type of non-traditional exchange traded fund, are negatively correlated to their Underlying and could lose money when the indices rise. This is a result that is opposite from conventional exchange traded fund.

Non-traditional Funds

- 98. Non-traditional funds are investment companies which differ from traditional equity and bond investments on account of their investment style. Such non-traditional funds include hedge funds, alternative investment funds and offshore funds.
- 99. If you invest in non-traditional funds, you should be aware of the following risks (which are not exhaustive):
 - (a) The investment strategies adopted by such non-traditional funds are often high risk and highly complex. Further, due to the use of leverage, a small movement in the market can lead to a major gain, but any Losses will also be magnified sharply. In certain circumstances, the entire amount of your Investment could be lost.
 - (b) The non-traditional fund industry is largely unregulated and the availability, quality and flow of information may be significantly less than that for traditional investment products. Investors may not be kept informed about the fund's strategies or changes to the fund management team.
 - (c) The liquidity and tradability of non-traditional funds can vary a great deal and fixed holding or "lockup" periods lasting many years are not unusual. Liquidations of such funds may also stretch over many years.
 - (d) Certain non-traditional funds may provide for powers to compulsorily redeem all or any portion of an investor's holdings at any time and for any reason upon short notice. The proceeds that an investor may receive upon such redemption may be substantially less than the amount invested in the fund.

- (e) Many non-traditional funds have an offshore domicile and may be subject to less stringent legislation and supervision, which in turn offers poorer investor protection. Problems or delays may arise in the settlement of buy and sell orders for units in such funds. There is also no guarantee that your legal rights under these non-traditional funds will be enforceable.
- 100. Non-traditional funds can take countless different forms and involve a high degree of risk. Before undertaking any Investment or Transaction involving non-traditional funds, you should seek independent advice about the particular risks involved and carefully study any information (including the Fund Offering Documents) on the relevant Investment or Transaction. You should ensure that you fully understand and agree to assume the risks involved and the exposure to potential Loss (which could involve the complete loss of your Investments).

Private Equity Investments

- 101. Private equity investments generally involve the placing of investment capital in private companies and/or Funds. Such capital may be used for a variety of purposes, including financing of high-risk projects which are expected to generate higher returns, making acquisitions and corporate restructuring.
- 102. The contractual conditions governing a private equity investment often require the contribution of liquid funds in a substantial amount and for a considerable period of time. Such contributions are made either by a single payment or by several payments over a certain period of time. Once you have made the commitment to invest, you must be ready to meet calls for capital contribution, known generally as "capital calls" or "commitment calls", which may be made at short notice. The penalty for failure to honour any capital or commitment calls can be extreme, including a complete forfeiture of any capital already invested. Any failure to honour these calls can constitute a Liability owing by you to the Bank as your custodian.

Where you invest alongside other investors via the Bank as your nominee / custodian, any failure by another investor to (i) honour his private equity investment obligations and/or (ii) continue to meet the eligibility requirements of the private equity investment may have a knock on adverse impact on the entire nominee / custodised holdings of the Bank, and in turn would adversely impact your holding. These typically arise from or are a consequence of the terms of the external fund/issuer's offering and constitutional documents.

Restrictions may be imposed on the entire nominee / custodised holdings of the Bank if deemed necessary for compliance with certain laws or regulatory directions. Whilst these restrictions are in place, you may not be able to deal in your Assets at all nor would you be able to receive any benefits (such as distributions) or exercise any rights in respect of them. Your Assets would continue to be subject to market fluctuations, changes and other implications that in the worst case scenario, could result in Losses to you.

- 103. Any capital invested by you may be tied up, either completely or with restricted access, during the term of the private equity investment. As there is no recognised secondary market in private equity investments, such Investments may not be sold and/or transferred freely.
- 104. Private equity investments may be realised in several ways, including a sale of the participations through eventual public listings on Exchanges, mergers with other companies, a sale to another interested party or a recapitalisation. Considerable Losses, or even a total loss of your investment may occur, for example, when such private companies and/or Funds are either wound up or declared insolvent and/or the commercial interest in the business of the private companies or Funds cease to exist.

Investments in Gold Bullion and other Commodities/Commodity Investments

- 105. Investments or Transactions involving Gold Bullion or other commodities carry a high degree of risk and may not be suitable for many members of the public. The extent of loss due to market movements can be substantial or even result in a total loss of the original investment.
- 106. Investments or Transactions involving Gold Bullion or other commodities may be undertaken in many forms, including futures contracts, forward contracts, leveraged trading contracts, contracts made pursuant to trading in differences, spot trading contracts, swaps, options and other derivative transactions involving any combination of one or more of any of the foregoing as well as any other Investment or Transaction which we may carry out with you from time to time.
- 107. The market for Gold Bullion or other commodities is speculative and may be highly volatile. Prices for Gold Bullion and other commodities are affected by a variety of factors, including changes in supply and demand relationship, governmental programmes and policies, national and international political and economic events, wars and acts of terror, changes in interest and exchange rates, trading activities in commodities and

related contracts, weather and agricultural harvest, trade, fiscal, monetary and exchange control policies. You will also be subject to the volatility of the Gold Bullion or other commodities markets and political and other risks in the jurisdictions in which the Gold Bullion or other commodities are traded. The value of Gold Bullion or other commodities may experience downward movements and may under some circumstances even become valueless. Past performance of Gold Bullion or other commodities is not indicative of future results as prices can go up or down. There is therefore an inherent risk that Losses rather than profits may be incurred as a result of buying or selling Gold Bullion or other commodities.

- 108. Changes in the price of Gold Bullion or other commodities can be unpredictable, sudden and large. Such changes may result in the price or value of the commodity or the Investment or Transaction moving adversely against the investor's interests and negatively impacting upon the return on, or settlement of such Investment or Transaction. In extreme circumstances, you may lose all, or a significant proportion of, your initial investment amount.
- 109. Under certain market conditions, it may be difficult or impossible to liquidate a position. Placing contingent orders, such as "stop-loss" or "stop-limit" orders, will not necessarily limit your Losses to the intended amounts, since market conditions may make it impossible to execute such orders.
- 110. The price volatility of a commodity also affects the value of the futures, options and forward contracts related to that commodity and therefore its price at any such time. The volatility of commodity prices is significant and often higher than for an equity portfolio. The commodity markets are in most cases less liquid as compared to the markets for equities, interest or currency-related products.
- 111. Depending on the Investment and/or Transaction in question, you should be aware of the following risks (which are not exhaustive):
 - (a) Commodity Investments are not deposits nor does the value of any Commodity standing to the credit of your Commodity Investment Account (as may be reflected in any Account Statement or Transaction Record) represent a deposit of money.
 - (b) Unless otherwise indicated by us, we act as principal in each Investment or Transaction involving commodities, including Gold Bullion Transactions and any sale or purchase of Commodities through your Commodity Investment Account with us.
 - (c) Certain Investments or Transactions involving Gold Bullion or other commodities may increase liquidity risk and introduce other significant risk factors of a complex character.
 - (d) Over-the-counter bilateral Investments or Transactions involving Gold Bullion or other commodities may be modified or terminated only by mutual consent and subject to agreement on individually negotiated terms. Accordingly, it may not be possible for you to modify, terminate or offset your obligations or your exposure to the risks associated with such Investment or Transaction prior to its agreed termination.
 - (e) There may be adjustments to the terms of your Investments or Transactions involving Gold Bullion or other commodities due to events, such as market disruption, insolvency and changes in any Applicable Law, and such adjustments may reduce the return on your Investments.
- 112. You should take note of the following additional risks if you undertake Commodity Investments through your Commodity Investment Account with us:
 - (a) There may be adjustments to the terms of your Commodity Investment Account opened and/or maintained with us upon the occurrence of certain events, including market disruption, insolvency and changes in any Applicable Law and such adjustments may reduce any amount that may be payable to you.
 - (b) You may not be permitted to take physical delivery of any Commodity purchased or otherwise acquired by you pursuant to any Investment or Transaction and/or standing to the credit of your Commodity Investment Account other than in accordance with the Applicable Agreement and/or the Product Terms and upon such additional terms as we may stipulate from time to time.
- 113. Funds placed with a commodity broker or spot commodity broker for the purpose of participating in foreign markets may not enjoy the same level of protection as funds placed in commodity markets in your home jurisdiction.
- 114. You should not take the risk disclosures set out herein as exhaustive and/or as business, legal, tax, accounting or other advice or as modifying any Applicable Law. These risk disclosures cannot disclose all the risks and other significant aspects of the commodity market or of the commodity futures market. You should therefore carefully study commodity futures trading, trading in commodity contracts and spot commodity trading before you trade.

Digital Tokens / Other Products referencing Digital Tokens

- Holders of Digital Tokens will be subject to the general risks associated with ownership of digital assets. The prices of Digital Tokens are subject to supply and demand and can fluctuate greatly within a short period of time in response to other factors, including security concerns, technological developments and negative publicity. Psychological market risks may have a particular effect on digital assets and their prices may be adversely affected by global or local economic, political, environmental or other factors.
- 116. Returns on new investment asset classes such as Digital Tokens are particularly exposed to legal and regulatory risk. Changes or uncertainty in the legal or regulatory framework relating to blockchain technology and/or digital assets may adversely impact the returns on your investment or even render a previously accepted investment illegal.
- 117. The tax treatment and accounting of Digital Tokens is uncertain and may vary amongst jurisdictions. Investors should seek their own tax advice in connection with purchasing, holding and transferring Digital Tokens, which may result in adverse tax consequences. Changes to tax laws and regulations may also have a large impact on the profitability of your investment in Digital Tokens.
- 118. Cryptographic tokens such as Digital Tokens are a new and relatively untested technology. In addition to the risks noted in this General Risk Disclosure Statement, there may be other risks associated with Digital Tokens that the Bank cannot anticipate. Such risks may have a material adverse effect on the value of the Digital Tokens.

Investments in Digital Tokens can take place via various forms, such as derivatives including options that would reference Digital Tokens. The value of these products is based on or references Digital Tokens (or other digital assets) as the Underlying. These can be traded on an exchange or over the counter. The risks associated with trading, dealing or investing in derivatives are compounded as these risks encompass risks relating to the derivative or structured investment product, and also specifically each Underlying. These risks are set out in greater detail in the other sections on risk disclosures of specific investments as well as on digital tokens, including and not limited to risks such as highly volatile price swings, uncertainty in the legal or regulatory framework and susceptibility to cybersecurity breaches.

Transactions involving Digital Tokens carry a high degree of risk and may not be suitable for all investors. The extent of loss can be substantial or even result in a total loss of the original investment. You should familiarize yourself with the features, risk factors, fees and expenses before investing, and consider whether such investments are appropriate for you with regard to your experience, objectives, financial resources and other relevant circumstances.

119. When purchasing Digital Tokens in initial offerings, investors should note that such offerings could carry a higher risk of being misused for illegal activities due to the pseudo-anonymous nature of the transactions. Investors are likely to be adversely affected if authorities investigate any alleged illicit activities related to the issuer, its business activities, or the trading of the Digital Tokens.

Cryptocurrencies and other Digital Payment Tokens

- 120. The value of cryptocurrencies and other Digital Payment Tokens can be unpredictable and may fluctuate significantly over a short period of time as a result of various factors, including market dynamics, regulatory changes, technical advancements, and economic and political factors.
- Any purchase or sale of Digital Payment Tokens made by the Bank on your behalf will be effected through the Digital Exchange, which is only accessible to members. Due to the different demand and supply and inventory of members, prices on the Digital Exchange may, at any given point in time, be significantly higher or lower than the prices of the same Digital Payment Token in the general market or quoted on other exchanges or trading platforms. You must therefore be prepared to bear the economic risk of an investment in the Security Tokens for an extended period of time, as it may be difficult or impossible for the investor to sell/ dispose of the Security Tokens, including the risk that your entire investment in the Security Tokens may be lost.
- 122. Digital Payment Tokens are intended to represent a new capability on emerging technology that is not fully proven in use. As the technology matures, new capabilities may dramatically alter the usefulness of Digital Payment Tokens or the ability to use or sell them. The functionality of Digital Payment Tokens is complex, will require enhancements and product support over time, and full functionality may take longer than expected. The full functionality of the Digital Payment Tokens is not yet complete and no assurance can be provided of such completion.
- 123. The Digital Payment Tokens may be targeted by hackers or malicious groups or organisations who may attempt to interfere with the Digital Payment Tokens and/or steal the Digital Payment Tokens in various ways, including malware attacks, distributed denial of service, consensus-based attacks, Sybil attacks, phishing, smurfing and hacking.

124. The public blockchains relating to Digital Payment Tokens are susceptible to mining attacks, including double-spend attacks, majority mining power attacks, "selfish-mining" attacks, and race condition attacks. Any successful attacks present a risk to the Digital Payment Tokens, including proper execution and sequencing of transactions involving Digital Payment Tokens, or may result in theft or loss of the Digital Payment Tokens.

Security Tokens

- 125. Security Tokens are complex financial instruments. Prospective investors should ensure that they have the expertise to understand how the product is structured (which may differ from case to case) and the applicable terms and conditions. In most cases, an investor will have no legal or beneficial interest in the asset represented by the Security Token; rather, the investor will have a contractual relationship with the issuer only and its rights and remedies with respect to the investment will be limited to contractual remedies against the issuer in accordance with the terms of the relevant Security Token.
- 126. Issuers of Security Tokens may be early stage companies without a track record of profitability. Investing in such companies involves a high level of risk and you may suffer a total loss of your investment.
- 127. Security Tokens typically confer no governance rights of any kind with respect to the issuer. Accordingly, key decisions, including decisions to liquidate the issuer and/or to create and sell more Security Tokens, will be made by the issuer at its sole discretion. These decisions could adversely affect the value of the Security Tokens.
- 128. Security Tokens are novel and the application of securities laws of various jurisdictions to investors of such assets is unclear in many respects. Changes or uncertainty in the legal or regulatory framework applicable to such assets may adversely affect the value of the Security Tokens.
- 129. The issuer of the Security Tokens may not be subject to disclosure and prospectus requirements under Part XIII of the Securities and Futures Act and the Digital Exchange's admission of the issuers is not subject to MAS' oversight. Where the documents or materials issued in connection with the offer or sale of the Security Tokens are not a prospectus as defined in the Securities and Futures Act and are not registered as a prospectus with the MAS, statutory liability under the Securities and Futures Act in relation to the content of prospectuses would not apply, and you should consider carefully whether the investment is suitable for you.
- 130. Security Tokens are typically only available to institutional and/or accredited investors. In addition, save in certain circumstances detailed in the product documentation, sale of or transfers of Security Tokens traded on the Digital Exchange may only take place between members of the Digital Exchange. These restrictions may further limit liquidity in the Security Tokens and/or have a material adverse effect on the development of any trading market in the Security Tokens. You must therefore be prepared to bear the economic risk of an investment in the Security Tokens for an extended period of time, as it may be difficult or impossible for the investor to sell/ dispose of the Security Tokens, including the risk that your entire investment in the Security Tokens may be lost.
- 131. As the Security Tokens traded on the Digital Exchange will be held by the Bank (an intermediary member of the Digital Exchange) on your behalf, the failure, administration or insolvency of the Bank may adversely affect your access to the Digital Exchange.
- 132. You may not have direct recourse against the issuer in respect of your holding of the Security Tokens and may have to enforce your rights through the Bank as the intermediary member. There may be constraints, e.g. due to conflicts of interest, on the Bank's ability to take action against the issuer on your behalf. In such a scenario, you may not be able to realise the value of the Security Tokens in the event of an insolvency of the issuer and/or the relevant intermediary member.
- 133. Members of the DBS Group may act in various capacities in the issuance of Security Tokens traded on the Digital Exchange, for example, as issuer, custodian and/or manager of the offering. The DBS Group also manages and/or operates the Digital Exchange, the Blockchain and the Trading System (and in doing so, would be involved in the minting and burning of the Security Tokens, as well as other decisions relating to the Digital Exchange including and not limited to the delisting of Security Tokens), and members of the DBS Group may also act as intermediaries involved in the distribution of the Security Tokens during the initial offering to end investors and hold these Security Tokens on behalf of such investors. As such, conflicts of interest cannot be completely avoided.

Risks Associated With the Digital Exchange, the Trading System and the Blockchain

- 134. The Digital Exchange's and DBS Group's network or services could be disrupted by various adverse events which are beyond the control of the Digital Exchange and DBS Group. There can be no assurance that any of the intended security measures will be effective. System failures may also result in the inability to trade in the Digital Tokens, as well as prevent the making of markets for the Digital Tokens, which may potentially impact on the liquidity for the Digital Tokens.
- The Digital Exchange has a short and limited operating history which subjects it to a number of uncertainties, in particular the ability to plan for and anticipate the prospects of the Trading System. The use of the Trading System could be affected by a number of factors, including market reception of the Trading System and slowing demand or increasing competition, resulting in reduced number of users of the Trading System which may in turn impact the liquidity of the Digital Tokens.
- Distributed ledger technology generally, and the private Ethereum blockchain software upon which the Blockchain is designed, is still in a relatively early development stage and is unproven. Any malfunction, flaws, breakdown or abandonment of the Ethereum blockchain may prevent access to your Digital Tokens. Furthermore, developments in cryptographic technologies and techniques or changes in consensus protocol or algorithms could present risks to the Digital Tokens, including by rendering ineffective the cryptographic consensus mechanism that the Blockchain uses. There are a variety of possible cryptographic consensus mechanisms, such as "proof of work" and "proof of stake" which may be used either now or in the future, and risks which may arise if there is any improper implementation of such consensus mechanisms, and unanticipated adverse effects may arise from these such consensus mechanisms. Any malfunction, breakdown or flaws in distributed ledger technology used by the Digital Exchange could have a material impact on its ability to execute or settle trades of the Digital Tokens, to maintain accurate records of the ownership of the Digital Tokens and could in turn have a material adverse effect on you.
- 137. While the use of a private blockchain and other logical and governance procedures may limit the impact of distributed ledger technology reliant upon cryptographic consensus mechanisms, such distributed ledger technology reliant upon cryptographic consensus mechanisms may be susceptible to mining attacks, including double-spend attacks, majority mining power attacks, "selfish-mining" attacks, and race condition attacks. Any successful attacks on the Blockchain may present a risk to the Digital Tokens, including proper execution and sequencing of transactions involving the Digital Tokens.
- Hackers, individuals, other malicious groups or organisations may attempt to interfere with the Blockchain, or the information technology systems they are run on or reliant upon, in a variety of ways, including malware attacks, denial of service attacks, ransomware attacks, phishing attacks, rootkits, viruses, worms, consensus-based attacks, smurfing and spoofing. As the Trading System and Blockchain are both based on third party software, there is a risk that a party may intentionally or unintentionally introduce weaknesses into the core infrastructure of the Blockchain.
- 139. The Trading System and Blockchain are reliant on effective and reliable cryptographic solutions. However, cryptography is evolving and cannot guarantee absolute security at all times. Advances in code cracking, or technical advances such as the development of quantum computers, could present risks to cryptographic solutions, and could result in the theft or loss of the Digital Tokens. The Trading System and Blockchain may be prone to attacks on their infrastructure intended to steal information about their technology, financial data or user information. Any significant breach of intended security measures or other disruptions resulting in a compromise of the usability, stability and security of the Trading System or Blockchain may adversely affect the trading of Digital Tokens on the Digital Exchange.
- 140. Settlement of transactions matched on the Digital Exchange takes place outside of the Trading System built on the Blockchain. The Trading System does not settle trades through a regulated central depository. Accordingly, you will not benefit from related legislative protections including provisions dealing with settlement finality, which ensures that transactions settled by a regulated central depository are generally not subject to reversal.
- 141. Prices of Digital Tokens change continuously 24 hours throughout the day. The investor should be aware that he/she may not be able to execute a trade on the Digital Exchange, even if the market price of Cryptocurrencies continues to change.



PART D – SPECIFIC TERMS AND CONDITIONS GOVERNING CUSTODIAL AND NOMINEE SERVICES

- 1. These Custody Terms shall apply if, at your request, we agree to hold in custody any Asset on your behalf, whether in Singapore or elsewhere. For the avoidance of doubt, these Custody Terms shall not apply to any Gold Bullion held by us on your behalf pursuant to the Gold Bullion Terms.
- 2. In the event of any conflict or inconsistency between:
 - (a) these Custody Terms and the Account Terms, these Custody Terms shall prevail; and/or
 - (b) these Custody Terms and any Applicable Agreement in respect of any Asset kept in custody with us, that Applicable Agreement shall prevail.
- 3. We may (but shall not be obliged to) open and operate a custodian account (each a "Custodian Account") for the purpose of holding any Asset on your behalf. We shall act as custodian and/or nominee of such Asset in accordance with these Custody Terms and any Applicable Law. Notwithstanding the foregoing, we may, in our absolute discretion, refuse to accept any Asset for custody.
- 4. Notwithstanding anything to the contrary in these Custody Terms, we shall not be obliged to provide any of the Services set out herein in relation to any Asset which is not acquired through us or supported by our product platform. If we agree to assist you in respect of any such Assets, such assistance shall be provided without responsibility on our part and you shall indemnify us against all Losses which we may suffer or incur as a result thereof.
- 5. You represent and warrant that your Assets or any applicable title or other documents submitted to us for custody are authentic, valid and correct in every respect. In addition to and without prejudice to any other provision in this Agreement, you hereby agree to fully indemnify us against any Losses that we may suffer due to or arising out of the foregoing representations and warranties being untrue or incorrect in any respect.
- 6. You hereby authorise us to take such steps as we may consider expedient to enable us to hold your Assets in accordance with these Custody Terms and without limitation to the generality of the foregoing, we shall be authorised to:
 - (a) acquire, hold in custody, dispose of or otherwise deal with such Assets on your Instructions or those of your Authorised Agent provided that we may, in our absolute discretion, require such Instructions to be in writing and in a form satisfactory to us before acting on the same;
 - (b) comply with all Applicable Laws, including any Applicable Law which impose or purport to impose on a holder of any Asset kept in custody with us a duty to make or refrain from making any payment of monies or other distribution or from taking any action in respect of such Asset;
 - (c) register such Assets in our name and/or in the name of our Nominee and keep such Assets in such locations as we shall think fit;
 - (d) use the services of any Agent for the purpose of holding in custody any Asset on such terms as we deem appropriate. For example, as part of the terms agreed between the Bank and such Agent, our Agents may take a risk based approach and apply certain steps or impose certain requirements on us in order to not be in breach of any laws or directions that might be deemed by them (or us) to have relevance to a particular Transaction or broadly applicable to their services. In turn, we would need to impose the same or stricter conditions on the services that we may offer to you. These requirements may be at short notice or may change due to the changing environment, markets or regulatory landscape in jurisdictions that may have some relevance (directly or indirectly) to your Transaction or the services.
 - Where appropriate, we may authorise such Agent to delegate the performance of some or all of its duties;
 - (e) request payment of, collect and receive interest, dividends, payments or other distributions in respect of any Asset;

- (f) collect monies payable or any other property in respect of any Asset, at your risk, in such currency as we may, in our absolute discretion, determine, and pay such monies or deliver other property to you in such manner as we may, in our absolute discretion, determine;
- (g) surrender such Assets against receipt of monies payable at maturity or, if called prior to maturity, upon redemption provided always that where any Asset is to be redeemed prior to maturity, we shall not be obliged to present such Asset for redemption unless you request in writing that we do so after the redemption call is made;
- (h) choose, on your behalf and in our absolute discretion, to receive a distribution in cash or in kind, unless you have expressly instructed us in writing of your preferred type of distribution, provided always that where a distribution is to be received in kind, you agree that we shall be entitled to disregard any fractional entitlement and you hereby waive any right and claim to such fractional entitlement.
- (i) where applicable, exchange any Asset in interim or temporary form for other Assets in definitive form and (where applicable), deliver physical scrip form of such Assets to any Depository set up for the purpose of and/or operating scripless trading and to complete and deliver any document that may be required in connection therewith;
- (j) deliver any document of title and any other instruments relating to such Assets to you at your own risk;
- (k) execute, as custodian, any necessary declaration or certificate of ownership under any Applicable Law now or in future; and
- (l) take any action as we shall think fit in the performance of our obligations under these Custody Terms.
- 7. You understand and agree that we are under no obligation to:
 - (a) supervise or monitor any of your Investments or Transactions involving any Asset kept in custody with us:
 - (b) advise or make any recommendation for any Investment or Transaction in relation to the sale, purchase or disposal of any Asset kept in custody with us unless we agree otherwise;
 - (c) assess the suitability and appropriateness of any Investment or Transaction involving any Asset kept in custody with us;
 - (d) assist you in claiming any tax benefit(s) or claims for a reduced withholding rate (whether under a tax treaty or any other Applicable Law) in connection with any Asset kept in custody with us, including any exemption from, or a reduced rate of, withholding for income derived from such Asset;
 - (e) assist you to seek reimbursement from any third parties of any amounts due to any withholding that had been applied at a higher rate than that which you are entitled to or which you view to be the applicable rate;
 - (f) insofar as regards Assets in physical scrip form that are kept in custody by us, provide any banking, processing or custodial service in respect of such Assets, save as expressly set out herein or as otherwise agreed to in writing; and
 - (g) attend or authorise you, as proxy, to attend any meeting or exercise any voting and other right attaching to or derived from such Assets or discharge any obligation conferred or imposed by reason by such holding (including rights or obligations in connection with any allotment, subscription, conversion, consolidation or reorganisation or any merger, receivership, bankruptcy, winding-up or other insolvency proceedings or any compromise or arrangement) or investigate, participate or take any affirmative action in connection therewith, provided always that we may, in our absolute discretion and subject to such terms and conditions as we may stipulate, accept and act in accordance with your Instructions in relation to any of the above-mentioned matters.

- 7A. Notwithstanding anything stated herein and without prejudice thereto, in the event that we agree to a request to apply a reduced rate of withholding tax in connection with your assets kept in custody with us, you agree that:
 - (a) we may decline at any time in our sole discretion to proceed further with your request;
 - (b) you will provide to us all information (in any form) as we may require from time to time for the purpose of processing your request. You further represent and warrant that (i) the information provided by you is complete and accurate; (ii) you shall notify us as soon as practicable when there is a change in your circumstances; (iii) you shall update your tax status periodically in such frequency, required form and within the time period as specified by us from time to time; and (iv) you meet all requirements of the relevant tax treaty(ies) or any other Applicable Law pursuant to which you are seeking the application of a reduced rate of withholding tax;
 - (c) you are aware that we shall be entitled to withhold the relevant amount of taxes in accordance with our existing processes and/or as we determine in our discretion to be necessary, and hence any relevant withholding rate(s) applied may be revised from time to time in our discretion;
 - (d) at any point in time, should it be determined by us that a revised/higher withholding rate should have been applied, we are entitled to automatically deduct from your account the relevant difference, such that the amounts due to us or which should have been received by us is equal to the full amount had no withholding or deduction been required; and
 - (e) in the event that a reduced withholding rate is applied, we are not responsible for crediting to you the difference from, or any backdated amounts that had been previously withheld at, the higher withholding rate.
- 8. You agree that any Asset held or registered in our name and/or the name of our Nominee may, in our absolute discretion, be re-deposited with any Agent as may be selected by us (whether in Singapore or elsewhere) for your account and at your sole risk. Provided that we have acted in good faith and used reasonable care in the selection and continued appointment of such Agent, you agree that we shall not be responsible for any act, omission or default or for the insolvency of such Agent.
- 9. Without prejudice to any provision in this Agreement, your Assets deposited with us or held by any third party in our name may be held on a tangible basis or commingled with assets belonging to other parties. You understand and agree that identification by distinctive numbers of any Asset owned by you may not be possible and that your interest in such Assets may not be identifiable by separate certificates or other physical documents or equivalent electronic records. In such case, we will maintain records of your interest in such Assets which have been commingled in the manner as provided herein. Neither us nor any Agent holding any Asset on your behalf shall be bound to return to you the identical Assets deposited in custody so long as such Assets returned to you are of the same class, denomination, quantity and nominal amount and rank pari passu with those originally deposited with or transferred to us (subject always to any capital reorganisation in the meantime).
- 10. You agree to pay such custody fees as may be specified in our published fees and benefits schedule in force at the relevant time in respect of the Services made available to you under these Custody Terms. Without prejudice to and in addition to the foregoing, we may charge a separate custodian or safe-keeping fee for our handling of any Asset which is in physical scrip form.
- 11. Unless otherwise instructed by you in writing, we will take reasonable steps to forward to you, as soon as practicable after receipt, any notice or other communication (including those relating to redemption, rights issues, bonus issues and matters relating to corporate changes) received in respect of Assets kept in custody by us pursuant to these Custody Terms. If we receive any notice or communication that does not call for any corporate action, we shall have no obligation to forward such notice or communication to you. Save for Losses arising from gross negligence, fraud or wilful default on our part, we and/or our Nominees shall not be liable for any Losses arising out of or in consequence of any failure to forward such notice or communication correctly or promptly or in sufficient time for Instructions to be given to any matter referred to in such notice or communication.

- 12. Without prejudice to Clause 7(f) of these Custody Terms, after becoming aware of any corporate event (including any actual or proposed takeover, offer, sale, merger, compromise, arrangement, bankruptcy, insolvency or administrative proceedings affecting or in relation to any Asset or the issuer of any Asset or in relation to any rights for conversion, transfer or exchange of Asset), we may (but shall not be obliged to), subject to such terms and conditions as we may stipulate, accept and act in accordance with your Instructions in relation to any of the above-mentioned matters, provided always that in the absence of any Instruction from you in due time, we may take or not take any action as we may, in our absolute discretion, consider reasonably appropriate under the circumstances. The foregoing is subject to our rights under this Agreement and/or any Applicable Agreement in respect of such Asset.
- 13. Without prejudice to any provision in these Custody Terms, you acknowledge and agree that we may, on occasions, hold the same class of assets on behalf of one or more customers and in exercising any voting and other rights attached to such assets, we will exercise such rights in accordance with the Instructions given to us by the majority of such customers, notwithstanding that you may have given contrary Instructions in respect of your Assets. Further, you acknowledge that we do not guarantee nor are we under any obligation to ensure that any corporate action that may be taken by the issuer of any Asset would reflect and/or be in accordance with your Instructions to us.
- 14. In the absence of any prior contrary Instruction from you, we may settle or compromise, for and on your behalf, any Liability arising from or in respect of our holding of your Assets. For this purpose, we are authorised to debit such amount payable from any of your Bank Accounts and where necessary, effect any currency conversion at the prevailing rate of exchange, to settle or compromise such Liabilities and we shall not be liable for any Losses arising from the settlement or compromise of such Liabilities or the failure to do so. The foregoing shall be without prejudice to any right of set-off and/or lien that we may have over such Assets.
- 15. We will furnish you with periodic statements in respect of your Assets and notify you of any Transaction carried out in respect of such Assets as soon as practicable after each Transaction.
- 16. Certain charges or expenses incurred by the Bank on your behalf and charged to your account may be rounded up. Depending on the product, such rounding may apply to the amount or percentage as the case may be. Charges and expenses refer to any expenses that may be incurred by DBS on your behalf and may not be foreseeable at the time of the transaction, including and not limited to gas/miners' fees (for crypto transactions), locally mandated charges, commissions, taxes, special handling transfer fees, and others. These may be rounded up and hence amounts may marginally exceed that which is actually incurred, is not refundable and would be dealt with by DBS.
- 17. The provision of Services set out in these Custody Terms does not constitute us as a trustee (other than a bare trustee) and we shall have no trust, fiduciary or other obligations in respect of any Asset kept in custody with us pursuant to these Custody Terms, save for those expressly provided for in these Custody Terms or as otherwise agreed in writing.
- 18. We will use the same care with respect to your Assets kept in custody with us as we use in respect of our own similar property, provided always that we are not required to maintain any insurance in respect of any Asset kept in custody with us pursuant to these Custody Terms.
- 19. Unless otherwise assigned or charged to us by way of security and subject always to Applicable Laws and/or any Applicable Agreement (including any subscription agreement, information memorandum or other offering document relating to the Assets), your Assets kept in custody with us pursuant to these Custody Terms are subject at all times to your control.
- 20. The Services to which these Custody Terms applies may be terminated by you or us in accordance with this Agreement. Upon termination, we will, subject to the release and discharge of any security created over any of such Assets in favour of us, deliver all your Assets to you or to such other party as you may designate, provided always that we shall not be required to make any such delivery unless and until we have received full payment of all Liabilities. You acknowledge that until payment in full of all Liabilities is made, we shall have a right of lien in respect of your Assets.
- 21. You shall hold us, our Personnel, Affiliates, Nominees and Agents harmless from any liability in connection with the matters set out in this section and fully indemnify the aforesaid parties against all Losses which they may suffer or incur in connection therewith.

PART E – SPECIFIC TERMS AND CONDITIONS GOVERNING PRODUCTS AND SERVICES

- 1. These Product Terms set out the terms and conditions applicable to Deposits and certain investment products which we may make available to you from time to time, and the Services which we may offer to you in relation to such Deposits or investment products.
- 2. In the event of any conflict or inconsistency between:
 - (a) these Product Terms and the Account Terms, these Product Terms shall prevail; and/or
 - (b) these Product Terms and any Applicable Agreement in respect of any Deposit or investment product, that Applicable Agreement shall prevail.
- 3. General
- 3.1 The principal amount in respect of any Deposit or investment product, as the case may be, must be received by us on or before the Deposit Date or, as the case may be, such date as we may stipulate in respect of any investment product, provided always that we may, in our absolute discretion, decline to accept any such funds received (or any part thereof) as the principal amount for such Deposit or investment product. Any monies received but not so accepted by us shall be returned to you in such manner as we consider appropriate, including by way of credit to any Bank Account.
- 3.2 Subject to the provisions of any Applicable Agreement, you acknowledge that all Deposits and investment products are meant to be held or maintained to maturity and you shall not be entitled to undertake any Premature Termination without our consent, which consent may be granted in our absolute discretion and on such terms and conditions as we may determine.
- 3.3 If we permit any Premature Termination of any Deposit or investment product, you agree that we may deduct any Premature Termination Costs from any amount payable to you in respect of such Deposit or investment product. In the event that the Premature Termination Costs exceeds such amount payable to you, the deficit shall form a debt due and payable by you to us in respect of which we shall be entitled to exercise our rights under this Agreement. Without prejudice to the foregoing, any payment to you in connection with any Premature Termination shall be determined solely by us and you agree that any such payment may be dependent upon our ability to successfully discharge any related hedge or arrangement.
- 3.4 **Deposit Insurance Scheme** Singapore dollar deposits of non-bank depositors and monies and deposits denominated in Singapore dollars under the Supplementary Retirement Scheme are insured by the Singapore Deposit Insurance Corporation, for up to S\$100,000 in aggregate per depositor per Scheme member by law. Monies and deposits denominated in Singapore dollars under the CPF Investment Scheme and CPF Retirement Sum Scheme are aggregated and separately insured up to S\$100,000 for each depositor per Scheme member. Foreign currency deposits, dual currency investments, structured deposits and other investment products are not insured.
- For the avoidance of doubt, the following terms and expressions used in Clause 3.4 shall have the meanings set out below:
 - **"Supplementary Retirement Scheme"** has the same meaning given to it in the Deposit Insurance and Policy Owners' Protection Schemes Act.
 - **"CPF Investment Scheme"** refers to "CPFIS" as defined in the Deposit Insurance and Policy Owners' Protection Schemes Act.
 - "CPF Retirement Sum Scheme" refers to "CPFRS" as defined in the Deposit Insurance and Policy Owners' Protection Schemes Act
- 4. Deposits
- 4.1 We may, at any time, prescribe such terms and conditions as we may deem appropriate in respect of any Deposit and/or revise any of such terms and conditions, including, where applicable, the currency of the Deposit, the minimum placement period, the minimum placement amount, subsequent placement and the interest rate (including negative interest, where applicable) or yield in respect of such Deposit.

- 4.2 We may, following placement of a Deposit, send you a deposit advice or statement. Such deposit advice or statement are only evidence of a deposit and not a document of title and may not be pledged as security.
- 4.3 Interest (including negative interest, where applicable) in respect of any Deposit shall accrue on the principal amount, at such rate and on such basis as may be agreed between you and us, from the Deposit Date up to but excluding the Deposit Maturity Date. Where interest is payable to you, such interest shall be paid on the Deposit Maturity Date. Where negative interest is chargeable on the Deposit, such interest shall be debited from your Account on the Deposit Maturity Date.
- 4.4 Unless otherwise agreed between you and us, the principal amount of any Deposit together with the interest accrued thereon (including negative interest, where applicable), will be paid to you (or debited in relation to negative interest, where applicable) on the Deposit Maturity Date. In the absence of any Instruction from you to the contrary, we are authorised (but not obliged) to renew such Deposit on such terms and conditions and for such tenor as we shall deem appropriate.
- 4.5 Without prejudice to any provision in this Agreement and/or any Applicable Agreement, upon the occurrence of an Extraordinary Event, we shall be entitled to take such action in respect of any Deposit as we deem fit including:
 - (a) terminating any Deposit, upon which we shall pay to you an amount to be determined by us, in our absolute discretion, as being the fair market value of such Deposit as at the date of termination less any Losses that may be suffered and/or incurred by us as a result of such termination; or
 - (b) where applicable:
 - (i) reducing, suspending or stopping the payment of interest on the Deposit;
 - (ii) varying the interest rate applicable to the Deposit (including to impose zero or negative interest rates on the Deposit), the currency of the Deposit and the Deposit Period and you shall pay us the relevant fees and charges and any amount owing to us (including negative interest, service fees, if any);
 - (iii) impose and debit such fees or charges from your Account.

5. Currency Linked Investments

- 5.1 The terms of each Currency Linked Investment shall be agreed prior to the CLI Value Date, either orally or in writing, and confirmed in such form as we shall, in our absolute discretion deem appropriate.
- 5.2 We may, at any time, prescribe such terms and conditions as we may deem appropriate in respect of any Currency Linked Investment and/or revise any of such terms and conditions, including the Base Currency, the Alternate Currency, the tenor, the minimum or maximum amounts which may be placed as the principal amount and the applicable interest rate in respect of such Currency Linked Investment.
- 5.3 Interest in respect of a Currency Linked Investment shall accrue on the principal amount, at such rate and on such basis as may be agreed between you and us, from the CLI Value Date up to but excluding the CLI Maturity Date.
- 5.4 The principal amount in respect of any Currency Linked Investment, together with interest accrued thereon, will be paid to you on the CLI Maturity Date in either the Base Currency or the Alternate Currency, depending on the level of the prevailing exchange rate between the Base Currency and the Alternate Currency against the Pre-Agreed Exchange Rate at the CLI Fixing Time, as determined by us in good faith. If such payment shall be made in the Alternate Currency, the principal amount and interest accrued thereon shall be converted into the Alternate Currency at the Pre-Agreed Exchange Rate.
- 5.5 Without prejudice to any provision in this Agreement and/or any Applicable Agreement, upon the occurrence of an Extraordinary Event, we shall be entitled to terminate any Currency Linked Investment. Upon such termination, we shall pay to you an amount as determined by us, in our absolute discretion, to be the fair market value of the Currency Linked Investment as at the date of termination less any Losses that may be suffered and/or incurred by us (including the costs of unwinding any related underlying hedge, realising

any option, future contract or any other financial instrument). In this regard, you acknowledge that the fair market value of the Currency Linked Investment may be substantially less than the principal amount placed with us and we shall not be liable to you for any Losses that may be suffered and/or incurred by you in connection with or by reason of such termination.

All rates, fixings and values required for the purpose of calculating any amount payable to you, and all other matters to be ascertained or established in respect of any Currency Linked Investment, shall be conclusively determined by us in good faith and in accordance with the prevailing market practices. In the absence of any manifest error, any such determination shall be conclusive and binding on you. Save as stated herein, in carrying out such calculations, we do not assume any obligation or duty as your agent or act in any way as your trustee.

6. Structured Deposits

- The terms and conditions of each Structured Deposit shall be agreed prior to its commencement, either orally or in writing, and confirmed in an Applicable Agreement. You shall if required by us, execute and return such Applicable Agreement to us forthwith, and in any event, within such time period as may be specified by
- 6.2 Interest (if any) in respect of a Structured Deposit shall accrue and be paid in accordance with the terms and conditions set out in the Applicable Agreement. You acknowledge that payment of interest may be contingent on the performance of an Underlying as specified in the Applicable Agreement.

7. Commodity Investments/Commodity Investment Account

- 7.1 You authorise us to open, maintain and continue an Account in your name for the purpose of any Commodity Investment (such Account being the "Commodity Investment Account").
- 7.2 We may, at any time, prescribe any additional terms and conditions as we may deem appropriate in respect of any Commodity Investment and/or the Commodity Investment Account and/or revise any such terms and conditions.
- 7.3 All Commodity Investments shall be effected in whole units of ounce or gram or appropriate measurement unit of the respective Commodity, subject to such minimum amount as we may determine from time to time.
- 7.4 Subject to the provisions of any Applicable Agreement, we will credit or debit your Commodity Investment Account with the quantity of Commodity purchased, acquired or sold pursuant to each Commodity Investment:
 - (a) within two (2) Business Days from the trade date or such other date as determined by us (in the case of a sale or purchase of Commodities); and
 - (b) on the settlement date(s) specified in the Applicable Agreement (in the case of any other Commodity Investments).
- Any Commodity standing to the credit of your Commodity Investment Account may only be dealt with through us. Unless otherwise agreed by us, you may not withdraw or require the transfer or delivery of any such Commodity in physical form. For the avoidance of doubt, we are under no duty or obligation to hold or deliver to you any such Commodity in physical form.
- 7.6 You shall have no interest, proprietary or otherwise, in any Commodity that may be held by us in physical form in connection with your Commodity Investments.
- 7.7 Subject to the provisions of this Agreement and/or any Applicable Agreement, you may, at any time, withdraw the cash value of any Commodity standing to the credit of your Commodity Investment Account by converting such Commodity into US Dollars or such other currency as may be agreed by us. Such conversions shall be effected at a rate of exchange to be determined by us, in our absolute discretion, based on the then prevailing exchange rate.

- 7.8 You acknowledge that the value of any Commodity standing to the credit of your Commodity Investment Account, as set out in any Account Statement and/or Transaction Record, is only an indication of the market value of such Commodity as at the applicable value date, and may differ from any amount you may receive upon a withdrawal of the cash value of such Commodity in accordance with Clause 7.7.
- 7.9 There shall be no yield or interest payable in respect of any Commodity standing to the credit of your Commodity Investment Account. All determinations and calculations required for any Commodity Investments based on any prevailing exchange rate will be made by us in good faith and in a commercially reasonable manner. In the absence of manifest error or fraud on our part, all such determinations and calculations will be conclusive and binding on you.

8. Fiduciary Placements

- 8.1 You authorise us to act as your agent to place deposits (each, a "Fiduciary Placement") with such banks or other financial institutions in any jurisdiction, including any of our Affiliates (each, a "Third Party Institution"). Each such Fiduciary Placement shall be placed in our name but for and on your behalf and at your sole risk and expense.
- 8.2 We may, in our absolute discretion, determine the Third Party Institution with which a Fiduciary Placement is placed.
- 8.3 Any Instruction for the placement and/or renewal of a Fiduciary Placement must be received by us no later than five (5) Business Days prior to the date of such placement and/or renewal. Such Instruction shall stipulate the tenor, currency and the amount of such Fiduciary Placement. Notwithstanding the foregoing, if we do not receive any Instruction from you for the renewal of any Fiduciary Placement, we are authorised (but not obliged) to renew such Fiduciary Placement on such terms and conditions and for such tenor as we shall deem appropriate.
- 8.4 You acknowledge that a Fiduciary Placement may not be terminated prior to its maturity.
- 8.5 Unless otherwise instructed by you, we shall credit all sums received by us on your behalf from the Third Party Institution in respect of any Fiduciary Placement, less any applicable tax, fee, charge or expense incurred by us, to any Bank Account or make payment of such sums by any means that we consider appropriate. Save as aforesaid, we shall not be liable to repay to you any Fiduciary Placement or any amount otherwise accruing thereto.
- 8.6 For the avoidance of doubt, we shall not be regarded as a trustee of any Fiduciary Placement. In the event that any Third Party Institution does not, for any reason whatsoever, fulfil its obligations in respect of any Fiduciary Placement, our sole obligation is to assign all rights and claims which we hold on your behalf against such Third Party Institution to you. Without prejudice to any provision in this Agreement, we shall not be liable in any way for any Losses suffered or incurred by you in connection with any Fiduciary Placement.
- 8.7 You acknowledge that each Fiduciary Placement and/or the repayment thereof is subject to Applicable Laws, as well as the rules and the terms and conditions of the Third Party Institution with which such Fiduciary Placement is placed (which rules and terms and conditions may be amended from time to time in the absolute discretion of such Third Party Institution) and you agree to comply with all such Applicable Laws, rules and terms and conditions.

PART F – SPECIFIC TERMS AND CONDITIONS GOVERNING SECURITIES TRANSACTIONS

- 1. These Securities Terms shall apply to all Securities Transactions which you may undertake with or through us and all Services which we may offer to you to enable you to undertake any Securities Transaction.
- 2. In the event of any conflict or inconsistency between:
 - (a) these Securities Terms and the Account Terms, these Securities Terms shall prevail; and/or
 - (b) these Securities Terms and any Applicable Agreement in respect of any Securities Transaction, that Applicable Agreement shall prevail.
- 3. Without prejudice to Clause 22 of the Account Terms, we may effect any of your Securities Transactions through such Agents as we may select or otherwise as we deem appropriate. You further agree that we may prescribe such further terms and conditions as we consider necessary in relation to any Securities Transaction.
- 4. You acknowledge that acceptance by us of your Instruction in relation to any Securities Transaction will not necessarily result in its execution. Your Instruction will only be executed if (a) market conditions permit and (b) such execution is in accordance with all Applicable Laws. If we are unable to execute any Instruction, we will inform you as soon as reasonably practicable, whereupon such Instruction shall expire.
- 5. You agree that we may, at our absolute discretion (a) aggregate and consolidate any Instruction received from you in relation to any Securities Transaction with similar instructions received from other customers before communicating them to any Agent for execution and (b) to the extent permitted under Applicable Laws, allot or distribute the Securities purchased or sold arising from such consolidated instructions in any manner as we deem fit. You accept that such allotment or distribution or action by us may result in Losses to you and you accept the risk thereof as being for your account. In the event that the consolidated instructions are only partially executed, you agree that we shall be entitled to allocate fewer units of Securities to you than that as may be instructed by you.
- 6. Unless otherwise specified, all Securities Transactions undertaken on your Instructions shall be executed in our name and/or in the name of our Nominee. All such Securities purchased by you shall be held by us or our Nominee, as custodian for you, and shall be subject to the Custody Terms.
- 7. Initial Public Offerings and Private Placements of Securities
- 7.1 We may, on your Instruction but subject always to the provision and/or execution of such documents and indemnities as we may require, subscribe for Securities on your behalf, pursuant to an initial public offering or private placement. Unless we agree otherwise, each such Instruction shall be irrevocable.
- 7.2 You acknowledge that unless required by Applicable Laws, we are under no obligation to provide you with the prospectus, information memorandum, application forms or other offering document relating to such initial public offering or a private placement of Securities (collectively, the "Offering Documents"). By instructing us to subscribe for such Securities on your behalf, you confirm that you have obtained a copy of the Offering Documents and have read, understood and fully accept all the terms and conditions stated therein.
- 7.3 You shall ensure that you fulfill and comply fully with any investor requirements, conditions of subscription, selling and/or other restrictions specified in the Offering Documents.
- 7.4 Where you have instructed us to subscribe for Securities on your behalf pursuant to any initial public offering or private placement of such Securities, you agree that you shall not make any other application for subscription of the same Securities, whether directly or through any agent or nominee.
- 7.5 Where we are required to make certain representations and/or warranties on your behalf to other third parties in connection with such subscription of Securities, you shall ensure that each such representation and/or warranty shall be true, correct and not misleading in any way and you shall forthwith notify us in writing of any matter arising in the future which may render any such representation and warranty untrue, inaccurate or misleading in any way.
- 7.6 You acknowledge that we are under no obligation to ensure the success of any application for subscription of Securities made on your behalf in any initial public offering or private placement. If any such application (or any part thereof) is successful, you agree that we may, subject to Applicable Laws, allocate the Securities allotted pursuant thereto to our customers in such manner as we deem appropriate and you shall be bound by any such allocation (notwithstanding that it may be fewer than the quantity you had instructed us to subscribe for).

- 8. In respect of each Securities Transaction undertaken on your Instructions, you shall ensure that all Applicable Laws (including any reporting and disclosure requirement and shareholding restriction) are strictly adhered to and complied with at all times and you agree that we need not enquire into or verify any action taken by you in connection therewith. You further warrant and represent that your entry into each Securities Transaction does not contravene any Applicable Law, including Applicable Laws relating to insider dealing, market manipulation and/or any other trading offences.
- 9. For subscriptions into initial public offerings, typically the issuers or the relevant exchanges will impose restrictions or prohibitions on investors who are directors or existing shareholders of issuers or their close associates, or who are connected persons/ clients to the issuer group. Unless you notify us of such (direct or indirect) connections, you are deemed not to be such a 'connected person'.
- 10. You shall make payment of all Liabilities in connection with the Securities Transactions undertaken by you on or before the due date of payment or such other date as may be notified by us to you. Without prejudice to any provision in this Agreement, you authorise us to debit any Bank Account for all such Liability as may be payable by you to us or for any payment in connection with any Securities Transaction.
- 11. Failure to deliver Securities or make payment of Liabilities
- 11.1 When giving Instructions to us for the sale of any Securities, you shall ensure that the relevant Securities shall be delivered to or deposited with us or our Nominee by such date as may be notified by us to you and that you are entitled to sell and deliver such Securities to the purchaser on the due date for settlement. If the Securities are not delivered to or deposited with us or our Nominee by the date as notified by us to you, you acknowledge and agree that we shall, subject to Applicable Laws, have the right to purchase and/or borrow such number of Securities as is equivalent to the number of Securities sold by you, at such time and on such terms as we think fit, for the purpose of effecting such delivery.
- 11.2 You acknowledge and agree that if payment of any Liability under these Securities Terms, including payment for the purchase of any Securities, is not made by the due date of such payment, we shall have the right to force sell all or any of such unpaid Securities at any time and in any manner as we think fit and set off the proceeds thereof against all or part of such Liability.
- 11.3 Any Losses suffered or incurred by us as a result of any action taken by us pursuant to this Clause 10, including any Losses suffered or incurred as a result of any fluctuation in the market price of the Securities, shall be a debt due from you to us repayable forthwith on demand but any profit or gain made shall be forfeited to us and you shall have no claim in respect thereof. For the avoidance of doubt, our rights in this Clause 10 shall be in addition to and without prejudice to any right or remedy which we may have under this Agreement and/or any Applicable Agreement.
- 12. You acknowledge that:
- 12.1 Unless otherwise disclosed to you, we shall be acting as your agent in respect of all Securities Transactions. You are therefore principally liable for, and shall assume all risks (including any counterparty or settlement risk) associated with, all Securities Transactions entered into pursuant to this Agreement and/or any Applicable Agreement, notwithstanding that such Securities Transactions may have been undertaken in our name without disclosure of such agency.
- 12.2 Subject to any limitation or condition prescribed by Applicable Laws, we and/or our Affiliate may, in respect of any Securities Transaction undertaken on your Instruction:
 - (a) be dealing as principal for our or its own account in purchasing from or selling to you Securities;
 - (b) be matching such Securities Transaction with that of another of our customers or our Affiliate;
 - (c) be receiving from any Agent charges, commissions, fees, rebates, discounts or other payments or benefits and you agree that we or our Affiliate (as the case may be) may retain any such payments or benefits and shall be under no obligation to account for or disclose the same to you except to the extent required by Applicable Laws; and
 - (d) have a direct or indirect material interest in any such Securities Transaction.
- 13. Without prejudice to any provision in this Agreement, you acknowledge that we may, from time to time, have to accept sole and principal responsibility to any broker (notwithstanding that between you and us, we act as your agent). You further acknowledge that by reason of the foregoing, the Securities purchased may be regarded by the broker as being Securities purchased by us for our own account and this may result in prejudice to you.

PART G – SPECIFIC TERMS AND CONDITIONS GOVERNING INVESTMENTS AND TRANSACTIONS IN FUNDS

- 1. These Fund Investments Terms shall apply to all Fund Investments which you may undertake through us and all Services which we may offer to you in connection with such Fund Investments.
- 2. In the event of any conflict or inconsistency between:
 - (a) these Fund Investments Terms and the Account Terms, these Fund Investments Terms shall prevail; and/or
 - (b) these Fund Investments Terms and any Applicable Agreement in respect of any Fund Investment or any Service offered to you in connection with any Fund Investment, that Applicable Agreement shall prevail.
- 3. Notwithstanding anything to the contrary in these Fund Investment Terms, we shall not be obliged to provide any of the Services set out herein in relation to any Fund Investment which is not acquired through us. If we agree to assist you in respect of any such Fund Investments, such assistance shall be provided without responsibility on our part and you shall indemnify us against all Losses which we may suffer or incur as a result thereof.
- 4. General
- 4.1 You acknowledge and agree that our distribution of any Fund is not to be seen as a recommendation of such Fund by us. We are not responsible for any Fund or its performance. Unless otherwise stated, you acknowledge that we are not the author of any Fund Offering Document and we shall have no liability whatsoever to you for any error, misstatement or omission in any Fund Offering Document or any other information received by you from the Fund Representative or otherwise on the relevant Fund or any Losses suffered or incurred by you in connection with any Fund Investment or steps taken or omitted to be taken by you on the basis of any Fund Offering Document.
- 4.2 Without prejudice to any provision in this Agreement and/or any Applicable Agreement, you agree that we shall not be liable or responsible for any action, omission, error, neglect, default or delay on the part of any Fund or Fund Representative.
- 4.3 You acknowledge that we may act as distributor, sub-distributor, or placement agent of a Fund or Fund Representative and hence would not be involved in the management / running of the Fund. If you wish to raise any questions or seek further details in respect of any Fund or Fund Investment, you may address such questions to us in writing and we will use reasonable endeavours to obtain a written response to such questions from the relevant Fund or Fund Representatives. Notwithstanding the above, we shall not be obliged to obtain satisfactory written responses to your questions and you agree that we shall have no liability for failing or omitting to do so.
- 4.4 You acknowledge that we may receive from the Fund or any Fund Representative commissions, fees or other payments in respect of your Fund Investments. You agree that we shall be entitled to retain such commissions, fees or other payments and shall not be liable to account to you for the same.
- 5. Acquisition and Custody of Fund Investments
- 5.1 At your request, we may (but shall not be obliged to) act as your agent in placing any order for the acquisition of any Fund Investment.
- 5.2 All Fund Investments shall be issued to, registered and held in our name and/or in the name of our Nominee for and on your behalf and shall be subject to the Custody Terms.
- 5.3 You hereby undertake, represent and warrant to us that:
 - (a) you shall promptly execute any subscription agreement, Applicable Agreement or other document that may be required by us, the Fund or the Fund Representative;
 - (b) you have received and read the Fund Offering Documents and fully understand the structure of and the terms and conditions applicable to the Fund Investment (including any restrictions on redemption and capital commitment) and the risks associated therewith;

- (c) you shall ensure that you fulfill and comply fully with any investor requirements, conditions of subscription, selling and/or other restrictions specified in the Fund Offering Documents;
- (d) we and/or our Nominee are authorised to execute all agreements and other documents (including any subscription agreement) and generally do all such acts and exercise such discretion as we or our Nominee consider expedient or necessary in connection with any Fund Investment; and
- (e) where we and/or our Nominee are required to provide any representation and warranty to the Fund or any Fund Representative on your behalf, you shall ensure that each such representation and warranty shall be true, accurate and not misleading as if the same is provided by you and you shall forthwith notify us in writing of any matter arising in the future which may render any such representation and warranty untrue, inaccurate or misleading in any way.
- 5.4 You agree that we and/or our Nominee may aggregate and consolidate your orders for the acquisition of any Fund Investment with orders received from other customers, either daily or from time to time, before transmitting such aggregated order to the Fund or the Fund Representative.
- You acknowledge and agree that, to the extent permitted under Applicable Laws, we may, in our absolute discretion, allot or distribute any Fund Investment issued to us pursuant to such aggregated order amongst you and other customers in any manner as we may deem appropriate and you shall be bound by any such allotment or distribution (notwithstanding that it may be less than the number of units specified in your order). You accept that such allotment or distribution by us may result in Losses to you and you accept the risk thereof as being for your account.
- You acknowledge that acceptance of any order placed on your behalf for the acquisition of any Fund Investment is at the absolute discretion of the Fund or, as the case may be, the Fund Representative and we are under no obligation to ensure the same. If any such order is rejected (whether in whole or in part) by the Fund or, as the case may be, the Fund Representative, any payment made for the Fund Investment shall be returned to you, without interest, within a reasonable period of time and in such manner as we shall deem appropriate.
- 5.7 We shall, or shall procure our Nominee to, record and hold in a separate account in our or our Nominee's books, all Fund Investments received and held by us or our Nominee from time to time for your account. In this regard, you acknowledge and agree that we shall be entitled, without prior notice to or consent from you and without any liability on our and/or our Nominee's part, round down the number of units of any Fund Investment held for your account in our books to two decimal places in the event the actual number of units of such Fund Investment is more than two decimal places. In such circumstances, you agree to waive and relinquish in our favour all claims for any shortfall in the number of units of Fund Investments that may arise (if any) due to such rounding down.
- 5.8 You acknowledge that:
 - (a) we may rely on valuations from the Fund, the Fund Representative and/or other third parties for the purposes of reporting to you the value of any Fund Investment; and
 - (b) we are under no duty to seek to verify the accuracy or otherwise of such valuations. Any price or value that we may provide to you in respect of the Fund Investment is not final and binding and is only indicative and we shall not be responsible or liable for any Losses whatsoever and howsoever arising from any use of or reliance on the same.
- 6. Redemptions and Distributions
- 6.1 You acknowledge that any redemption of your Fund Investments may only be made in accordance with the terms of the Fund Offering Documents.
- 6.2 Unless otherwise notified to us, we may credit any redemption proceeds, dividends and other income and monies in relation to any Fund Investment received by us on your behalf into any Bank Account or make payment of the same by any means that we consider appropriate. The proceeds, dividends, income and monies credited to any Bank Account or otherwise paid to you shall be net of any applicable taxes, fees, charges or expenses incurred by us and you agree that we shall not be required to ascertain, or otherwise be responsible for the adequacy of such redemption proceeds, income or monies received.

- 7. Switching
- 7.1 You acknowledge that any switching or exchange of your Fund Investments may only be effected in accordance with the terms of the Fund Offering Documents.
- 7.2 If you instruct us to effect a switch of any Fund Investments, we will subscribe for the Fund Investments required by you only after the confirmation and completion of the redemption of the relevant existing Fund Investments which are being switched.
- 8. Transfer of Fund Investments
- Unless otherwise provided in these Fund Investments Terms or in any Applicable Agreement, any transfer of your Fund Investment, if permitted under the Fund Offering Documents, shall be subject to our prior approval and on such terms as we may, in our absolute discretion, specify. Without prejudice to the foregoing, you shall ensure compliance with the procedures (if any) set out in the Fund Offering Documents, including obtaining any approval (whether from the board of directors of the Fund, the Fund Representative or otherwise) as may be required for such transfer, and neither we nor our Nominee shall be under any obligation in this regard.
- You further acknowledge and agree that we shall not be obliged to arrange for the transfer of any Fund Investment until all your Liabilities under these Fund Investment Terms (including any outstanding fees and expenses payable) are fully discharged.
- 9. Fund-initiated Transactions
- 9.1 If, at any time, for any reason whatsoever, the Fund or any Fund Representative instructs us or our Nominee, as registered holder of any Fund Investment, to divest, transfer or otherwise dispose of any such Fund Investment, we shall, subject to these Fund Investment Terms and/or any Applicable Agreement, seek your Instructions as to how you wish to proceed. In the absence of any Instruction from you in due time or if a satisfactory course of action cannot be agreed with the relevant Fund Representative or Fund within any time period specified for this purpose, we shall, subject to the terms of the Fund Offering Documents, redeem or procure our Nominee to redeem, the relevant Fund investment and any redemption proceeds received in connection therewith shall be paid to you in accordance with these Fund Investments Terms.
- 10. Commitment Calls by the Fund
- 10.1 You acknowledge that certain Funds may require you to make capital contributions from time to time up to the amount you have committed to invest in the Fund. You shall ensure that there are sufficient funds in your Account to satisfy all such commitment calls. Failure to meet any such commitment calls shall constitute a Liability owing by you to us as your nominee and/or agent.
- 10.2 If there are insufficient funds in your Account as of the deadline stipulated in any drawdown notice issued by us and/or if you fail to make the required capital contribution in respect of any commitment call, you agree that we may take such action as we consider necessary to preserve our rights and interests, including and not limited to the following transferring your Fund Investment to be held directly by you in your name without your prior consent, taking such action as may be provided for in any Applicable Agreement, and/ or treating such failure as an Event of Default which entitles the Bank to enforce its rights upon an Event of Default as set out in these Account Terms, including but not limited to:
 - (a) taking such action as we think necessary or appropriate to cancel, settle, redeem or terminate any outstanding Transactions between us as principal or with any third party entered into by us as your agent;
 - (b) sell, realise, assign, transfer or otherwise dispose of any of your Investments or Assets held by us in such manner and on such terms as we may think fit (in no particular order or priority) and to apply the net proceeds thereof (after deduction of any expense incurred in connection therewith) in or towards satisfaction of your Liabilities to us; and/or
 - (c) exercise our rights of appropriation, debit, set-off and/or consolidation of accounts in accordance with Clause 28 of Part B General Terms and Conditions Governing Accounts and Services.

You are aware and agree that concurrently or as a result thereof, your Account may fall into a margin shortfall and that the Bank is also entitled to take all necessary action including pursuant to the Applicable Agreement in relation to any credit or margin facility that you have entered into.

- 10.3 In the event that a capital commitment call results in your Account becoming overdrawn or exceeding any overdraft limit (whether pursuant to the Bank's debiting of your Account or otherwise), interest and bank charges shall be computed on such overdrawn amount at such rate and on such basis as we may in our absolute discretion determine. The overdrawn amount and all accrued charges and interest shall be repayable by you to us on demand.
- 10.4 The rights conferred on us in this Clause 10 are in addition to, and shall not prejudice any other rights or remedies which we may have under this Agreement and/or any Applicable Agreement.
- 11. Disclosure of Information
- 11.1 You agree that we and our Nominee may, upon any request by the Fund and/or the Fund Representative, disclose any Customer Data (including your identity and/or the identity of any beneficial owner(s) of any Fund Investment) to such person(s) (including the Fund, the Fund Representative or any government, quasi-government, regulatory, fiscal, monetary or other authority or agency) and for such purpose(s) in connection with your Fund Investment as may be specified in such request.
- 11.2 The rights conferred on us in this Clause 11 are in addition to, and shall not prejudice, any other rights that we may have under this Agreement, any Applicable Law or any Applicable Agreement.

PART H – SPECIFIC TERMS AND CONDITIONS GOVERNING GOLD BULLION SERVICES

1. Agreement for Gold Bullion Services

- 1.1 These Gold Bullion Terms shall apply to the Gold Bullion Services which we may make available to you at your request.
- 1.2 In the event of any conflict or inconsistency between:
 - (a) these Gold Bullion Terms and the Account Terms, these Gold Bullion Terms shall prevail; and/or
 - (b) these Gold Bullion Terms and any Applicable Agreement in respect of Gold Bullion Services, that Applicable Agreement shall prevail.

2. General

2.1 Gold Bullion Service Fees

- (a) You agree to pay to us such service fees (the "**Gold Bullion Service Fees**") at the rates and in such manner as we may prescribe from time to time in respect of the Gold Bullion Services.
- (b) The Gold Bullion Service Fees may be subsumed under a single price or rate quoted to you, in connection with a Gold Bullion Transaction (including the price quoted in the purchase and sale of the Gold Bullion) part of which includes ours and any Agent's fees, charges or commissions, and payment of which will be deducted from such price or rate.
- (c) The Gold Bullion Service Fees do not include or provide insurance coverage on the Gold Bullion unless otherwise expressly stated.
- (d) Unless otherwise disclosed to you, we act as counterparty to you, and deal with you as principal, for any Gold Bullion Service or in any Gold Bullion Transaction which you have instructed us to effect. We shall be absolutely entitled to all gains, profits and benefits derived from any Gold Bullion Service or Gold Bullion Transaction, including any spread on the Gold Bullion Service or Gold Bullion Transaction entered into with you and corresponding or back-to-back arrangements or transactions entered into by us with third party brokers or counterparties.

2.2 Currency

The reference currency for all Gold Bullion Transactions and payments pursuant to these Gold Bullion Terms shall be US Dollars, unless we otherwise notify you in writing.

2.3 Representations and Warranties

- (a) In addition to the representations and warranties in Clause 23 of the Account Terms, you hereby represent and warrant to us, as a continuing warranty, that:
 - (i) you are the beneficial owner of the Gold Bullion that may be the subject of these Gold Bullion Terms, free and clear from any and all contingent or existing charges, pledges, mortgages, security interests, encumbrances, liens or other rights or claims whatsoever permitted or created by you or any third party;
 - (ii) you acknowledge that we shall not be bound by or compelled in any way to recognise (even when we have notice thereof) any equitable, contingent, future or partial interest in any Gold Bullion except (where applicable) your absolute right to the entirety thereof; and
 - (iii) the receipt, handling or the delivery of the Gold Bullion will not expose us or our Personnel, insurers, insurance brokers, reinsurers, Agents or Affiliates to any sanction, prohibition or penalty (or any risk of sanction, prohibition or penalty) whatsoever imposed by any relevant state, country/region, international governmental organisation or other relevant authority.

(b) For the purpose of Clause 5 of these Gold Bullion Terms, the representations and warranties in Clause 2.3(a) shall, where applicable, be deemed to apply with the necessary modifications to Precious Metals.

2.4 Acknowledgements

In relation to the Gold Bullion Services offered by us, you hereby acknowledge that:

- (a) we are under no obligation whatsoever to ascertain, check or confirm the purity or authenticity of the Gold Bullion;
- (b) prior to accepting any offer we make and entering into these Gold Bullion Terms, you have read and understood all these Gold Bullion Terms;
- (c) our obligations are limited to those expressly set out in these Gold Bullion Terms;
- (d) you have not entered into these Gold Bullion Terms in reliance on any representation or warranty we, or any person associated with or representing us, make;
- (e) except as we may expressly agree to in writing, we are not acting as your fiduciary in providing the Gold Bullion Services to you. For the avoidance of doubt, neither these Gold Bullion Terms nor the provision of any Gold Bullion Service to you will give rise to any fiduciary or equitable duty on our and/or our Affiliates' part that would prevent us and/or our Affiliates from carrying on any business or doing business with or for our respective customers;
- (f) your acceptance of these Gold Bullion Terms in connection with the provision of Gold Bullion Services is based on us acting as an independent contractor and principal and not in any other capacity, unless otherwise stated in writing; and
- (g) nothing in these Gold Bullion Terms nor our receipt of confidential information nor any other matter in respect of the provision of any Gold Bullion Services shall be construed to give rise to any fiduciary or equitable duty of a nature which would prevent or hinder us and/or our Affiliates from acting as market maker, market dealer, broker, principal or agent, whether such business is for our account or not, while doing business with or for:
 - (i) you;
 - (ii) our Affiliates; and/or
 - (iii) any of our customers or investors.

2.5 Undertakings

In addition to the undertakings in Clause 23 of the Account Terms, you hereby undertake to us, by way of a continuing undertaking, to provide such further information as may reasonably be required by us or any governmental or quasi-governmental organisation with respect to the Gold Bullion. You acknowledge that the Gold Bullion may be subject to security controls of government or quasi-government entities, and that we will submit to all required security controls. On behalf of yourself and all others with a legal interest in the Gold Bullion, you hereby consent to a search of any Gold Bullion under these Gold Bullion Terms, if and as may be required by any government or quasi-government entity wherever the Gold Bullion may be located during or as a result of the performance of any Gold Bullion Service.

2.6 Provision of information

(a) You shall on request provide details in writing to us, our Affiliates, Agents and/or regulatory or governmental bodies of the person(s) originating any Instruction as soon as possible, but in any case within one (1) Business Day or any other period as required by the relevant regulatory or governmental body having jurisdiction (including the MAS), whether or not the requirement has the force of law and consents to our disclosures in respect of you, the Account and/or any Instruction, to any such person or body have been given.

(b) You hereby agree to sign, execute and/or complete such documents, certificates or forms as we may require from time to time for the transport of the Gold Bullion or for any purpose in connection with the Gold Bullion Transactions, and hereby agree to the disclosure of any information relating to the Gold Bullion Services, the Gold Bullion Transactions or you to any relevant agent or third party.

2.7 Exclusion and Limitation of Liability

- (a) Without prejudice to Clause 31 of the Account Terms, we shall not be liable under any circumstance whatsoever, for any Losses arising from or in connection with any of the following:
 - (i) indirect, consequential, or special Losses of or damage to the Gold Bullion, including loss of profits, revenue, business opportunity or reputation, business interruption, loss of market, whether or not caused by our fault or neglect and whether or not we had knowledge that such Loss or damage might be incurred;
 - (ii) any pre-existing defect or nature of the Gold Bullion as we may note and promptly notify you in writing of the same;
 - (iii) shortage or disappearance or loss from or damage to the Gold Bullion, unless:
 - (aa) resulting from our gross negligence, fraud or wilful default when providing the Gold Bullion Services under these Gold Bullion Terms; or
 - (bb) as a result of the Gold Bullion being opened and inspected by customs officials, authorities or other governmental or quasi-governmental authority;
 - (iv) the:
 - (aa) ordinary leakage, ordinary loss in weight or volume, shrinkage, or ordinary wear and tear; or
 - (bb) inherent vice, defect or nature of the Gold Bullion;
 - (v) incorrect or insufficient description or valuation of the Gold Bullion whereby our liability is reduced or extinguished;
 - (vi) any movement or change in the value of the Gold Bullion purchased pursuant to these Gold Bullion Terms;
 - (vii) war, civil war, revolution, rebellion, insurrection, or civil strife arising there from, or any hostile act by or against a belligerent power; or capture, seizure, arrest, restraint or detainment, arising from risks described in this Clause 2.7(a)(vii), and the consequences thereof or any attempt thereat;
 - (viii) derelict mines, torpedoes, bombs or other derelict weapons of war;
 - (ix) any weapon or device employing atomic or nuclear fission and/or fusion or other like reaction or radioactive force or matter;
 - (x) ionising radiation from or contamination by radioactivity from any nuclear fuel or from any nuclear waste or from the combustion of nuclear fuel;
 - (xi) the radioactive, toxic, explosive or other hazardous or contaminating properties of any radioactive matter, nuclear installation or any, reactor or other nuclear assembly or nuclear component thereof provided that this sub-clause does not extend to radioactive isotopes, other than nuclear fuel, when such isotopes are being prepared, carried, stored, or used for commercial, agricultural, medical, scientific or other similar peaceful purposes; or
 - (xii) any chemical, biological, bio-chemical or electromagnetic weapon;

- (xiii) any act of terrorism being an act of any person acting on behalf of, or in connection with, any organisation which carries out activities directed towards the overthrowing or influencing, by force or violence, of any government whether or not legally constituted, or any person acting from a political, ideological or religious motive; or
- (xiv) confiscation, seizure, appropriation, expropriation, requisition for title of use or wilful destruction of the Gold Bullion, or portion thereof, by/or under the order of any government (whether civil, military or de facto) and/or public authority.
- (b) Without prejudice to any provision of this Agreement and/or any Applicable Agreement which excludes us from liability and notwithstanding any provision of this Agreement and/or any Applicable Agreement which imposes liability on us, our liability for any Losses that you may suffer shall be limited to the maximum amount recovered from an Agent whom we may enter into a back-to-back arrangement with for the purposes of carrying out the Gold Bullion Services, and to the extent the amount recovered from such Agent is referable to the Losses of more than one of our customers, such recovered amount shall be pro-rated amongst our customers in the manner we deem fit.
- (c) Any claim arising in respect of any Gold Bullion Service provided or undertaken to be provided to you under these Gold Bullion Terms must:
 - (i) be notified to us within one (1) Business Day of the discovery of the event or occurrence alleged to give rise to such a claim; and
 - (ii) be made in writing to us within a reasonable time but in any event no later than fifteen (15) days from the date of discovery of the event or occurrence alleged to give rise to such a claim.
- (d) In addition and without prejudice to any provision of this Agreement and/or any Applicable Agreement, we shall in any event be discharged of all liabilities whatsoever and howsoever arising in respect of any Gold Bullion Services we provide to you, or which we have undertaken to provide, unless legal proceedings have been brought and written notice thereof given to us within one (1) year from the date of:
 - (i) (in relation to any claim relating to the Gold Custody Services) our notification to you of an event or occurrence alleged to give rise to such claim; or
 - (ii) (in relation to all other claims) the happening of event or occurrence alleged to give rise to such claim.

2.8 Indemnity

- (a) You shall promptly indemnify us against any claim, demand, action and proceeding that may be brought against us and all Losses which may be incurred or suffered by us or our Personnel, Agents, Affiliates, insurers, insurance brokers or re-insurers (where applicable) (collectively, "Aggrieved Parties") as a result of or arising from or otherwise in connection with:
 - (i) your breach of any of the warranties and/or undertakings given under these Gold Bullion Terms;
 - (ii) our reasonable exercise, in good faith, of our rights pursuant to these Gold Bullion Terms; or
 - (iii) us properly acting in accordance with your Instructions.

The indemnity in this Clause 2.8(a) shall include all Losses resulting from claims against us by an Aggrieved Party in connection with the Losses arising from this Clause 2.8(a), to the extent that the said Losses have arisen as a direct result of the causes specified in Clauses 2.8(a)(i), 2.8(a)(ii) or 2.8(a)(iii).

- (b) In the event that we are instructed to clear the Gold Bullion through customs, you shall be liable for any duty, tax, fee, levy, charge or outlay of whatsoever nature levied by any applicable authority for or in connection with the Gold Bullion or the Gold Bullion Services and shall indemnify us for the same, save where the same was incurred as a direct result of our, or our Personnel's, Agents' or Affiliates' gross negligence, fraud or wilful default.
- (c) Save where directly caused by our, and/or our Personnel's, Agents' or sub-contractors' gross negligence, fraud or wilful default, you shall promptly indemnify us against any and all Losses assumed, incurred or suffered by us, our Personnel, Agents, sub-contractors, insurers, insurance brokers, re-insurers, as a result of any claim made by the owner of the Gold Bullion or any other person who is or may become interested in the Gold Bullion if it is not you or any customs authority against us (each, a "Third Party Claim") when the Third Party Claim arises from or in connection with the Gold Bullion Services, except when the Third Party Claim arises from death, personal injury or damage to property other than the Gold Bullion.
- (d) Under these Gold Bullion Terms, you hereby acknowledge and recognise that we may enter into a back-to-back arrangement with an Agent for the purposes of carrying out the Gold Bullion Services. You hereby agree to indemnify us against any claim such Agent may have against us under the back-to-back arrangement, provided that such claims are not a direct result of our breach under these back-to-back arrangements.
- (e) For the avoidance of doubt, this Clause 2.8 shall be in addition to and without prejudice to any provision in this Agreement.

2.9 Availability of Gold Bullion Services

- (a) We have no obligation to provide any Gold Bullion Service to, or enter into any Gold Bullion Transaction with, you or on your behalf, and we may, in our absolute discretion, refuse to do so. We shall not be obliged to provide any reason for such refusal, suspension, termination or discontinuance. If we decline to execute a Gold Bullion Transaction, we shall promptly notify you of the same. We shall not be liable for any Losses that you may incur if we (i) fail to comply with any of your order or Instruction, or (ii) exercise or fail to exercise any discretion, power or authority conferred upon us by these Gold Bullion Terms, unless failure to do so is a direct result of our or our Personnel's gross negligence, fraud or wilful default.
- (b) Our obligation to settle any Gold Bullion Transaction is conditional upon our receipt, or the satisfactory confirmation of our settlement agents' receipt, of all necessary documents, funds or Gold Bullion (as the case may be).

2.10 Limits

Without prejudice to any provision in this Agreement, we may, in our absolute discretion, apply a limit (such limit as may be amended by us from time to time) on:

- (a) the size of any Gold Bullion Transaction or series of Gold Bullion Transactions which you may enter into; and/or
- (b) the amount of Losses or liabilities which you may be exposed to.

2.11 Execution-only

- (a) Without prejudice to any provision of this Agreement, we shall not provide investment advice as a Gold Bullion Service and shall not act as your adviser in relation to any Gold Bullion Transaction, and any Gold Bullion Transaction will be effected on an execution-only basis.
- (b) You acknowledge that in entering into a Gold Bullion Transaction on an execution-only basis:
 - (i) all decisions on whether to invest in, hold or dispose of any Gold Bullion or to enter into any Gold Bullion Transaction are yours only;

- (ii) you are deemed to have made an independent analysis and decision with respect to all Gold Bullion Transactions and all dealings in any Gold Bullion Transaction shall be deemed to be undertaken by you in reliance only upon your own judgment and not in reliance upon any view, representation (whether written or oral), advice, recommendation, opinion, report, analysis, research, materials, information or other statement by us or our Personnel, Agents or Affiliates;
- (iii) you are aware that we do not hold out any of our Personnel, Agents or Affiliates as having any authority to advise you and we do not purport to advise you on the terms of, or any other matters connected with, any Gold Bullion Transaction; and
- (iv) you are deemed to have informed yourself of the characteristics of your Account, including holding or transfer restrictions, foreign ownership limitations or requirements of ownership disclosure.

2.12 Anti-Money Laundering and Countering the Financing of Terrorism

- (a) You acknowledge that we and our Affiliates are required to act in accordance with the Applicable Laws which relate to the prevention of money laundering, terrorist financing and the provision of financial and other services to any persons or entities which may be subject to sanctions. We and/or our Affiliates may take any action in accordance with or by reference to all such Applicable Laws.
- (b) In order to comply with Applicable Laws on anti-money laundering, countering the financing of terrorism as well as our policies, reporting requirements under Applicable Laws on financial transactions and any requests of authorities, we may be:
 - (i) prohibited from entering into or concluding certain Gold Bullion Transactions; and/or
 - (ii) required to report suspicious Gold Bullion Transactions to an authority.
- (c) We and/or our Affiliates may intercept and investigate any payment message and other information or communications sent to or by you or on your behalf and may delay, block or refuse to make any payment and such payment screening may cause a delay in processing certain information.
- (d) Neither we nor our Affiliates shall be liable for any Losses arising out of any action taken or any delay or failure by us or our Affiliates, in performing any of our duties or other obligations, caused in whole or in part by any step taken as set out above.

2.13 Authorisation

You hereby authorise us, with your prior Instruction, to sign on your behalf and in your name any client agreement or other document of any Affiliate to enable or better enable us to carry out your Instructions in relation to any Gold Bullion Transaction.

3. Gold Custody Services

3.1 General

- (a) This Clause 3 shall apply to you if, at your request, we agree to provide the Gold Custody Services to you.
- (b) For the avoidance of doubt, Clause 20 of the Account Terms and the Custody Terms shall not apply to the Gold Custody Services.

3.2 Custody of Gold Bullion

- (a) You hereby agree to appoint us, and we agree to act, as the custodian of the Gold Bullion and to provide the Gold Custody Services on the terms and conditions set out in this Clause 3.
- (b) The Gold Bullion shall be held at such designated facility as we may make known to you from time to time (the "**Designated Facility**").

- (c) All Gold Bullion held at the Designated Facility shall be held on an Allocated Basis.
- (d) We may at any time in our absolute discretion prescribe the relevant procedures in relation to the deposit and/or withdrawal of the Gold Bullion under these Gold Bullion Terms.
- (e) Notwithstanding any provision of this Agreement, there shall be no yield or interest payable on the Gold Bullion held in custody by us under these Gold Bullion Terms.

3.3 Sub-custodians

- (a) We may, without any prior notice to or approval from you, appoint one or more Agents as subcustodians to hold in custody the Gold Bullion.
- (b) You irrevocably appoint us to be your attorney and on your behalf and in your name or otherwise to execute and do all such assurances, deeds, acts and things which you may or ought to do in your name and on your behalf to exercise all or any of the rights, powers, authorities and discretions conferred by these Gold Bullion Terms on us to seal and deliver and otherwise perfect any deed, assurance, agreement, instrument or act which we may deem proper in or for the purpose of exercising any of such rights, powers, authorities and discretions and you hereby ratify and confirm and agree to ratify and confirm whatsoever any such attorney shall do or purport to do by virtue of these Gold Bullion Terms.
- (c) We shall exercise due skill, care and diligence in the selection, appointment and periodic review (such review to be carried out in our absolute discretion) of the Agents and the arrangements for the holding and safekeeping of the Gold Bullion, in accordance with Applicable Laws.
- (d) For so long as the Gold Bullion remain deposited with the Agent(s), this Clause 3 shall be subject to, and you agree to be bound by, the administrative and such other procedures and the relevant provisions (including any agreement that we may enter into with the Agent(s)) relating to the provision of custodial services by such Agent(s).
- (e) You acknowledge that any arrangement that we may have with any Agent may be subject to terms and conditions which may include limitations and exclusions for the benefit of such Agent. You understand and acknowledge that you have no direct contractual relationship with the Agent and shall not have any right of recourse against such Agent.

3.4 Our right to transfer the Gold Bullion

- (a) We may, in our absolute discretion, transfer the Gold Bullion from the original Designated Facility to another vault within the same country/region where the original Designated Facility is located.
- (b) In the event that any hazard, impairment or circumstance beyond our control presents a risk of loss or damage to the stored Gold Bullion or otherwise renders it desirable to remove the stored Gold Bullion to other facilities, or if the original Designated Facility where the Gold Bullion are stored shall no longer have the necessary storage capacity, we may transfer, and you hereby authorise and consent to such transfer (at no additional charge to you insofar as storage space is available) of the stored Gold Bullion to any other Designated Facility within the same country/region where the original Designated Facility is located.
- (c) Subject to Clause 3.10, you hereby agree that upon termination of the Gold Bullion Services, we shall have an absolute and unqualified right to require you to remove the Gold Bullion from a Designated Facility at your sole expense, and that we shall also have an absolute and unqualified right to exercise the enforcement of a lien on the stored Gold Bullion in accordance with the law where the Designated Facility is located for settlement of all Gold Bullion Service Fees owed to us for Gold Custody Services, including reasonable legal fees we incur in the exercising of such lien. You shall also hold us harmless and release us from all claims, obligations, suits, actions or causes of actions or legal fees which you may incur, in relation to or arising from our exercise of such rights and the final disposition of the Gold Bullion under this Clause 3.4.

(d) You hereby authorise us, as your agent, at any time and for any reason to procure the transfer to us of all of your right and title to some or all of the Gold Bullion we hold under these Gold Bullion Terms (the "Transferred Portion") in exchange for our transfer to you of the same number of substitute bars of like quality of the relevant Gold Bullion (the "Substituted Portion"), by removing the records identifying the Transferred Portion and simultaneously recording the specific Gold Bullion identified by the serial numbers of the relevant bars (or by other appropriate means) comprising the Substituted Portion.

3.5 Your right to transfer the Gold Bullion

- (a) You may at any time give us Instructions to accept for deposit the Gold Bullion. Such deposit shall be on such terms and conditions that we may impose from time to time.
- (b) You may at any time give us Instructions to withdraw the Gold Bullion, subject to any lien or charge we have over the Gold Bullion.
- (c) For the avoidance of doubt, we shall have the right not to process any Instruction received under this Clause 3.5 and shall not be liable for any Losses that you may suffer as a result thereof. In such an event, we shall notify you of such refusal as soon as practicable.
- (d) We shall be given a reasonable time to carry out any withdrawal Instruction in accordance with Clause 3.5(b) above. If we are unable to do so because of any Event of Force Majeure, or because of loss or destruction of Gold Bullion, then we shall not be liable for the failure to carry out such withdrawal Instructions, and the Gold Bullion remaining in storage will continue to be subject to the Gold Bullion Service Fees.

3.6 Transport of Gold Bullion

The transportation, delivery and all matters incidental thereto of the Gold Bullion to and from the Designated Facility for all inward and outward transfers of Gold Bullion shall be effected at your sole responsibility, expense and risk.

3.7 Visit to Designated Facility

- (a) Upon your request in writing and subject to our prior written approval, we shall permit you access to a Designated Facility, for the purpose of viewing your Gold Bullion held at such Designated Facility.
- (b) For the avoidance of doubt, each such visit shall be subject to such conditions as we may impose on you from time to time.

3.8 Insurance

You hereby acknowledge that we do not and are not required to maintain any type of insurance over the Gold Bullion that is the subject of the Gold Custody Services.

3.9 Indemnity

Without prejudice to any provision of this Agreement and/or any Applicable Agreement, you undertake to indemnify us against any claim that any Agent may have against us, provided that such claims are not a direct result of our gross negligence, fraud or willful default.

3.10 Termination of Gold Custody Services

- (a) Either you or we shall have the right to terminate the provision of Gold Custody Services at any time on giving not less than fourteen (14) days written notice to the other party.
- (b) Where you have given the notice of termination in accordance with Clause 3.10(a) above, you shall forthwith notify us in writing of all necessary arrangements for the delivery to you of the Gold Bullion before the expiry of the relevant notice period set out in Clause 3.10(a) above (the "**Notification Period**"), and such delivery shall take place no later than the date falling ninety (90) days from the date of expiry of the Notification Period (both dates inclusive).

- (c) Where we have given the notice of termination in accordance with Clause 3.10(a) above, you shall notify us in writing of all necessary arrangements for the delivery to you of the Gold Bullion within sixty (60) days from the date of such notice of termination (also the "**Notification Period**"), and such delivery shall take place no later than the date falling ninety (90) days from the date of expiry of the Notification Period (both dates inclusive).
- (d) For the avoidance of doubt, for as long as the Gold Bullion have not been removed from the Designated Facility, the Gold Bullion shall continue to be held by us in accordance with these Gold Bullion Terms and any applicable Gold Bullion Service Fee shall continue to be chargeable.
- (e) Where you (i) fail to make and implement the arrangements required under Clause 3.10(b) or 3.10(c) (as the case may be) or (ii) make arrangements which are not acceptable to us (as we may determine in our absolute discretion) under Clause 3.10(b) or 3.10(c) (as the case may be), we shall, at any time commencing from the date falling ninety (90) days from the last day of the relevant Notification Period (both dates exclusive such that the date will fall on the 91st day), have the option to carry out either of the following:
 - (i) to continue to hold the Gold Bullion in accordance with the terms of these Gold Bullion Terms, and any applicable Gold Bullion Service Fee shall continue to be chargeable; or
 - (ii) to sell the Gold Bullion on such terms as we may determine in our absolute discretion and account to you for the proceeds after deducting any amount due to us under these Gold Bullion Terms.
- (f) We shall not be liable to you for any Losses you suffer as a result of:
 - (i) our exercise of our right to sell the Gold Bullion under Clause 3.10(e)(ii) above;
 - (ii) any fall in the market price of the Gold Bullion between the time the right to sell the Gold Bullion arises under Clause 3.10(e)(ii) and the time we actually sell the Gold Bullion; or
 - (iii) any failure to exercise our right to sell the Gold Bullion under Clause 3.10(e)(ii) above.
- (g) For the avoidance of doubt, our provision of any Gold Bullion Services under these Gold Bullion Terms shall cease upon the termination of the Gold Custody Services, unless we, in our absolute discretion, agree otherwise.

4. Gold Trading Services

4.1 General

- (a) This Clause 4 shall apply to you if, at your request, we agree to provide the Gold Trading Services to you.
- (b) We will provide the following Gold Trading Services to you in accordance with your Instructions:
 - (i) to sell Gold Bullion to you; and
 - (ii) to purchase Gold Bullion from you.
- (c) The Gold Bullion that is the subject of the Gold Trading Services shall at all times also be held pursuant to the Gold Custody Services that we provide, in accordance with these Gold Bullion Terms.
- (d) The minimum or maximum quantity of Gold Bullion that can be purchased or sold in any single Gold Trading Transaction with respect to the Gold Trading Services shall be such quantity as we may, in our absolute discretion, determine from time to time.
- (e) We will enter into Gold Trading Transactions with you as principal.
- (f) You shall be deemed to be transacting as principal for your own account in all Gold Trading Transactions and not as agent or trustee for any third party.

- (g) You accept that we do not provide any advice in relation to any Gold Trading Transaction, and in such circumstances, you are required to seek all necessary independent financial and other advice prior to transacting with us in respect of any Gold Trading Transaction.
- (h) In addition and without prejudice to any provision of these Gold Bullion Terms, you acknowledge that:
 - (i) in the case of a Purchase Request, we shall fulfil our obligations in delivering Gold Bullion to you by entering into a Corresponding Purchase Transaction with a third party; and
 - (ii) in the case of a Sale Request, we shall fulfil our obligations in purchasing Gold Bullion from you by entering into a Corresponding Sale Transaction with a third party.

4.2 Purchase of Gold Bullion from us

- (a) You may give us Instructions to request for the purchase of Gold Bullion from us (each, a "**Purchase Request**"). You shall be bound by such Purchase Request.
- (b) Each Purchase Request shall specify the quantity of Gold Bullion that you are seeking to purchase from us.
- (c) We may (but shall not be obliged to) accept your Purchase Request and sell to you the specified quantity of Gold Bullion at such prices as agreed between you and us. You shall not be entitled to cancel your Purchase Request. All of your purchases of Gold Bullion from us shall be settled in accordance with Clause 4.4.
- (d) All determinations and calculations made by us in connection with the Purchase Request shall be based on any prevailing applicable rate (including spot gold prices) and such determinations and calculations will be made by us in good faith and in a commercially reasonable manner. In the absence of any manifest error or fraud on our part, all determinations and calculations will be conclusive and binding on you.
- (e) Notwithstanding the above, we reserve the right to reject any Purchase Request we receive from you in our absolute discretion, without providing any reason for such rejection. In such an event, we shall notify you as soon as practicable of such rejection.
- (f) Upon the purchase of Gold Bullion pursuant to a Purchase Request, we shall provide you written confirmation of the same (each, a "**Purchase Confirmation**").

4.3 Sale of Gold Bullion to us

- (a) You may give us Instructions to request that we purchase Gold Bullion from you (each, a "Sale Request"). You shall be bound by such Sale Request.
- (b) We shall at all times reserve the right to determine in our absolute discretion the specific bars of Gold Bullion to be removed from storage for the purpose of a Sale Request, provided always that the Gold Bullion removed from storage shall always be of the quantity as may be specified under a Sale Request.
- (c) Each Sale Request shall specify the quantity of Gold Bullion you are seeking to sell to us, and the abovementioned specified quantity of Gold Bullion shall not exceed the quantity of Gold Bullion we hold for you under the Gold Custody Services.
- (d) We may (but shall not be obliged to) accept your Sale Request and purchase from you the specified quantity of Gold Bullion at such prices as agreed between you and us. Upon our acceptance of the Sale Request, you shall not be entitled to cancel the Sale Request. All of your sale of Gold Bullion to us shall be settled in accordance with Clause 4.4.

- (e) All determinations and calculations made by us in connection with the Sale Request shall be based on any prevailing applicable rates (including spot gold prices) and such determinations and calculations will be made by us in good faith and in a commercially reasonable manner. In the absence of any manifest error or fraud on our part, all determinations and calculations will be conclusive and binding on you.
- (f) Notwithstanding the above, we reserve the right to reject any Sale Request we receive from you in our absolute discretion, without providing any reason for such rejection. In such an event, we shall notify you as soon as practicable of such rejection.
- (g) Upon the sale of Gold Bullion pursuant to a Sale Request, we shall provide you written confirmation of the same (each, a "Sale Confirmation").

4.4 Cash settlement

- (a) All payments to be made by either you or us under any Gold Trading Transaction shall be made in US Dollars, in freely transferable funds.
- (b) The total amount payable to you or us (as the case may be) shall be as set out in the Purchase Confirmation or Sale Confirmation (as the case may be).
- (c) You shall pay the total amount payable to us within one (1) Business Day following the Gold Bullion Trade Date.
- (d) We shall pay the total amount payable to you within two (2) Business Days following the Gold Bullion Trade Date, or such other date as we may determine in our absolute discretion.
- (e) All payments you make to us shall be made free of any deduction or withholding in respect of duties, fees and taxes imposed by Applicable Laws and any right of set off.

4.5 Title, Delivery and Risk

- (a) Title in the Gold Bullion you purchase or sell pursuant to any respective Purchase Confirmation or Sale Confirmation shall pass to the relevant purchaser on and from the later of:
 - (i) receipt of payment by the seller for the relevant Gold Bullion, as set out in the Purchase Confirmation or Sale Confirmation (as the case may be); and
 - (ii) deemed delivery of the relevant Gold Bullion to the purchaser, pursuant to Clause 4.5(c) or 4.5(d) (as the case may be) of these Gold Bullion Terms.
- (b) In the event that you fail to pay to us any outstanding payment due and owing to us, irrespective of title to the Gold Bullion passing to you, we shall have the right to hold the Gold Bullion and not to deliver the Gold Bullion to you until we receive full receipt of the same from you.
- (c) The relevant Gold Bullion shall be deemed delivered to you when we receive them for storage on your behalf on an Allocated Basis.
- (d) The relevant Gold Bullion shall be deemed delivered to us (or such other party that we may nominate) by you when they are removed from storage pursuant to a Sale Confirmation and no longer held on your behalf on an Allocated Basis.
- (e) Delivery of the Gold Bullion to you pursuant to a Purchase Confirmation shall be evidenced by the issue of a written confirmation acknowledging receipt of the Gold Bullion for storage in accordance with Clause 3.
- (f) The estimated delivery date of the Gold Bullion to you will be as stated in the Purchase Confirmation. Such estimated delivery date shall be for reference purposes only and provided by us in good faith. You understand and accept that the actual delivery date of the Gold Bullion may be different from the estimated delivery date because of practical issues. If the actual delivery date of the Gold Bullion is later than the estimated delivery date, this shall not be considered a breach of our contractual obligations under the relevant Purchase Confirmation.

- (g) All risks in and for the Gold Bullion shall pass to the purchaser upon the passing of title to the Gold Bullion pursuant to the provisions of Clause 4.5(a).
- 4.6 Condition Precedent to Payment and Delivery
 - (a) Our payment or delivery obligations under Clauses 4.4 and 4.5 are subject to:
 - (i) the condition precedent that no Event of Default or potential Event of Default has occurred; and
 - (ii) any other condition as may be specified in this Agreement to be a condition precedent for the purposes of this Clause 4.6.
 - (b) For the avoidance of doubt, we may in our absolute discretion and upon such terms as we think fit, waive the fulfillment of any condition precedent and such condition precedent so waived shall be deemed to have been satisfied.

4.7 Additional Events of Default

- (a) In addition to Clause 25 of the Account Terms, each of the following shall also constitute an Event of Default for the purposes of any Gold Trading Transaction:
 - (i) in the case of a Purchase Confirmation, any default under a Corresponding Purchase Transaction and, after giving any applicable notice, requirement or grace period, such default results in a liquidation of, an acceleration of obligations under, or an early termination of, that Corresponding Purchase Transaction;
 - (ii) in the case of a Sale Confirmation, any default under a Corresponding Sale Transaction and, after giving any applicable notice, requirement or grace period, such default results in a liquidation of, an acceleration of obligations under, or an early termination of, that Corresponding Sale Transaction; or
 - (iii) the termination of the Gold Custody Services.
- (b) At any time after the occurrence of an Event of Default, we may, without notice to you, declare all or any outstanding Gold Trading Transactions, and the obligations of the parties under such Gold Trading Transactions, to be terminated as of such date as we may determine in our absolute discretion (the "Early Termination Date"), and regardless of whether the relevant Event of Default is continuing on the relevant Early Termination Date, such Gold Trading Transactions (being "Terminated Transactions) and obligations shall be terminated as of such Early Termination Date. We may send you a notice (the "Termination Notice") to notify you of the Gold Trading Transactions that have been or will be terminated, the relevant Early Termination Date and the relevant Event(s) of Default. Notwithstanding our failure to send you a Termination Notice, the termination of the relevant Gold Trading Transactions shall be effective.
- (c) All outstanding monies which would otherwise be payable had such Terminated Transactions not been terminated shall forthwith be promptly returned to you or us (as the case may be) and you shall have no claim or right of action against us in any respect whatsoever and we shall cease to have any obligation to you hereunder.
- (d) For the avoidance of doubt, we shall not be liable to you for any Losses arising from the termination of any Gold Trading Transaction pursuant to this Clause 4.7.
- (e) The provisions set out in this Clause 4.7 are in addition to, and shall not prejudice any other rights we may have under this Agreement and/or any Applicable Agreement.

4.8 Indemnity

You undertake to indemnify us against any Losses which we may sustain or incur as a result of the entry into a Corresponding Purchase Transaction and/or a Corresponding Sale Transaction (for which we are not reimbursed by you through the terms of a Purchase Confirmation and/or Sale Confirmation) including:

- (a) any fee, expense and charge (including late delivery charge, default interest, insurance and transportation cost) we incur pursuant to or in connection with the Corresponding Purchase Transaction and/or the Corresponding Sale Transaction; or
- (b) all taxes we incur (apart from taxes on our overall income) arising from our entry into and performance of the Corresponding Purchase Transaction and/or the Corresponding Sale Transaction.

4.9 Exclusion and Limitation of Liability

- (a) You acknowledge that our timely supply of the Gold Bullion depends on availability and other practicalities. You further acknowledge that we shall not be liable to you under any circumstances whatsoever for any Losses directly or indirectly caused by any delay in delivery or any unavailability of Gold Bullion, whether as a result of any third party (including any counterparty pursuant to the Corresponding Purchase Transaction) or otherwise, unless such delay in delivery or unavailability of Gold Bullion arises directly out of our gross negligence, fraud or wilful default.
- (b) In the event of settlement of any Losses you suffer, we (including our insurers) shall become subrogated, to the extent of such payment, to all of your rights of recovery against any responsible party respecting such Losses. You undertake to execute and deliver all documents required by us (including our insurers), and to cooperate with and assist our insurers fully in connection with the exercise of such rights, including the maintenance of any legal proceeding in your name. You further undertake that neither you nor your agents shall do anything to prejudice our or our insurers' rights.
- (c) Without prejudice to any provision of these Gold Bullion Terms, we will not be liable to you in respect of any Losses incurred by reason of, arising from or resulting from any loss of or damage to the Gold Bullion once they are delivered to you in accordance with Clause 4.5.

4.10 No implied warranty

- (a) Without prejudice to any provision of these Gold Bullion Terms and for the avoidance of doubt, you hereby agree and acknowledge that:
 - (i) the operation of all implied warranties as to the condition of the Gold Bullion under the Sale of Goods Act; and
 - (ii) the operation of Sale of Goods (United Nations Convention) Act, are excluded from this Clause 4.
- (b) You acknowledge that no warranty shall be herein implied whether that relates to merchantability, fitness for a particular purpose or any other matters.

5. Gold Assaying Services

5.1 General

- (a) This Clause 5 shall apply if, at your request, we agree to provide the Gold Assaying Services to you.
- (b) Upon your request for the provision of the Gold Assaying Services, we shall obtain the services of an Agent that we may select from time to time to provide the Gold Assaying Services to you (each, a "Gold Bullion Service Provider"). For the avoidance of doubt, we do not directly provide the Gold Assaying Services, and shall enter into a service agreement (the "Assaying Agreement") for and on your behalf and on an undisclosed basis with the Gold Bullion Service Provider.
- (c) The Assaying Agreement comprises several documents including a set of master terms and service orders. Upon reaching an agreement to provide the Gold Assaying Services, you will execute a service order (as we may specify) which will contain the details of the items to be assayed and other provisions which are specific to the Gold Assaying Services that are to be rendered to you.
- (d) We shall exercise due skill, care and diligence in the selection, appointment and periodic review of the Gold Bullion Service Provider and the arrangements for the Gold Assaying Services, in accordance with Applicable Laws.

- (e) For so long as you request for the Gold Assaying Services, this Clause 5 shall be subject to, and you shall be bound by, the Assaying Agreement, the administrative procedures and other relevant provisions or documents relating to the provision of assaying and refining services by such appointed Gold Bullion Service Provider.
- (f) We shall not be liable for the acts, default or insolvency of any Gold Bullion Service Provider, nor for any Losses you suffer in connection with those acts, default or insolvency save for where such Losses arise directly from our gross negligence, fraud or wilful default in the initial selection of the Gold Bullion Service Provider.

5.2 Gold Bullion Service Provider

- (a) The Assaying Agreement and all other legal, fiscal, taxation, customs or other documentation (collectively, the "Assaying Documentation") as may be required for the Gold Assaying Services shall be entered into by us for and on your behalf on an undisclosed principal basis with the Gold Bullion Service Provider.
- (b) You hereby agree to indemnify us against any Losses that we may suffer in connection with our execution of the Assaying Agreement or the Assaying Documentation with the Gold Bullion Service Provider or any other party, for and on your behalf. We shall not be liable in any way for Losses that you may suffer from or in connection with the Assaying Agreement or Assaying Documentation that we may execute for and on your behalf.
- (c) In addition and without prejudice to any provision of this Agreement, you hereby agree to any disclosure of Customer Data in connection with any dispute with the Gold Bullion Service Provider, or in accordance with these Gold Bullion Terms.

5.3 Our authority

- (a) You irrevocably appoint us to be your attorney and on your behalf and in your name or otherwise to execute and do all such assurances, deeds, acts and things which you may or ought to do in your name and on your behalf to exercise all or any of the rights, powers, authorities and discretions conferred by these Gold Bullion Terms on us to seal and deliver and otherwise perfect any deed, assurance, agreement, instrument or act which we may deem proper in or for the purpose of exercising any of such rights, powers, authorities and discretions and you hereby ratify and confirm and agree to ratify and confirm whatsoever any such attorney shall do or purport to do by virtue of this Clause 5 of these Gold Bullion Terms.
- (b) You hereby authorise us, at our absolute discretion, to act on your behalf to take any action or exercise any right as may seem to be advisable or expedient for the purposes of carrying out the Gold Assaying Services.

PART I – SPECIFIC TERMS & CONDITIONS GOVERNING TRANSACTIONS ON THE CHINA STOCK CONNECT

Section I: Supplemental Terms for China Stock Connect

1. Applicability

- 1.1 These China Connect Terms shall apply where you (the "**Customer**") informs or indicates to the Bank that the Customer wishes to trade China Connect Securities through China Connect.
- 1.2 In the event of any conflict or inconsistency between:
 - (a) these China Connect Terms and any other provision of this Agreement, these China Connect Terms shall prevail in respect of any Transaction involving China Connect Securities; and/or
 - (b) these China Connect Terms and any Applicable Agreement in respect of any Transaction involving China Connect Securities, that Applicable Agreement shall prevail.

2. Definitions

Capitalised terms used in these China Connect Terms will have the meanings given below.

"A Shares" means any securities issued by companies incorporated in Mainland China which are listed and traded, from time to time, on the Mainland China A Share market (i.e. the SSE or the SZSE) and not on SEHK.

"Cash" means all cash or cash equivalents in Renminbi received and held by the Bank on the terms of these China Connect Terms.

"CCASS China Connect Rules" means the CCASS, as amended for the purposes of implementing China Connect, and as amended, supplemented, modified and/or varied from time to time.

"ChinaClear" means China Securities Depository and Clearing Corporation Limited.

"China Connect" means the Shanghai Hong Kong Stock Connect and/or the Shenzhen Hong Kong Stock Connect (as the case may be).

"China Connect Authorities" means the regulators which regulate China Connect and activities relating to China Connect, including without limitation, the CSRC, the People's Bank of China, the State Administration of Foreign Exchange, SFC and any other regulator, agency or authority with jurisdiction, authority or responsibility in respect of China Connect.

"China Connect Entities" means the exchanges, clearing systems and other entities which provide services relating to China Connect, including without limitation, Stock Exchange of Hong Kong, Hong Kong Securities Clearing Company Limited, Shanghai Stock Exchange, Shenzhen Stock Exchange and ChinaClear.

"China Connect Laws" means the laws and regulations of Hong Kong and Mainland China from time to time in respect of China Connect or any activities arising from China Connect.

"China Connect Market" means SSE and/or SZSE (as the case may be).

"China Connect Market System" means the system used for the trading of China Connect Securities on SSE and/or SZSE (as the case may be), as operated by SSE and SZSE respectively.

"China Connect Rules" means any rules, policies or guidelines published or applied by any China Connect Authority or China Connect Entity from time to time in respect of China Connect or any activities arising from China Connect.

"China Connect Securities" means any securities listed and traded on a China Connect Market which may be traded by Hong Kong and international investors under China Connect.

"China Connect Service" means the order-routing service through which Northbound orders placed by an Exchange Participant may be transmitted by a SEHK Subsidiary to the corresponding China Connect Market for the buying and selling of China Connect Securities and any related supporting services.

"ChiNext Shares" means any securities listed and traded on the ChiNext Board of the SZSE which may be traded by Hong Kong and international investors under China Connect.

"Clearing Participant" has the meaning given to such term in the rules of CCASS.

"Client Identity Rules" means the SFC's client identity rules in the SFC's Code of Conduct and Client Identity Rule Policy.

"Client Securities Rules" means the Securities and Futures (Client Securities) Rules (Cap 571H of the Laws of Hong Kong).

"CSC" means the China Stock Connect System for receiving and routing orders under China Connect to the trading system on a China Connect Market for automatic matching and execution.

"CSRC" means China Securities Regulatory Commission.

"CSRC China Connect Rules" means the Several Provisions regarding Mainland China - Hong Kong Stock Connect Mechanism, as promulgated by CSRC to prescribe the launch and operation of China Connect.

"Exchange Participant" has the meaning given by the rules of SEHK and shall include DBS Vickers (Hong Kong) Limited or other affiliate of the Bank as notified by the Bank to the Customer.

"Forced-sale Notice" has the meaning given in Clause 10.1.

"H Shares" means any securities issued by companies incorporated in Mainland China and listed on SEHK.

"HKEx" means Hong Kong Exchanges and Clearing Limited.

"HKSCC" means the Hong Kong Securities Clearing Company Limited, a wholly owned subsidiary of HKEx.

"Mainland China" means the People's Republic of China (excluding Hong Kong, Macau and Taiwan).

"Mainland China Listco" means a Mainland China incorporated company which is listed on a Mainland China stock exchange.

"Mainland China Resident" means a person who is a citizen of the People's Republic of China and who does not have permanent right of abode in a jurisdiction outside Mainland China.

"Northbound" denotes the trading of China Connect Securities by Hong Kong and international investors through China Connect.

"Pre-Trade Checking" means the requirement under the China Connect Laws pursuant to which a China Connect Market may reject a sell order if an investor does not have sufficient available China Connect Securities in its account.

"Renminbi" or "RMB" means the lawful currency of Mainland China, deliverable in Hong Kong.

"SEHK" means the Stock Exchange of Hong Kong.

"SEHK China Connect Rules" means the rules of SEHK, as amended for the purposes of implementing China Connect, and as amended, supplemented, modified and/or varied from time to time.

"SEHK Subsidiary" means a wholly-owned subsidiary of SEHK duly authorised as an automated trading service provider under the SFO and licensed under applicable laws in Mainland China to provide the order-routing service under China Connect.

- "SFC" means the Securities and Futures Commission of Hong Kong.
- "SFO" means the Securities and Futures Ordinance (Cap. 571 of the Laws of Hong Kong).
- "Shanghai Hong Kong Stock Connect" means a securities trading and clearing links programme developed or to be developed by SEHK, SSE, HKSCC and ChinaClear for the establishment of mutual market access between SEHK and SSE.
- "Shenzhen Hong Kong Stock Connect" means a securities trading and clearing links programme developed or to be developed by SEHK, SZSE, HKSCC and ChinaClear for the establishment of mutual market access between SEHK and SZSE.
- "Special China Connect Securities" means any securities listed on a China Connect Market which SEHK (after consulting with the relevant China Connect Market) accepts or designates as eligible only for China Connect sell orders and not China Connect buy orders.
- "SSE" means the Shanghai Stock Exchange.
- "SSE China Connect Rules" means the SSE regulations on the Shanghai Hong Kong Stock Connect which have been published by SSE for the purposes of implementing China Connect, as amended, supplemented, modified and/or varied from time to time.
- "SSE Listing Rules" means the Rules Governing the Listing of Stocks on SSE, as amended, supplemented, modified and/or varied from time to time.
- "SSE Rules" means the SSE China Connect Rules and the business and trading rules and regulations of SSE, as amended, supplemented, modified or varied from time to time.
- "SZSE" means the Shenzhen Stock Exchange.
- "SZSE China Connect Rules" means the SZSE regulations on the Shenzhen Hong Kong Stock Connect which have been published by SZSE for the purposes of implementing China Connect, as amended, supplemented, modified and/or varied from time to time.
- "SZSE Listing Rules" means the Rules Governing the Listing of Stocks on Shenzhen Stock Exchange (including the Rules Governing the Listing of Stocks on the ChiNext Board), as amended, supplemented, modified and/or varied from time to time.
- "SZSE Rules" means the SZSE China Connect Rules and the business and trading rules and regulations of SZSE, as amended, supplemented, modified or varied from time to time.
- "Taxes" means all taxes (including, but not limited to, all capital gains taxes), duties, levies, imposts, charges, assessments, deductions, withholdings and related liabilities, including additions to tax, penalties and interest imposed on or in respect of (i) China Connect Securities or Cash, (ii) any transaction effected under these China Connect Terms or (iii) the Customer (including, in each of cases (i), (ii) and (iii), as imposed by the relevant authorities in Hong Kong and/or Mainland China).
- "**Trading Day**" means a day on which SEHK is open for Northbound trading for (a) Hong Kong and Shanghai (in the case of Shanghai Hong Kong Stock Connect) or (b) Hong Kong and Shenzhen (in the case of Shenzhen Hong Kong Stock Connect), where "T day" denotes the Trading Day on which a transaction is executed and "T+1 day" denotes the day which is one Trading Day after T day.
- 2.1 References to Clauses are to Clauses of these China Connect Terms.
- 2.2 The headings used in these China Connect Terms are for convenience only and shall not affect the construction or interpretation hereof.
- 2.3 Any reference in these China Connect Terms to words importing the singular shall include the plural and vice-versa.
- 2.4 A document includes any variations, modifications, and/or replacement thereof and supplements thereto.

2.5 A statute and other legislations shall be construed as a reference to such statute or other legislations in force and as amended, re-enacted or replaced from time to time and any regulation promulgated or issued thereunder.

3. Eligible Investors

The Customer represents and undertakes on a continuing basis (including without limitation on the first day that these China Connect Terms are effective and on each date the Customer places an order or gives an instruction under these China Connect Terms), that:-

- 3.1 the Customer (if a natural person) is not a Mainland China Resident or (if it is a legal person) it is not an entity incorporated or registered under the laws of Mainland China, or the Customer falls within any relevant excluded categories/ exemptions as the case may be, whereby the compliance with the relevant requirements is the responsibility of the Customer; and
- 3.2 the Customer or the Customer's entry into of any transaction under these China Connect Terms does not violate the laws and regulations of Mainland China, including those in relation to foreign exchange control and reporting.

4. Trading of ChiNext Shares

- 4.1 Subject to Clause 4.2 below, the Bank will not accept any orders to buy or sell ChiNext Shares through the China Connect Service from the Customer.
- 4.2 Where, as a result of any distribution of rights (including the right to subscribe for rights issues or open offers) or entitlements, conversion, takeover, other corporate actions or special circumstances arising from China Connect Securities, the Customer receives any ChiNext Shares and such ChiNext Shares are accepted or designated by the SEHK as China Connect Securities or Special China Connect Securities, the Bank may, at its sole discretion, accept sale orders from the Customer to sell such ChiNext Shares through the China Connect Service.

5. Compliance with China Connect Laws and China Connect Rules

- 5.1 Trading in China Connect Securities will be subject to all China Connect Laws and China Connect Rules.
- 5.2 These China Connect Terms highlight certain key features of China Connect as of the date hereof. The Bank is not liable for any inaccuracies or misstatements in the information set out in these China Connect Terms. These China Connect Terms do not purport to cover all China Connect Laws and China Connect Rules. The Customer understands that the Customer shall be fully responsible for understanding and complying with all China Connect Laws and China Connect Rules and for any consequences of Northbound trading. The Bank does not, and does not intend to, advise the Customer on any China Connect Laws or China Connect Rules.
- 5.3 The Bank shall have the right to apply any procedures or requirements in respect of any trading of China Connect Securities which it determines in its absolute discretion to be necessary or desirable for the purpose of any China Connect Laws, China Connect Rules, requests/requirements of the relevant Exchanges or market practice.
 - For example, clients who are (or become) Mainland Investors are not permitted to hold a BCAN and shall be subject to restrictions or prohibitions from trading under Stock Connect, and this includes participation in voluntary corporate action. In certain situations, such clients may not be able to deal in their existing holdings (acquired prior to the change in their status) and/or may face limited or no voting rights at all. There may also be possible audit checks or inspections on Customers who are end investors or users. The Bank shall have no liability for any losses or risks which may result directly or indirectly from such procedures or requirements.
- 5.4 The Bank may, in its absolute discretion, refuse to execute any instruction provided by the Customer if such instruction is not or if the Bank reasonably believes may not be compliant with any China Connect Laws or China Connect Rules. The Bank shall have no liability for any losses or risks which may result directly or indirectly from such refusal.

5.5 Without limitation to the foregoing, the Bank may in its absolute discretion suspend, terminate or limit the Customer's ability to access China Connect through the Bank without advance notice to the Customer, including but not limited to where requested or directed by a China Connect Authority or China Connect Entity or pursuant to any arrangements with them.

6. Risk Disclosures and Acknowledgement

- 6.1 By instructing the Bank in respect of any transaction relating to China Connect Securities, the Customer acknowledges:
 - 6.1.1 that (i) it has read and understood the risk disclosures and other information set out in Part B hereof; (ii) it understands that there is a risk of prohibition of trading China Connect Securities; and (iii) it understands its obligations set out in Part B hereof including the consequences of a breach of China Connect Laws:
 - 6.1.2 the Bank shall not be liable for any loss, liability or third party claim or demand that it may suffer directly or indirectly as a result of any action or inaction by the Bank in connection with the provision of services in respect of China Connect Securities to the Customer by the Bank including, without limitation, the materialisation of any of the risks described in Part B hereof;
 - 6.1.3 that SEHK has the power not to extend the China Connect Service to the Customer if it is found that the Customer has or may have committed any abnormal trading conduct set out in the SSE Rules and/ or the SZSE Rules (as the case may be);
 - 6.1.4 that if the SSE Rules and/or the SZSE Rules (as the case may be) are breached, or the disclosure and other obligations referred to in any China Connect Laws or China Connect Rules are breached, the relevant China Connect Market has the power to carry out an investigation, and may, through SEHK (or any other governmental or regulatory body), require the Bank to (a) provide relevant information and materials relating to it including, without limitation, its identity and trading activity; and (b) to assist in a China Connect Authority's investigation in relation to it and/or its trading activity;
 - 6.1.5 that where a China Connect Authority considers that there is a serious breach of the SSE Rules and/ or the SZSE Rules (as the case may be), the Bank may be required by a China Connect Authority to (a) issue warning statements (verbally or in writing) to it; and (b) cease providing it with any service relating to trading China Connect Securities through China Connect;
 - 6.1.6 and agrees that prior to the Bank informing the Customer that a Northbound buy order made by the Customer has been settled, it shall not make a Northbound sell order in respect of the China Connect Securities which are the subject of such Northbound buy order;
 - 6.1.7 and consents to the Bank providing information relating to its profile, the type and value of Northbound buy and sell orders and transactions executed on its behalf to a China Connect Authority at such intervals and in such form as such China Connect Authority may specify from time to time, and includes and is not limited to that further described in this paragraph;
 - 6.1.8 and accepts responsibility for paying all fees, charges, levies and taxes and shall comply with any filing or registration obligations as may be required under any China Connect Laws or China Connect Rules relating to any China Connect Securities and any dividends or entitlements in respect of such China Connect Securities;
 - 6.1.9 that SEHK may, upon the request of the relevant China Connect Market, require the Bank to reject its order; and
 - 6.1.10 that none of the China Connect Authorities or their respective directors, employees or agents shall be responsible or held liable for any loss or damage that it may suffer, whether directly or indirectly in connection with the trading of China Connect Securities or the operation of the CSC in respect of China Connect Securities.
- 6.2 The Customer acknowledges and agrees that in instructing the Bank in respect of any transactions relating to China Connect Securities, the Bank will be required to:

- (a) tag each of the Customer's orders with a Broker-to-Client Assigned Number ("**BCAN**") that is unique to the Customer and/or the Customer's account;
- (b) provide to the SEHK, directly or indirectly through an Exchange Participant, the Customer's assigned BCAN and Customer Data automatically prior to or at the time of each trade, as well as whenever the SEHK may request. Under the SEHK China Connect rules, the Customer Data that will be disclosed includes the BCAN and the following:
 - (i) for individuals: name (in English and Chinese), identity document ("ID") issuing country/region, ID type and ID number; and
 - (ii) for companies/entities: entity name, place of incorporation, ID type/certificate of incorporation and ID number.

These categories of information are referred to as "Client Identification Data" or "CID" under the SEHK China Connect rules.

The Customer is aware that the Customer Data required to be disclosed is indicated by the authorities around the time the BCAN investor identification model was implemented, and that the type of Customer Data that will be disclosed is hence always subject to further changes. By trading through the Bank, the Customer's authorisation to the Bank on the disclosure of Customer Data is not limited to the aforementioned CID.

- 6.3 Without prejudice to and without limiting any notification that the Bank has given the Customer or consent that the Customer has earlier provided to the Bank in respect of the collection, use, storage and/or processing of all of the Customer's Customer Data, the Customer agrees that its consent includes and is not limited to the following:
 - 6.3.1 Disclosure to SEHK and SEHK Subsidiaries
 - (i) to disclose and transfer the Customer's CID to the SEHK and the relevant SEHK Subsidiaries from time to time at the time of order taking and/or after (whether on a real-time basis or otherwise); and
 - (ii) to allow each of SEHK and the relevant SEHK Subsidiaries to: (i) collect, use, process and store the Customer's CID and any consolidated, validated and mapped BCANs and CID whether for market surveillance and monitoring purposes, enforcement of SEHK China Connect Rules or otherwise; (ii) transfer such information to the relevant China Connect Market from time to time for the purposes set out in this entire section; and (iii) disclose such information to the relevant regulators and law enforcement agencies in Hong Kong so as to facilitate the performance of their statutory functions with respect to the Hong Kong financial markets;

6.3.2 Disclosure to Mainland China entities

- (i) to allow ChinaClear to: (i) collect, use, process and store the Customer's BCAN and CID to facilitate the consolidation and validation of BCANs and CID and the mapping of BCANs and CID with its investor identification database, and provide such consolidated, validated and mapped BCANs and CID information to the relevant China Connect Market, SEHK and the relevant SEHK Subsidiary; (ii) use the Customer's BCAN and CID for the performance of its regulatory functions including and not limited to securities account management; and (iii) disclose such information to the Mainland China regulatory authorities and Mainland China law enforcement agencies having jurisdiction over it so as to facilitate the performance of their regulatory, surveillance, enforcement and other functions with respect to the Mainland China financial markets; and
- (ii) to allow the relevant China Connect Market, China Connect Authorities and China Connect Entities to: (i) collect, use, process and store the Customer's BCAN and CID to facilitate their surveillance and monitoring of securities trading on the relevant China Connect Market through the use of the China Connect Service, enforcement of and compliance with the rules of the relevant China Connect Market; and (ii) disclose such information to the Mainland China regulatory authorities and Mainland China law enforcement agencies so as to facilitate the performance of their regulatory, surveillance, enforcement and other functions with respect to the Mainland China financial markets.

6.3.3 Disclosure to all intermediaries

In order to provide the China Connect Service, there may be various intermediaries involved to submit, execute and/or process the Customer's trade and other related instructions. These intermediaries may be external parties, may not work directly with the Bank, and may not be known to the Customer. By instructing or trading in China Connect securities through the Bank, the Customer's consent to the disclosure, use, storage and processing of all Customer Data shall likewise be extended to all intermediaries involved.

6.4 By instructing the Bank in respect of any transaction relating to China Connect Securities, the Customer agrees that the Bank may use the Customer's Customer Data for the purposes of complying with the requirements of SEHK, the Mainland and Hong Kong authorities and regulatory/ enforcement agencies, and their respective rules in force from time to time relating to China Connect securities.

7. Representations and Undertakings

- 7.1 The Customer makes the representations set out in this Clause 7.1 to the Bank on a continuing basis (including without limitation on the first day that these China Connect Terms are effective and on each date that it places an order or gives an instruction in respect of China Connect Securities) and such representations shall apply to each transaction conducted under the China Connect Terms:
 - 7.1.1 it is aware of and shall comply with all China Connect Laws and China Connect Rules to which it may be subject;
 - 7.1.2 the execution of any instruction it gives to the Bank shall not result in any breach of any China Connect Laws or China Connect Rules;
 - 7.1.3 it understands and has assessed the risks relating to China Connect and is willing to undertake such risks;
 - 7.1.4 it is not an insider as defined or interpreted under Mainland Chinese law and it is not in possession of inside information when trading China Connect Securities or procuring others to do so;
 - 7.1.5 it does not hold more than 5% of the shares of any Mainland China-incorporated company which is listed and traded on any stock exchange in Mainland China; and
 - 7.1.6 in purchasing China Connect Securities, it has no intention to manipulate the market.
- 7.2 The Customer makes the following representations to the Bank on each date it places an order to sell China Connect Securities that:
 - 7.2.1 it does not know of any fact that might impair the validity of such China Connect Securities and it has full authority to receive, deal with and give instructions, authorisations or declarations in respect of the same;
 - 7.2.2 there is no adverse claim to such China Connect Securities; and
 - 7.2.3 there is no restriction on the transfer of such China Connect Securities other than those expressly provided for under the SEHK China Connect Rules or CCASS China Connect Rules.
- 7.3 The Customer undertakes to the Bank to inform it immediately in writing (and in any event no later than one business day after the relevant event) of anything that may potentially render any of the representations in these China Connect Terms inaccurate or misleading.

8. Compliance with Pre-Trade Checking Requirements

- 8.1 The Customer undertakes to:
 - 8.1.1 comply with any requirements relating to Pre-Trade Checking mandated by the China Connect Authorities, the China Connect Entities or as notified to the Customer by the Bank; and

- 8.1.2 ensure there are sufficient available China Connect Securities in its account by the applicable cut-off time on the applicable day (as notified to it (orally or through email or other means of communication as agreed between the Bank and the Customer) by the Bank from time to time) to cover any proposed sell order given on that Trading Day.
- 8.2 The Customer understands that if the Bank considers that it does not for whatever reason have sufficient available China Connect Securities in its account to settle a sell order by the applicable cut-off time (as notified to the Customer by the Bank from time to time), the Bank may in its absolute discretion:
 - 8.2.1 reject its sell order; or
 - 8.2.2 perform any other act which the Bank considers necessary or desirable to comply with Pre-Trade Checking and/or relevant China Connect Laws or China Connect Rules and to cover the shortfall (including but not limited to applying any other China Connect Securities available to the Bank from other sources).
- 8.3 In respect of any instruction to make a Northbound buy order, if the Bank determines in its absolute discretion that the Customer does not have sufficient funds to settle the payment obligation in respect of such order on the settlement day, the Bank may, in its absolute discretion, reject the buy order.
- 8.4 Any risk, loss or cost resulting from non-compliance or potential non-compliance with Pre Trade Checking and/ or the relevant China Connect Laws or China Connect Rules shall be borne by the Customer. The Customer shall reimburse the Bank for any costs, losses or expenses which the Bank incurs as a result of its failure to deliver in respect of its sell order on such terms and price (including any associated fees and expenses) and at such time as the Bank shall determine in its absolute discretion.

9. Settlement and Currency Conversion

- 9.1 As all Northbound trading is effected and settled in Renminbi, if the Bank does not receive sufficient Renminbi before settlement of a Northbound buy order to settle such purchase of China Connect Securities, settlement may be delayed and/or fail and the Customer may not acquire title to sell or transfer the relevant China Connect Securities. Where the Bank holds any funds on the Customer's behalf, if there are insufficient Renminbi funds to settle any Northbound buy order or other payment obligation in connection with China Connect, the Customer authorises the Bank to convert any funds in another currency which the Bank holds on its behalf into Renminbi at the prevailing exchange rate of the Bank for the purposes of settlement thereof.
- 9.2 Notwithstanding any other provision in this Agreement, where it is necessary to convert one currency to another pursuant to these China Connect Terms and if the Bank chooses to do so at its absolute discretion, such conversion may be carried out automatically by the Bank in a commercially reasonable manner without prior notice to the Customer. Any risk, loss or cost resulting from any conversion of one currency into another currency pursuant to these China Connect Terms shall be borne by the Customer. For avoidance of doubt, the Customer understands and acknowledges that the Bank does not have any obligation to make any such conversion.
- 9.3 The Customer agrees that if it fails to settle any payment obligation in relation to an instruction to purchase China Connect Securities in a timely manner, the Bank has the right to immediately and without prior notice to the Customer take such action as the Bank considers appropriate to reduce or eliminate any loss or liability that the Bank suffers or may suffer (including but not limited to taking any steps to sell, realize, dispose of or otherwise deal with the relevant China Connect Securities) and the Customer shall indemnify and hold the Bank harmless for any liabilities, expenses or other losses the Bank may incur in exercising the foregoing right. The Customer further agrees that the Bank shall have no liability to it for any loss, diminution in value or other damages whatsoever for any action or inaction of the Bank or its agents pursuant to this Clause.
- 9.4 Notwithstanding any other provision in this Agreement, where the Bank determines that there is insufficient liquidity in RMB to settle any buy orders, the Bank may, in its sole and absolute discretion, reject such buy order.

10. Sale, Transfer and Disgorgement

- 10.1 Where, under the terms of the China Connect Rules, the Bank receives notice (a "Forced-sale Notice") from a China Connect Authority requiring the Bank to sell and liquidate a specified number of China Connect Securities, the Bank shall be entitled to issue a corresponding notice (a "Client Forced-sale Notice") to the Customer requesting the Customer to sell and liquidate any number of such China Connect Securities that the Customer holds in its account with the Bank (as determined by the Bank in its sole discretion) within the time period specified by the relevant China Connect Authority, and the Customer undertakes to comply with any such Client Forced-sale Notice.
- 10.2 In relation to any Forced-sale Notice, the Customer authorises the Bank to sell or arrange for the sale of such China Connect Securities on its behalf at such price and on such terms as the Bank may determine in its absolute discretion if it fails to comply with a Client Forced-sale Notice in a timely manner, to the extent necessary to comply with all China Connect Laws and China Connect Rules.
- 10.3 Where China Connect Securities owned by the Customer that are the subject of a Client Forced-sale Notice have been transferred from the Clearing Participant that settled the relevant Northbound buy order (the "Original CP") to another Clearing Participant or custodian (the "Recipient Agent"), the Customer authorises the Bank to provide instructions to the Recipient Agent on its behalf to return the relevant China Connect Securities to the Original CP for sale and liquidation in accordance with all China Connect Laws and China Connect Rules. The Customer also undertakes to inform the Recipient Agent of such authorisation and, where required, the Customer undertakes to instruct the Recipient Agent to act accordingly.
- 10.4 The Customer authorises the Bank to sell or arrange for the sale of any quantity of China Connect Securities owned by the Customer if the Bank receives notice from any China Connect Authority requiring it to disgorge any profits as a result of the "short swing profit rule", as more fully described under paragraph 21 (Short Swing Profit Rule) of Part B of these China Connect Terms.
- 10.5 In addition to the above, the Customer authorises the Bank to sell, transfer or carry out any other action in relation to China Connect Securities owned by the Customer if the Bank is instructed to do so by any China Connect Authority or if the Bank otherwise determines in its absolute discretion that it is necessary or desirable to do so in order to comply with any China Connect Laws or China Connect Rules.
- 10.6 The Bank shall have no liability for any losses or risks which may result to the Customer directly or indirectly from any actions taken by the Bank under this Clause.

11. Fees and Taxation

- 11.1 The Bank shall be entitled in its absolute discretion, without further notice or demand, forthwith to satisfy any obligation of the Bank or the Customer to pay or account for any amounts in respect of any Taxes by selling, realising or otherwise dealing with, in such manner as the Bank in its absolute discretion may determine, all or part of any of the Customer's property held by the Bank for any purpose in any of its accounts, and to apply the proceeds in reduction of all or part of its liability to any tax authority or the Bank.
- 11.2 The Customer is responsible for paying all fees, charges, levies and taxes, and the Customer undertakes that it will comply with any filing or registration obligations, in each case as may be required under any China Connect Laws or China Connect Rules relating to any China Connect Securities and any dividends or entitlements in respect of such China Connect Securities.
- 11.3 Notwithstanding any other provision in these China Connect Terms, the Bank shall not be responsible for and shall have no liability to the Customer for any damage, liability or loss (including loss of profit) which may result directly or indirectly from any actions taken by the Bank under this Clause unless such damage, liability or loss is a direct result of the Bank's fraud, wilful default or negligence.

12. Indemnity

In addition and without prejudice to any of the Bank's rights under this Agreement, including without limitation Clause 9.3 above, the Customer will indemnify the Bank on a full indemnity basis against any claims, demands, actions, proceedings, damages, costs, expenses, losses and all other liabilities whatsoever arising directly or indirectly from the Bank providing any services to the Customer in respect of its trading or investment in China Connect Securities, including, without limitation, to (a) any Taxes resulting from any trading of China Connect

Securities pursuant to China Connect; (b) the materialisation of any risk referred to in Part B hereof; (c) any legal costs which the Bank may incur in connection with any instruction given by the Customer; or (d) any costs incurred in connection with Clause 10 (Sale, Transfer and Disgorgement) above, in each case other than those claims, demands, actions, proceedings, damages, costs, expenses, losses and liabilities which are a direct result of the Bank's fraud, wilful default or negligence.

13. Payments in respect of China Connect Securities

All sums payable by the Customer under this Agreement shall be paid or accounted for free and clear of any tax (including goods and services and value added tax), levy, duty, charge, impost, fee, interest, penalty, deduction or withholding (collectively the "taxes") of any nature now or hereafter imposed, withheld or assessed by any taxing or other authority. If the Customer is required by any law or regulation to make any deduction or withholding on account of any taxes from any such sums payable by it, the Customer shall, together with such sums, immediately pay or account for such additional amount as will ensure that the Bank receives or is credited (free and clear of the taxes) the full amount which the Bank would have received or been credited if no such deduction or withholding had been required. Upon the Bank's request, the Customer will promptly forward to the Bank copies of official receipts or other evidence showing that the full amount of any such deduction or withholding has been paid over to the relevant taxing or other authority.

14. Miscellaneous

- 14.1 The Customer will provide to the Bank all information (including translations into Chinese, if required) which the Bank requests if such information is requested by an exchange, regulatory authority or an organisation (whether within or outside Hong Kong) with which HKEx or SEHK has entered into an information sharing arrangement or agreement. The Customer acknowledges that, the Customer's failure to comply with this provision may, among other things, result in the suspension of China Connect Service to it.
- 14.2 The Customer will execute any further documents and provide any materials and/or information as the Bank may reasonably request to enable the Bank to perform its duties and obligations under these China Connect Terms which may become necessary as and when the China Connect Rules are amended from time to time.
- 14.3 The Bank reserves the right at its discretion to review, add to, alter, vary or modify all or any of these China Connect Terms by giving written notice to the Customer in accordance with the Account Terms.

15. Governing Law and Jurisdiction

15.1 The parties agree that the governing law and jurisdiction clauses in the Account Terms shall also apply to these China Connect Terms.

Section II: Risk Disclosures and Other Information

This Part describes some of the key risk factors and other information concerning China Connect. This Part does not disclose all the risks and other significant aspects of Northbound trading through China Connect. You acknowledge that you understand the nature and risks of China Connect and Northbound trading and you have considered carefully (and have consulted your own advisers where necessary) whether trading in China Connect Securities is suitable for you in light of your circumstances. The decision to trade in China Connect Securities is yours, and you have fully understood and are willing to assume the risks associated with China Connect and are able to comply with all relevant China Connect Laws and China Connect Rules. You acknowledge the risks and agree to the terms set out in these China Connect Terms.

You understand the Bank does not represent that the information set out in this Part is up to date or comprehensive and does not undertake to update the information set out in this Part. You are responsible for monitoring changes in the China Connect Laws and China Connect Rules and complying with any new requirements.

THIS IS NOT AN EXHAUSTIVE LIST OF ALL THE RISKS AND OTHER SIGNIFICANT ASPECTS OF ANY TRANSACTION. THE BANK RECOMMENDS THAT YOU OBTAIN ALL RELATED TERMS AND CONDITIONS AND CAREFULLY STUDY AND EVALUATE THE SAME AND CONSULT YOUR OWN LEGAL, TAX AND FINANCIAL ADVISORS OR OTHER PROFESSIONAL ADVISORS AS APPROPRIATE

Home Market Rules

1. Home Market Rules

As Mainland China is the home market of China Connect Securities, the SSE Rules and/or SZSE Rules (as the case may be) and other Mainland China securities laws and regulations shall apply to you. If such rules and regulations are breached, the relevant China Connect Market has the power to carry out an investigation.

Nevertheless, certain Hong Kong legal and regulatory requirements will also continue to apply to Northbound trading.

Trading and Settlement Restrictions

2. Pre-Trade Checking

SEHK is required to check that, in respect of any Northbound sell orders given by an Exchange Participant, the relevant Exchange Participant holds sufficient available China Connect Securities to be able to fill such Northbound sell orders by the commencement of trading on the Trading Day you wish to execute. Otherwise, you may be unable to execute Northbound sell orders due to Pre-Trade Checking related requirements.

You shall bear any risk, loss, cost or expenses resulting from non compliance or potential non-compliance with Pre-Trade Checking and/or the relevant China Connect Laws or China Connect Rules and/or the above actions taken by the Bank.

3. Settlement

Northbound trades will follow the A Shares settlement cycle. For settlement of China Connect Securities trades, ChinaClear will debit or credit the securities accounts of its participants (including HKSCC as clearing participant) on T day free of payment and settlement of funds relating to such trading will be effected on T+1 day. Title to China Connect Securities will not be released until the receipt of confirmation of payment. Accordingly, for purposes of contract notes, the settlement date would be T+1 day when both the securities and the cash are settled. You understand that, until the Bank has enhanced its trading system, stock and money settlements of China Connect Securities may only be reflected in your account on T+1 (as opposed to T for stock and T+1 for money settlements respectively). This system issue, however, will not affect the time at which title to China Connect Securities will be transferred to investors on T+1.

4. Quota Restrictions

Purchases of China Connect Securities are subject to daily quota controls. As a result, there is no assurance that a buy order can be successfully placed through China Connect. Any instruction to buy that has been submitted but not yet executed may be restricted or rejected.

SEHK and the relevant China Connect Market may also set pricing and other restrictions on buy orders in order to prevent the artificial use or filling of the applicable quotas.

5. Restriction on Day Trading

Day (turnaround) trading is not permitted on the Mainland China A Shares market.

6. No off-exchange trading and transfers

Unless otherwise provided by the CSRC, you understand that the Bank shall not, with certain limited exceptions, trade or provide services to facilitate trading of any China Connect Securities through any venue other than through the China Connect Market System, and the Bank shall not match, execute or arrange the execution of any sale and purchase instructions or any transfer instructions from you in respect of any China Connect Securities in any manner otherwise than through China Connect in accordance with the China Connect Rules.

7. Placing Orders

Only limit orders with a specified price are allowed pursuant to China Connect Laws and China Connect Rules, where buy orders must not be lower than the current best price and sell orders may be executed at or higher than the specified price. Market orders will not be accepted.

8. China Connect Market Price Limits

China Connect Securities are subject to a general price limit of a range of percentages based on the previous trading day's closing price. The price limit may be changed from time to time. All orders in respect of China Connect Securities must be within the price limit and any orders with a price beyond the price limit will be rejected by the relevant China Connect Market.

9. Restrictions on selling China Connect Securities

You are prohibited from using China Connect Securities purchased through China Connect to settle any sell orders placed through channels other than China Connect. Accordingly, there may be a limited market and/ or lower liquidity for China Connect Securities purchased through China Connect (as compared to the same shares purchased through other channels). In addition, any scrip entitlements received by you in respect of China Connect Securities may not be eligible for trading through China Connect. Accordingly, there is a risk of low/no liquidity for such shares received by way of scrip entitlement.

10. Delisting of companies listed on a China Connect Market

According to the SSE Listing Rules and the SZSE Listing Rules, if any SSE-listed company or SZSE-listed company (as the case may be) is in the delisting process, or its operation is unstable due to financial or other reasons such that there is a risk of being delisted or exposing your interest to undue damage, the SSE-listed company or SZSE-listed company (as the case may be) will be earmarked and traded on the "risk alert board". Any change to the risk alert board may occur without prior notice. You understand that you will only be allowed to sell such shares and are restricted from further buying.

11. Account Information of Beneficial Owner

You understand that your identity as the beneficial owner of China Connect Securities which are the subject of a sell order may need to be disclosed to HKSCC and/or relevant Mainland China authorities.

12. Scripless Securities

China Connect Securities are traded in scripless form and accordingly, China Connect Securities may not be physically deposited into and/or withdrawn from CCASS.

13. Short Selling

Covered short selling of China Connect Securities may be available provided such covered short selling satisfies the requirements specified by the relevant China Connect Authorities. However naked short selling of China Connect Securities is prohibited. You are fully responsible for understanding and complying with short selling requirements as amended from time to time and for any consequences of non-compliance.

14. Amendment of Orders and Loss of Priority

Consistent with the current practice in Mainland China, if you wish to amend an order, you must first cancel the original order and then input a new one. Accordingly, order priority will be lost and, subject to the daily quota restrictions, the subsequent order may not be filled on the same Trading Day.

15. Warning Statements

You understand that SEHK may require the Bank

- (a) to issue warning statements to you; and/or
- (b) to cease to extend the China Connect Service to you.

16. Special China Connect Securities

You understand that SEHK will accept or designate securities which cease to meet the eligibility criteria for China Connect Securities as Special China Connect Securities (provided that they remain listed on the relevant China Connect Market). In addition, any securities or options (which are not already accepted as China Connect Securities) received by you as a result of any distribution of rights or entitlements, conversion, takeover, other corporate actions or abnormal trading activities will be accepted or designated by SEHK as Special China Connect Securities. You understand that you will only be able to sell, but not buy, any Special China Connect Securities.

17. Margin Trading

Subject to certain conditions prescribed by the China Connect Authorities, you may only conduct margin trading in certain China Connect Securities as determined by the relevant China Connect Authorities to be eligible for margin trading from time to time. A China Connect Market may suspend margin trading activities in any specific China Connect Security where the volume of margin trading activities in such China Connect Security exceeds certain thresholds and may resume margin trading activities when the volume drops below a prescribed threshold. The Bank shall not have any obligation to update you in respect of the list of eligible margin trading securities or any restrictions or suspensions in respect of margin trading from time to time.

18. Rights Issuances

Where you receive as entitlements shares or other types of securities from the issuer of a China Connect Security, whether you can buy and/or sell the entitlement security through China Connect will depend on a number of factors including whether the entitlement security is a China Connect Security, whether or not it is traded in RMB and whether or not HKSCC has put in place arrangements enabling you to do so.

19. Odd Lot Trading

Odd lot trading in China Connect Securities is available only for sell orders and all odd lots must be sold in one single order. A board lot order may be matched with different odd lot sell orders, resulting in odd lot trades. Board lot and odd lot orders are matched on the same platform on China Connect and subject to the same share price. The maximum order size is 1 million shares and the tick size is uniformly set at RMB0.01.

20. Disclosure of Interests

If you hold or control shares on an aggregate basis in a Mainland China Listco above a certain threshold, you may have to disclose such interest within the period specified by the relevant China Connect Authority, and you must not buy or sell any such shares within a specified period. You may also have to disclose any substantial change in your holdings.

At the same time, you may have to comply with disclosure obligations under Part XV of the SFO.

21. Short Swing Profit Rule

Under Mainland China laws, rules and regulations, the "short swing profit rule" requires you to return any profits made from purchases and sales in respect of China Connect Securities of a Mainland China Listco if (a) your shareholding in the Mainland China Listco exceeds the threshold prescribed by the relevant China Connect Authority from time to time; and (b) the corresponding sale transaction occurs within the six months after a purchase transaction, or vice versa. You will comply with the "short swing profit rule".

22. Foreign Ownership Limits

Mainland China foreign ownership limits on the number of shares a foreign investor is permitted to hold in a single Mainland China Listco and on the maximum combined holdings of all foreign investors in a single Mainland China Listco may have an adverse effect on the liquidity and performance of an investment in China Connect Securities. As a result, you may suffer losses in China Connect Securities.

You understand that if the Bank becomes aware that you have breached (or reasonably believe that you may breach upon execution of further Northbound buy orders) any foreign ownership limits, or if the Bank is so required by any China Connect Authority, including without limitation to any Forced-sale Notice issued by the relevant China Connect Market, the Bank will sell any China Connect Securities pursuant to Clause 10 (Sale, Transfer and Disgorgement) in Section I of these China Connect Terms if you fail to comply with the corresponding Client Forced-sale Notice in order to ensure compliance with all China Connect Laws and China Connect Rules. In such case, no China Connect Securities buy orders for the relevant China Connect Securities will be accepted until the relevant China Connect Market informs the corresponding SEHK Subsidiary or SEHK that the aggregate foreign shareholding has fallen below a certain percentage. SEHK may determine in its absolute discretion which Exchange Participants and what quantity of China Connect Securities should be subject to a Forced-sale Notice (this will generally be on a "last-in, first-out" basis), and SEHK's (or the relevant SEHK Subsidiary's) own records shall be final and conclusive.

Moreover, under Mainland China laws, where foreign investors hold in aggregate more than a specified percentage (the "Cautionary Level") of the issued shares of a single Mainland China Listco, upon notification by the relevant China Connect Market to the corresponding SEHK Subsidiary, SEHK and the relevant SEHK Subsidiary are required as soon as practicable thereafter to suspend accepting China Connect Securities buy orders in respect of the relevant China Connect Securities. In such circumstances, the Bank may reject your buy orders until the aggregate shareholding of foreign investors is reduced to below the specified percentage (the "Permitted Level") as advised by the relevant China Connect Market.

23. Taxation

You will be fully responsible for any Taxes in respect of China Connect Securities including, without limitation, any Mainland China taxes, and will indemnify the Bank from and against all Taxes which the Bank may incur arising in connection with any China Connect Securities which you hold, trade or otherwise deal in.

You understand and agree that the Bank assumes no responsibility for advising on or handling any tax issues, liabilities and/or obligations in connection with China Connect, nor will the Bank provide any service or assistance in this regard. Prior to investing in China Connect Securities, you understand that you should consult your own tax advisers and counsel with respect to the possible tax consequences to you of such investment since such tax consequences may differ in respect of different investors.

24. Insider Dealing, Market Manipulation and Other Market Conduct Rules

Northbound trading through the China Connect will be subject to Mainland China laws and regulations prohibiting activities that constitute market manipulation, insider dealing and related offences. The scope of these restrictions may not be the same as equivalent requirements under Hong Kong law. In particular, defences applicable under Hong Kong market misconduct rules may not be applicable under Mainland China laws and regulations. You understand that you should seek specialist advice before engaging in trading through China Connect.

25. Client Securities Rules and Client Identity Rules

You understand that as the China Connect Securities traded through China Connect are not listed or traded on SEHK, you will not have protection either under the Client Securities Rules nor under the Client Identity Rules.

26. Investor Compensation Fund

You understand that, in trading China Connect Securities, you will not enjoy the protection afforded by any investor compensation fund, whether established under the SFO or otherwise.

27. Ownership of China Connect Securities

You acknowledge that (i) you should conduct your own review of the materials published by HKEx on China Connect in relation to the ownership of China Connect Securities and the applicable China Connect Rules as they may be amended and supplemented from time to time; and (ii) you should also consult your own legal advisers to make your own assessment of your rights as a Northbound investor in China Connect Securities.

Under the rules of CCASS, HKSCC is prepared to provide assistance to the beneficial owners of China Connect Securities where necessary. HKEx notes that any beneficial owner who decides to take legal action is responsible for seeking its own independent legal advice to satisfy itself and HKSCC that a cause of action exists and the beneficial owner should be prepared to conduct the action and take up all costs in relation to the action, including providing HKSCC with indemnities and legal representation in proceedings. Further details are set out in the HKEx published materials.

28. Stock lending and borrowing

Stock borrowing and lending of China Connect Securities will be permitted only in certain circumstances. Restrictions on the duration of stock loans and record keeping requirements may apply.

Clearing house Risk

29. Risk of ChinaClear Default

If ChinaClear defaults, HKSCC may, but has no obligation to, take any legal action or court proceedings to seek recovery of outstanding China Connect Securities and monies from ChinaClear. If it does so, HKSCC will distribute the China Connect Securities and/or monies recovered to clearing participants on a pro-rata basis as prescribed by the relevant China Connect Authorities. The Bank in turn will only be distributing the China Connect Securities and/or monies to the extent recovered directly or indirectly from HKSCC. Although the likelihood of a default by ChinaClear is considered to be remote, you understand that you should be aware of this arrangement and of this potential exposure before engaging in Northbound trading.

30. Risk of HKSCC Default

The Bank's provision of services pursuant to these China Connect Terms also depends upon the performance by HKSCC of its obligations. Any action or inaction of the HKSCC or a failure or delay by the HKSCC in the performance of its obligations may result in a failure of settlement of China Connect Securities and/or monies in connection with it and you may suffer losses as a result. The Bank shall have no responsibility or liability for any such losses.

31. Company Announcements on Corporate Actions

You understand that you should refer to the website of the relevant China Connect Market and the relevant newspapers for the latest listed company announcements or, alternatively, the HKEx website's page entitled "China Stock Markets Web" (or such other replacement or successor page from time to time) for corporate actions in respect of China Connect Securities. You understand that SSE-listed and/or SZSE-listed issuers (as the case may be) publish corporate documents in Chinese only, and English translations will not be available.

In addition, HKSCC will endeavour to collect and distribute cash dividends relating to China Connect Securities to Clearing Participants participating in China Connect in a timely manner.

Unlike the current practice in Hong Kong in respect of SEHK-listed shares, you understand that you may not be able to attend meetings by proxy or in person.

You understand the Bank does not and cannot ensure the accuracy, reliability or timeliness of any company announcements of corporate actions and accepts no liability (whether in tort or contract or otherwise) for any loss or damage arising from any errors, inaccuracies, delays or omissions or any actions taken in reliance thereon. You understand the Bank has expressly disclaimed all warranties, expressed or implied, as to the accuracy of any company announcement or as to the fitness of the information for any purpose.

32. Disclosure of Information and Publication of Trade Information

You understand that SEHK may require the Bank to provide information on your identity, profile, and the type and value of your orders in relation to Northbound trading of China Connect Securities and the trades which the Bank executed for you at such intervals and in such form as SEHK may specify from time to time for purposes of their publication, dissemination or public distribution of aggregated information in respect of China Connect Securities trades under China Connect, trading volumes, investor profiles and other related data. SEHK may forward such information to the relevant China Connect Market for surveillance and investigation purposes.

33. Client Error

You understand that the Bank will not be liable for any loss, damage or expense or consequential loss, damage or expense suffered by you as a result of any trading based on your instructions. The Bank will not be able to unwind any trade, and you should also take note of the settlement arrangements in respect of China Connect Securities under China Connect, including but not limited to quota restrictions which may affect your ability to mitigate the consequences of any error trades. The Bank shall have no liability for any losses which may result directly or indirectly from such error trades.

The Bank shall have absolute discretion to determine whether to conduct any transfer to rectify any error trade and shall have no obligation to do so. The Bank shall have no liability for any losses which may result directly or indirectly from such error trades or any refusal to conduct a transfer to correct an error trade.

34. Retention of Information

You acknowledge and accept that the Bank will be required under the China Connect Rules to keep records in relation to Northbound trading for a period of not less than 20 years.

35. China Connect Market System

The China Connect Market System is a platform for trading of China Connect Securities under China Connect. The Bank is not responsible for any delay or failure caused by any China Connect Market System and you accept all risks arising from trading China Connect Securities through any China Connect Market System. You understand that the Bank shall not be responsible or held liable for any loss or damage directly or indirectly suffered by you arising from or in connection with the China Connect Service or the CSC through Northbound trading including, without limitation, the following:

(a) a suspension, restriction or cessation of the China Connect Service or the CSC, or any inability to access or use the CSC or the China Connect Service;

- (b) any special arrangement put in place or any action, step or measure taken or not taken to deal with an emergency, including but not limited to the cancellation of any or all China Connect orders input by Exchange Participants;
- (c) any suspension, delay, interruption or cessation of trading of any China Connect Securities;
- (d) any delay, suspension, interruption of trading or of any order cancellation in respect of China Connect Securities as a result of the hoisting of a Typhoon Signal No. 8 or above or the issuance of the Black Rainstorm Warning in Hong Kong;
- (e) any delay or failure to route any China Connect orders, any delay or failure to send any order cancellation requests or to provide the China Connect Service due to any system, communication or connection failure, power outage, software or hardware malfunction or events beyond the control of the Bank or SEHK:
- (f) in the event that a China Connect order which a China Connect Exchange Participant has requested to be cancelled is not cancelled for any reason whatsoever;
- (g) in the event that SEHK or the relevant China Connect Market requires that the Bank reject any order for China Connect Services;
- (h) any delay, failure or error of any China Connect Market System or any system upon which the relevant SEHK Subsidiary is reliant in providing the China Connect Service; and
- (i) any delay or failure to execute, or any error in matching or executing any, China Connect order due to reasons beyond the control of the Bank, SEHK, HKEx or the relevant SEHK Subsidiary including, but not limited to, any action or decision taken or made, or not taken or made, by a China Connect Market, any China Connect Authority or a relevant governmental or regulatory body.

If there is any delay or failure to send any order cancellation requests in the circumstances described in paragraph (e) above, you shall, in the event that the order is matched and executed, remain responsible to the Bank for any settlement obligations in respect of such order.

36. Operational Hours

SEHK has absolute discretion to determine from time to time the operational hours of the China Connect Service, and will have absolute discretion to change the operational hours and arrangements of the China Connect Service, whether on a temporary basis or otherwise, at any time and without advance notice.

37. RMB Conversion and RMB Risks

If you belong to certain counterparty types (such as an individual) in certain jurisdictions that are subject to currency conversion restrictions, (as required by the relevant authority from time to time) any conversion of any currency into RMB pursuant to Clause 9 (Settlement and Currency Conversion) of Part A of these China Connect Terms is subject to any applicable daily maximum conversion limit. Settlement of a Northbound buy order may be delayed and/or fail if there is a delay in converting the relevant currency into RMB. Any risk, loss or cost resulting from any such delay or failure of settlement shall be borne by you.

There are also significant restrictions on the remittance of RMB into and out of Mainland China. If the issuer of the RMB securities is not able to remit RMB to Hong Kong or make distributions in RMB due to exchange controls or other restrictions, the issuer may make distributions (including dividends and other payments) in other currencies. You may therefore be exposed to additional foreign exchange risk and liquidity exposures.

The liquidity and trading price of China Connect Securities may be adversely affected by the limited availability of RMB outside Mainland China and the restrictions on the conversion of RMB. These factors may affect the amount of liquidity in RMB and accordingly adversely affect the market demand for China Connect Securities.

38. Other risks associated with investing in China Connect Securities

ChiNext Shares

ChiNext Shares involve a high investment risk. In particular, profitability and other financial requirements for listing on the ChiNext Board are less stringent than the Main Board and the SME Board of the SZSE.

Companies listed on the ChiNext Board may include enterprises in the innovation and technology sector as well as other start-up and/or growth enterprises with smaller operating scale and share capital. Stock prices may also be more susceptible to manipulation due to fewer circulating shares. Accordingly, the ChiNext Shares may be very volatile and illiquid. In addition, current information on such companies may be limited and may not be widely available.

It may be more common and easier for companies listed on the ChiNext Board to be delisted. The ChiNext Shares may become very illiquid after delisting. You acknowledge that you may suffer a total loss of your investment in the event of a delisting.

You acknowledge that you should seek independent professional advice if you are uncertain of or have not understood any aspect of this Section II or the nature and risks involved in holding or disposal of ChiNext Shares.

General Mainland China related risk

Mainland China is an emerging market that possesses one or more of the following characteristics: a certain degree of political instability, relatively unpredictable financial markets and economic growth patterns, a financial market that is still at the development stage or a weak economy. Emerging markets investments usually result in higher risks such as event risk, political risk, economic risk, credit risk, currency rate risk, market risk, liquidity/gapping risk, regulatory/legal risk, trade settlement, processing and clearing risks and bondholder/shareholder risk.

Equity risk

Investing in China Connect Securities may offer a higher rate of return than investing in short term and longer term debt securities. However, you understand that the risks associated with investments in China Connect Securities may also be higher, because the investment performance of China Connect Securities depends upon factors which are difficult to predict. Such factors include the possibility of sudden or prolonged market declines and risks associated with individual companies. The fundamental risk associated with any equity portfolio is the risk that the value of the investments it holds might suddenly and substantially decrease.

General legal and regulatory risk

You will comply with all China Connect Laws and China Connect Rules. Furthermore, you understand that any change in any China Connect Laws or China Connect Rules may have an impact on the market sentiment which may in turn affect the performance of China Connect Securities. You understand that it is impossible to predict whether such an impact caused by any such change will be positive or negative for China Connect Securities. You understand that in the worst case scenario, you may lose a material part of your investments in China Connect Securities.

PART J - SPECIFIC TERMS AND CONDITIONS GOVERNING TRANSACTIONS IN DIGITAL TOKENS

1. General

- 1.1 These Digital Token Terms shall apply to the Digital Token Services which we may make available to you at your request.
- 1.2 In the event of any conflict or inconsistency between:
 - (a) these Digital Token Terms and any other provision of this Agreement, these Digital Token Terms shall prevail in respect of your Digital Token Account, the Digital Token Services, and/or any Digital Token Transaction; and/or
 - (b) these Digital Token Terms and any Applicable Agreement in respect of Digital Token Transactions, that Applicable Agreement shall prevail.

1.3 Access Conditions

- (a) You acknowledge that all U.S. Persons are prohibited from using any Digital Token Services provided by the Bank under these Digital Token Terms. The Bank reserves the right to immediately cease the provision of any Digital Token Services if:
 - (i) we have reason to believe that you are a U.S. Person;
 - (ii) you and/or any of your Authorised Agents are found to be accessing your Digital Token Account and/or the Digital Token Services from the United States;
 - (iii) you are found to have given false representations as to your location, place of incorporation or establishment, citizenship or place of residence, or
 - (iv) we have reason to believe that you are facilitating access to your Digital Token Account and/or the Digital Token Services on behalf of a U.S. Person.
- (b) You shall not, and shall not permit any person acting on your behalf (including your Authorised Agents)
 - (i) access or use your Digital Token Account and/or the Digital Token Services from or within the United States: and/or
 - (ii) take any action to circumvent the restrictions in Clauses 1.3(a) and 1.3(b)(i), above, including using Virtual Private Networks.
- (c) The Bank may, in our absolute discretion, implement controls to restrict access to the Digital Token Account and/or the Digital Token Services without any notice to you. You acknowledge, agree and understand that if you travel to the United States, the Digital Token Services will not be available and your access to the Digital Token Services will be blocked, except in cases of emergency as determined by the Bank, in our absolute discretion. You acknowledge that this may impact the Bank's ability to execute trades on the Trading System for and on your behalf, and your ability to issue Instructions to the Bank, monitor the balance held in your Digital Token Account, or otherwise use the Digital Token Services. The Bank shall not be liable or responsible to you for any Losses arising out of or in connection with loss of use of your Digital Token Account or the Digital Token Services in such circumstances.
- (d) In addition to and without prejudice to the rest of this Clause 1.3 and any other provision of this Agreement, the Bank may, in our absolute discretion, without assigning any reason whatsoever suspend and/or freeze the operations of your Digital Token Accounts and provision of any Digital Token Service at any time and for such duration as the Bank deems fit for any reason whatsoever. The Bank shall not be liable or responsible to you for any Losses arising out of or in connection with loss of use of your Digital Token Account or the Digital Token Services in such circumstances.

1.4 Instructions

In addition to and without prejudice to any other provision in this Agreement, the Bank may in our absolute discretion and without assigning any reason whatsoever, refuse to act on any Instruction or Order.

The dealing team receiving trading instructions may accept, without verification, all such instructions and shall be under no duty to inquire into the authenticity of any instructions or the identity or authority of the person giving or purporting to give such instructions.

When giving instructions, you shall clearly identify the Account through which the Transaction is to be undertaken, by specifying the name of the account holder and the account number.

You acknowledge that instructions given through the dealing team should not be regarded as having been executed unless and until you receive confirmation from the Bank of the same.

Information relating to the Account would be set out in official statements from the Bank, and any other information provided upon your request may not be accurate or up-to-date as transactions may have been undertaken by you through your relationship manager, investment counsellor or other means without the relevant entry being made in time; accordingly, the data or information provided is not binding except where it is expressly stated as such.

1.5 Your Undertakings, Representations and Warranties

- (a) In addition to and without prejudice to any undertaking, representations and warranties you provide under this Agreement, you hereby undertake, represent and warrant to us that:
 - (i) any Digital Tokens which you deliver or cause to be delivered to the Bank are not related to any activity in contravention of any Applicable Laws, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions;
 - (ii) all payments received or made and Digital Tokens deposited, traded or withdrawn do not represent and/or relate to, or facilitate, the retention or control of proceeds, property, funds or investments derived from or used in connection with fraud, money laundering, terrorism financing and/or sanctioned activities;
 - (iii) you will not use the Digital Token Services, issue or cause to be issued any Instructions, or use or carry out any activities in relation to the Digital Tokens or your Digital Token Account, in a manner that could cause or result in a violation by the Bank or any other entity in the DBS Group of any Applicable Laws, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions;
 - (iv) in entering into any Digital Token Transactions, you shall make your own assessment based on your financial needs and investment objectives, rely on your own judgement in relation to any investment in Digital Tokens and shall not rely on any oral or written representation made by the Bank or any person on its behalf;
 - (v) all information and documents provided to the Bank are true and accurate in all respects;
 - (vi) you will execute all documents and instruments, and will do all acts and things as may be required by the Bank in connection with the provision, opening, operation, maintenance, administration and management of your Digital Token Account and the submission and execution of Digital Token Orders;
 - (vii) you will provide the Bank, in a timely manner, with information and proof as the Bank reasonably requests, as to your tax status or residence. Information and proof may include, as appropriate, executing certificates, making representations and warranties, or providing other information or documents, as the Bank deems necessary or proper to fulfil obligations under Applicable Laws;
 - (viii) you and your Authorised Agents are not U.S. Persons; and

- (ix) you qualify as an accredited investor or institutional investor under the Securities and Futures Act and in addition, where applicable, any class of investor equivalent thereto under the laws or regulations of any other jurisdiction which would allow: (a) you to trade on the Trading System; and (b) Digital Tokens to be offered to or traded with you without giving rise to any obligation on the offeror or trading counterparty to publish or register a prospectus, registration statement or any similar document, or take any other steps or actions in connection with the offer of the Digital Tokens to you.
- (b) Each of the undertakings, warranties and representations set out in this Clause 1.5 shall continue to have full force and effect for so long as this Agreement remain in force and shall be deemed to be repeated by you each time you utilise any Digital Token Service or undertake any Digital Token Transaction.
- (c) You agree and undertake to immediately inform the Bank in writing if any of the undertakings, warranties and representations set out in this Clause 1.5 or in this Agreement become untrue or inaccurate in any respect.

1.6 Extraordinary Event

In addition to and without prejudice to any provision in this Agreement, upon the occurrence of any Extraordinary Event, we may in our absolute discretion determine any adjustment or action necessary in relation to any of your Digital Token Transactions, your Digital Token Account and/or the Digital Token Services. Such adjustments or actions may include altering or varying the quantities of currencies and/or Digital Tokens or the exchange rates or specifications of currencies and/or Digital Tokens bought or sold pursuant to any Digital Token Transaction, terminating any Digital Token Transaction, or making payments, or converting the currency of your Accounts, in or to a currency which is a freely transferable currency at that time, at such rate of exchange as we may, in good faith, deem appropriate, and you agree to be bound by such adjustment or action.

1.7 Fees and Expenses

Without prejudice to this Agreement:

- (a) you agree to pay all fees, commissions and other charges and expenses in connection with your Digital Token Account, the Digital Token Services and the Digital Token Transactions under these Digital Token Terms as specified in our published fees and benefits schedule in force at the relevant time, together with any other amounts payable to the Bank under these Digital Token Terms;
- (b) you acknowledge that whenever a Digital Payment Token transaction is effected, there are costs involved. Digital Payment Token transactions are typically effected over a public blockchain and while different Digital Payment Tokens apply different methodologies to compute the cost of confirming transactions, the general rule of thumb is that a greater network load will increase transaction costs. By entering into any Digital Payment Token transaction, you acknowledge and agree to bear such and all related costs. Due to operational or system constraints, certain charges or expenses incurred by the Bank on your behalf and/or charged to your account (e.g. gas/mining fees) may be rounded up to a decimal place or to an amount that marginally exceeds the actual charges or expenses incurred. The difference is not refundable to you and you authorise the Bank to do the necessary in respect thereof;
- (c) the Bank may debit any of your Accounts to pay any fees, commissions and other charges and expenses imposed or incurred by the Bank (including blockchain transaction fees, charges, costs and expenses (such as mining fees, network fees or gas), as well as taxes, duties or disbursements) in connection with your Digital Token Account, the Digital Token Services and the Digital Token Transactions under these Digital Token Terms, together with any other amounts payable to the Bank under these Digital Token Terms;
- (d) in addition to and without prejudice to our other rights (whether under this Agreement or otherwise), the Bank reserves the right to charge interest on overdue amounts from the due date until actual payment at such rate as the Bank customarily charges for similar overdue amounts; and
- (e) you will maintain or make available sufficient cleared funds in order to cover the Bank's fees, commissions, charges and expenses as aforesaid and to enable us to fulfil our duties under these Digital Token Terms.

Any and all payments to the Bank shall be made in full without any deductions, and exclusive of any goods and services tax or other similar sales taxes. You will be responsible for the payment of any and all such taxes arising.

1.8 Indemnity and Exclusion of Liability

- (a) In addition to and without prejudice to any provision in this Agreement, save for Losses directly arising from our gross negligence, fraud or wilful default, we shall not be responsible or liable to you for any Losses suffered or incurred by you howsoever caused in connection with your Digital Token Account, any of your Digital Token Transactions and/or pursuant to any Digital Token Service provided to you, including any Losses arising from or in connection with:
 - (i) any devaluation in a Digital Token resulting from loss of quality, trade-ability, marketability, or issues pertaining to the integrity or asset trustworthiness of a Digital Token;
 - (ii) volatility or fluctuations in the price or value of Digital Tokens;
 - (iii) any delay, inability or failure to execute Digital Token Orders due to any reason whatsoever, including you having insufficient cleared funds in your Settlement Account or insufficient freely available Digital Tokens in your Digital Token Account, or system failure or other inaccessibility of the Trading System;
 - (iv) any Matched Trade fails to settle for any reason whatsoever, including default by any counterparty member of the Digital Exchange that directly or indirectly causes any Matched Trade (which may or may not be in the same transaction) to fail to settle; and
 - (v) any trade that was made, or deemed by the Digital Exchange to be made, as a result of any error, including error in any term (including price and volume) of the Digital Token Order(s) forming such trade, error due to the operation of the Trading System, and/or error due to a technical and/or system failure or other inaccessibility of the Trading System, the Blockchain or any of their respective component parts ("Error Trade").
- (b) In addition to and without prejudice to the above and any other provision in this Agreement, you agree to indemnify and to hold us, our Personnel, Affiliates and Agents harmless from and against any Losses, save for Losses arising directly from our or their gross negligence, fraud or wilful default, which we or they may suffer or incur in connection with your Digital Token Account, any of your Digital Token Transactions and/or pursuant to any Digital Token Service provided to you.
- (c) You acknowledge, and are fully aware and fully assume, the risks associated with the purchase, sale, transfer, deposit, receipt, keeping and administering of the Digital Tokens, including as disclosed and explained in the General Risk Disclosure Statement and/or any Applicable Agreement.

2. Digital Token Trading Services

2.1 General

- (a) Without derogating from the generality of Clause 1.2, where a Security Token is a unit in a Fund or a Security, the relevant terms applicable to such financial instruments under this Agreement shall apply equally in respect of each Digital Token Transaction involving such Security Tokens.
- (b) In respect of each Digital Token Transaction undertaken on your Instructions, you shall ensure that all Applicable Laws (including any reporting and disclosure requirements and/or shareholding restrictions) are strictly adhered to and complied with at all times and you agree that we need not enquire into or verify any action taken by you in connection therewith. You further warrant and represent that your entry into each Digital Token Transaction does not contravene any Applicable Law, including Applicable Laws relating to insider dealing, market manipulation and/or any other trading offences.
- (c) You shall make payment of all Liabilities in connection with the Digital Token Transactions that we undertake for and on your behalf on or before the due date of payment or such other date as may be notified by us to you. Without prejudice to any provision in this Agreement and these Digital Token Terms, you authorise us to debit any of your Accounts for all such Liabilities as may be payable by you to us or for any payment in connection with any Digital Token Transaction. If payment of any Liability under

or in connection with these Digital Token Terms, including payment for the purchase of any Digital Tokens, is not made by the due date of such payment, we shall have the right (but not the obligation) to force sell, without giving you prior notice or obtaining your prior consent, all or any of such unpaid Digital Tokens at any time and in any manner and on such terms as we think fit and set off the proceeds thereof against all or part of such Liability. If the proceeds thereof are insufficient to set off against all of such Liability, you shall indemnify us for the difference in amount and any interest accrued thereon.

- (d) You acknowledge and agree that:
 - (i) Unless otherwise disclosed to you, we shall be acting as your agent in respect of all Digital Token Transactions. You are therefore principally liable for, and shall assume all liabilities and risks (including any counterparty or settlement risk) associated with, all Digital Token Transactions entered into for and on your behalf pursuant to these Digital Token Terms, notwithstanding that such Digital Token Transactions may have been undertaken in our name without disclosure of such agency.
 - (ii) Subject to any limitation or condition prescribed by Applicable Laws, we and/or our Affiliate may, in respect of any Digital Token Transaction entered into for and on your behalf:
 - (aa) be dealing as principal for our own account in purchasing from or selling to you any Digital Token;
 - (bb) be matching such Digital Token Transaction with that of another of our customers or our Affiliate:
 - (cc) be receiving from any Agent charges, commissions, fees, rebates, discounts or other payments or benefits and you agree that we or our Affiliate (as the case may be) may retain any such payments or benefits and shall be under no obligation to account for or disclose the same to you except to the extent required by Applicable Laws; and
 - (dd) a direct or indirect material interest in any such Digital Token Transaction.
- (e) Without prejudice to any provision in this Agreement and/or these Digital Token Terms, you acknowledge and agree that we may, from time to time, have to accept sole and principal responsibility to any broker, the Digital Exchange or a member of the Digital Exchange (notwithstanding that between you and us, we act as your agent). You further acknowledge and agree that:
 - (i) by reason of the foregoing, any Digital Token purchased by us for and on your behalf may be regarded by the broker, the Digital Exchange or the member as being Digital Tokens purchased by us for our own account and this may result in prejudice to you; and
 - (ii) notwithstanding that such broker, the Digital Exchange and/or a member may look to us to honour and settle any Digital Token Transaction, you shall be principally liable for, and shall assume all liabilities and risks (including any market, counterparty or settlement risk) associated with, all Digital Token Transactions entered into pursuant to these Digital Token Terms for and on your behalf, and you shall indemnify us, in accordance with Clause 1.8, for any Losses we may suffer or incur in connection with such Digital Token Transactions (including any amounts required or deemed required under Applicable Laws, any amount that the Digital Exchange may require us to reimburse the Digital Exchange, or compensate the counterparty member for, or any other penalties that may be imposed by the Digital Exchange in the event a Matched Trade fails to settle, or any other costs or expenses incurred on your behalf).

2.2 Exchange Rate, Transactions and Limits

(a) The exchange rate or purchase price (as denominated in fiat currency) for Digital Tokens is determined by the Digital Exchange and not the Bank. You acknowledge and agree that (i) the exchange rate or purchase price for Digital Tokens may vary from time to time (including within the same Business Day); (ii) the exchange rate or purchase price at which a particular Digital Token Order is executed may not necessarily be the same as that which a separate Digital Token Order by you involving the same Digital Token was executed; and (iii) the exchange rate or purchase price for Digital Tokens acquired through the Digital Exchange may be significantly different from the exchange rates or purchase prices in the general markets or quoted on other exchanges or trading platforms.

(b) The Bank may in our absolute discretion, without giving any reason and without notice to you, at any time and from time to time, impose any limits, including without limitation, position limits and limits on contract size in respect of your Digital Token Account and you shall not exceed such limits. Any limits imposed by the Bank may be reviewed by the Bank from time to time. Where the Digital Exchange imposes limits which are different from the limits imposed by the Bank, you shall be obliged to comply with the limits imposed by the Bank.

2.3 Conduct of Transactions

- (a) The Bank will execute your Digital Token Order only if there is sufficient cash and/or Digital Tokens in your Settlement Account and/or Digital Token Account. In particular:
 - (i) we will submit your "buy" Digital Token Order to the Digital Exchange for execution only if your Settlement Account has sufficient available cleared funds to fund such "buy" Digital Token Order and all applicable charges and expenses for the Digital Token Transaction; and
 - (ii) we will submit your "sell" Digital Token Order to the Digital Exchange for execution only if your Digital Token Account has sufficient freely available Digital Tokens to enable the "sell" Digital Token Order to be settled.
- (b) You acknowledge and agree that:
 - (i) upon submitting a Digital Token Order to the Bank, you are solely responsible for ensuring that the balance of cash and/or Digital Tokens in the Settlement Account and/or Digital Token Account is at all material times sufficient to fund any such Digital Token Order and/or to meet the resulting settlement obligations;
 - (ii) upon matching of a "buy" or "sell" Digital Token Order on the Trading System, a trade is formed and executed which is final and binding on you (a "**Matched Trade**");
 - (iii) all Matched Trades are to be settled in full on the same day ("Digital Token Settlement Date");
 - (iv) in the event a Matched Trade fails to settle for any reasons whatsoever, including default by the counterparty member of the Digital Exchange or if you do not have sufficient freely available Digital Tokens in your Digital Token Account to effect settlement of the Matched Trade, we may, but shall not be obliged to, subject to Applicable Laws, take any action in our absolute discretion to facilitate settlement of the Matched Trade, including (i) any action under Clause 2.1(c) or any other provision of this Agreement, or (ii) purchasing such number of Digital Tokens at such time and on such terms as we think fit, for and on your behalf for the purposes of effecting settlement of such Matched Trade;
 - in the event of an Error Trade, the Bank may take any action as it deems appropriate to comply with Applicable Laws or other direction of the Digital Exchange and/or to resolve the Error Trade; and
 - (vi) you shall be responsible for and indemnify us, in accordance with Clause 1.8, for any Losses that we may suffer or incur in connection with such action taken by us.
- (c) All "buy" and "sell" Digital Token Orders referable to you on the Trading System shall be automatically settled where so designated by the Digital Exchange. All Error Trades designated as such by the Digital Exchange shall be automatically reversed where so designated by the Digital Exchange.
- (d) The investor should be aware of the trading hours of the Digital Exchange and application of the market phases (e.g. "Pre-Open", "Non-Cancel", "Trading" phases and others) which are published by the Exchange and are subject to changes by the Exchange without prior notice.

2.4 Digital Token Orders

- (a) You acknowledge that acceptance by us of your Digital Token Order will not necessarily result in its execution. Without prejudice to the rest of this Clause 2.4, your Digital Token Order may not be executed if, without limitation, (a) market conditions do not permit (e.g. due to volatility in the price or value of Digital Tokens or exchange rates or limited market liquidity), and/or (b) such execution is inconsistent with or contrary to any Applicable Laws (including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions). If we are unable to execute any Digital Token Order, we will inform you as soon as reasonably practicable, whereupon such Digital Token Order shall expire. You acknowledge and agree that we shall not be responsible or liable to you for any Losses you may suffer arising from any delay or omission in connection with the receipt, transmission or execution of any Digital Token Order, except where such Losses are directly caused by our gross negligence, fraud or wilful default.
- (b) You agree that we may, at our absolute discretion:
 - (i) aggregate and consolidate any Digital Token Order received from you in relation to any Digital Token Transaction with similar orders received from other customers before transmitting them to the Digital Exchange or communicating them to any Agent for execution; and
 - (ii) to the extent permitted under Applicable Laws, allot or distribute the Digital Tokens purchased or sold arising from such consolidated orders in any manner as we deem fit. You accept that such allotment or distribution or action by us may result in Losses to you and you accept the risk thereof as being for your account. In the event that the consolidated orders are only partially executed, you agree that we shall be entitled to allocate fewer units of Digital Tokens to you than that you placed an order for.
- (c) Unless otherwise specified, all Digital Token Transactions undertaken on your Digital Token Orders shall be executed in our name. All such Digital Tokens purchased by you shall be held by us (as custodian for you in accordance with these Digital Token Terms).
- (d) After you have submitted any Digital Token Order to the Bank and if such Digital Token Order has been accepted and transmitted to the Digital Exchange, such Digital Token Order may be cancelled only if the cancellation is accepted by the Digital Exchange. Digital Token Orders are irrevocable once a Matched Trade has been formed.
- (e) You may submit your Digital Token Orders to the Bank by such means as may be prescribed by the Bank and communicated to you from time to time. The Bank may, in our absolute discretion, impose requirements for Digital Token Orders, including in relation to minimum order size, as notified to you from time to time. The Bank shall not be under any obligation to accept any Digital Token Order submitted by means other than as prescribed pursuant to this Clause 2.4(e), and unless in compliance with such requirements prescribed pursuant to this Clause 2.4(e).

3. Digital Token Custody Services

3.1 Establishment of Accounts

You authorise the Bank to establish on its books, pursuant to the terms of these Digital Token Terms, a custody account (the "**Digital Token Account**") for the receipt, safekeeping and administration of Digital Tokens.

3.2 Designation of Accounts

- (a) Any Digital Tokens deposited in, and the rights in respect of, the Digital Token Account shall belong exclusively to you legally (other than in the case of Security Tokens where required under Applicable Laws) and beneficially, regardless of whether the blockchain addresses are held in your name, in the name of several customers of the Bank (including you), in the name of the Bank as a custodian for its customers (including you) from time to time, or in the name of an Agent, and shall be entrusted to the Bank for safekeeping as follows:
 - (i) The Digital Token Account will be in your name to:

- (aa) reflect that the Digital Tokens belong to you legally (other than in the case of Security Tokens where required under Applicable Laws) and beneficially, and are merely in the custody of the Bank. The Bank acting as custodian does not constitute us as a trustee (other than a bare trustee in the case of Security Tokens where required under Applicable Laws) and we shall have no trust, fiduciary or other obligations in respect of any Digital Tokens kept in custody with us pursuant to these Digital Token Terms, save for those expressly provided for in this Agreement or as otherwise agreed in writing;
- (bb) reflect that the Digital Tokens do not belong to the Bank, and are received, kept, and administered by the Bank for you; and
- (cc) subject to Clause 3.3 below, to segregate, on the books and records of the Bank, the Digital Tokens held by the Bank for its customers from the Digital Tokens and other cryptocurrencies held by the Bank for our own account and/or the account of our Affiliates.
- (b) You agree and acknowledge that while any Digital Tokens which the Bank has agreed to accept custody of shall remain in the Digital Token Account, the Bank shall be entitled to use any number of blockchain addresses in connection with the receipt, safekeeping, and/or administration of such Digital Tokens, and you shall not be entitled to have any type of access to any such blockchain addresses, regardless of whether the blockchain addresses are held in your name, in the name of several customers of the Bank (including yours), in the name of the Bank as a custodian for its customers (including you), or in the name of an Agent, from time to time.
- (c) You undertake that any Digital Tokens deposited in, and the rights in respect of, the Digital Token Account are legally (other than in the case of Security Tokens where required under Applicable Laws) and beneficially owned by you and are not received, kept and/or administered on trust for any other person.
- (d) Your Digital Token Account and all Digital Tokens are non-interest bearing.

3.3 Segregation of Digital Tokens

- (a) The Bank may determine, in its absolute discretion, how it shall receive, keep and administer the Digital Tokens. The Bank may commingle your Digital Tokens with the Digital Tokens of the Bank's other customers. The Bank may hold your Digital Tokens through an Agent. You acknowledge that your interest in the Digital Tokens may therefore not be identifiable by specific individualised coins, tokens or cryptocurrency unit or specific transaction history, blockchain address or private key or any form of physical documentation or electronic records (including separate certificates, or other physical documents or equivalent electronic records), and the Bank shall maintain records of your interest in the Digital Tokens that have been commingled. Without prejudice to Clause 3.2(a), you further acknowledge that the Bank shall be entitled to use shared blockchain addresses to receive and keep your Digital Tokens with the Digital Tokens and other cryptocurrencies of the Bank's other customers.
- (b) You further agree and acknowledge that the Bank shall be entitled to receive, keep and administer the Digital Tokens on a fungible basis and to re-deliver the Digital Tokens not in specie but of the same denomination. In particular, the Bank does not generally support Bitcoin's "coloured coins", or any other method which presently exists or which may hereafter be developed, to associate certain Digital Payment Tokens or other cryptocurrencies with any particular metadata, and you acknowledge and agree that where any Digital Payment Token is delivered to the Bank, any associated metadata may be lost, and the Bank has no liability for any Losses in relation thereto.

3.4 Bank's Discretion

(a) The Bank reserves the right to at any time and for such duration as the Bank deems fit, in its absolute discretion, without assigning any reason whatsoever, suspend and/or freeze the operations of the Digital Token Account and/or any other Account, or to delay or withhold the crediting of the Digital Token Account and/or any other Account, including in situations where:

- (i) the Bank has grounds to believe, based on information gathered or received by the Bank, that the Digital Token Account and/or other Account, Instructions or Digital Token Transactions is or was conducted for an illegal purpose; and/or
- (ii) the Bank deems necessary for compliance with any Applicable Law, or any policy of the DBS Group, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions.
- (b) Without prejudice to any other provision in this Agreement, the Bank may refuse, in our absolute discretion, without assigning any reason whatsoever, to accept any Digital Payment Tokens or other cryptocurrencies for custody in the Digital Token Account, including:
 - (i) any Digital Payment Tokens or other cryptocurrencies which are not delivered to the Bank in accordance with the terms of these Digital Token Terms or in such form and manner as may be prescribed by the Bank from time to time:
 - (ii) any Digital Payment Token or other cryptocurrencies transferred to a blockchain address controlled by the Bank, without a valid and subsisting Transfer-In Instruction (e.g. you did not or failed to submit a Transfer-In Instruction, the Transfer-In Instruction has lapsed or been withdrawn or cancelled by you, or the Bank has rejected or cancelled the Transfer-In Instruction, or any Digital Payment Tokens or other cryptocurrencies transferred in excess of the amount stipulated in the Transfer-In Instruction);
 - (iii) any Digital Payment Tokens or other cryptocurrencies transferred to a blockchain address controlled by the Bank but cannot be traced to the information provided by you as part of the Transfer-In Instructions:
 - (iv) any Digital Payment Tokens or other cryptocurrencies which are not or cease to be Supported Cryptocurrencies; and/or
 - (v) any Digital Payment Tokens or other cryptocurrencies which the Bank in our absolute discretion determines to be associated with unusual, illegal or illicit activities, circumstances or transactions, or the acceptance of which will or may to cause the Bank to be in breach of any Applicable Laws, or any policy of the DBS Group, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions,

in which event, the Bank reserves the right to, in our absolute discretion, detain, return (to the originating blockchain address after deducting the administrative fees and transaction costs for processing such return (including all transfer fees as incurred)), dispose of or otherwise deal with such Digital Payment Tokens or other cryptocurrencies (including taking any action as may be required for compliance with any Applicable Laws, or any policy of DBS Group, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions).

- (c) Without prejudice to any other provision in this Agreement, the Bank may refuse, in our absolute discretion, without assigning any reason whatsoever, to accept any Security Tokens for custody in the Digital Token Account, and in which event, the Bank reserves the right to, in our absolute discretion, detain, return (by whatever means appropriate in the context, and after deducting the administrative fees and transaction costs for processing such return (including all transfer fees as incurred)), dispose of or otherwise deal with such Security Tokens (including taking any action as may be required for compliance with any Applicable Laws, or any policy of DBS Group, including those pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which may be subjected to sanctions).
- (d) In addition to and without prejudice to any other provision in this Agreement, the Bank may, in our absolute discretion, conduct any transfer in relation to your Digital Token Account or other Account, or add, alter, rectify and/or correct the particulars set out in your Digital Token Account or other Account, including to rectify or reverse any Error Trade or any other error.

3.5 Compliance with Guidelines etc.

You agree to comply with any and all guidelines, notices, rules and policies pertaining to the Digital Tokens and the use and/or access of the Digital Token Services, as well as any amendments to aforementioned, issued by the Bank from time to time. The Bank reserves the right to unilaterally amend, modify, replace and/or update these guidelines, notices, rules and policies from time to time by giving you written notice thereof, specifying a date on which such amendment, modification, replacement and/or update shall take effect.

3.6 Treatment of Instructions to Withdraw

- (a) Upon the Bank's receipt and acceptance of any Instruction to withdraw Digital Tokens from the Digital Token Account, the relevant amount of Digital Tokens will be earmarked and will not available for the purposes of entering into or concluding any Digital Token Transactions.
- (b) You hereby agree and acknowledge that settlement of any Instructions to withdraw Digital Tokens from the Digital Token Account may be delayed. Such delay may be due to, without limitation, the time taken for the Bank or an Agent to process your Instructions, or to perform screening or other checks to ensure compliance with Applicable Laws pertaining to the prevention of fraud, money laundering, terrorism financing and the provision of financial or other services to any person or entity which might be subjected to sanctions.

3.7 Credits to the Digital Token Account

- (a) Subject always to Clause 3.4(a) and Clause 3.4(b) above, the Bank is only obligated to credit the applicable quantity of Digital Tokens to the Digital Token Account:
 - (i) upon receipt and acceptance by the Bank of the relevant Digital Tokens following settlement of the relevant "buy" Digital Token Order, or of the relevant Security Tokens following the issuance and allotment thereof in relation to an initial offering, corporate action or otherwise;
 - (ii) upon receipt and acceptance by the Bank of the relevant Digital Payment Tokens delivered pursuant to a Transfer-In Instruction; or
 - (iii) where a Network Transfer-Out Instruction is submitted, identifying the transferee Account in your name, upon the processing of such Network Transfer-Out Instruction by us.
- (b) For the purposes of Transfer-In Instructions, you acknowledge and agree that the relevant Digital Payment Tokens shall be deemed to be received by the Bank only after:
 - (i) the required number of network confirmations has occurred on the relevant public blockchain (i.e. the confirmation and reconfirmation of a transaction through the creation and addition of blocks to the relevant blockchain), such number as determined by the Bank in its absolute discretion; and
 - (ii) the Bank successfully matches the details of the Digital Token Transaction to the information provided by you as part of the Transfer-In Instructions.

3.8 Debits from the Digital Token Account

- (a) The Bank is authorised by you to debit the applicable quantity of Digital Tokens from the Digital Token Account:
 - (i) upon a Matched Trade being formed and executed pursuant to a relevant "sell" Digital Token Order;
 - (ii) in relation to any Security Token, upon a de-listing, corporate action or otherwise;
 - (iii) upon receipt and acceptance by the Bank of the relevant Transfer-Out Instruction from you; or
 - (iv) upon receipt and acceptance by the Bank of the relevant Network Transfer-Out Instruction from you.

3.9 Transfers of Digital Tokens

- (a) Unless the Bank agrees otherwise, any Security Token which is not acquired through the Bank may not be transferred into your Digital Token Account.
- (b) Transfer-In of Digital Payment Tokens
 - (i) If you wish to transfer any Digital Payment Tokens into your Digital Token Account, you shall submit (or cause to be submitted) to the Bank a transfer-in Instruction ("**Transfer-In Instruction**") by the applicable cut-off time prescribed by the Bank in such form and manner as may be prescribed by the Bank from time to time and together with such supporting documents as the Bank may require. In addition to and without prejudice to any other provision in this Agreement, the Bank may in our absolute discretion and without assigning any reason whatsoever, refuse to act on any Transfer-In Instruction.

The Bank reserves the right to, in our absolute discretion, without assigning any reason whatsoever, reject the Transfer-In Instruction or otherwise decline to provide you with a blockchain address for the deposit of the Digital Payment Tokens underlying the Transfer-In Instruction.

- (ii) Where the Bank provides you with a blockchain address for the deposit of the Digital Payment Tokens underlying the Transfer-In Instruction, you shall within the applicable cut-off time prescribed by the Bank cause:
 - (aa) the exact quantity and type of Digital Payment Tokens underlying such Transfer-In Instruction to be delivered to the relevant blockchain address(es) designated by the Bank; and
 - (bb) such information as may be prescribed by the Bank from time to time to be provided to the Bank,

in such form and manner as may be prescribed by the Bank from time to time.

- (c) Transfer-Out of Digital Payment Tokens
 - (i) If you wish to withdraw any Digital Payment Tokens from your Digital Token Account, you shall submit to the Bank a transfer-out Instruction ("**Transfer-Out Instruction**") by the applicable cut-off time prescribed by the Bank in such form and manner as may be prescribed by the Bank from time to time and together with such supporting documents as the Bank may require.
 - (ii) You undertake and represent that for all sale and Transfer-Out Instructions, you shall have the sufficient quantity of freely available Digital Payment Tokens in your Digital Token Account. The Bank reserves the right to, in our absolute discretion, without assigning any reason whatsoever, reject any sale or Transfer-Out Instruction.
- (d) Transfers to other Digital Token Accounts
 - (i) If you wish to transfer Digital Tokens from a Digital Token Account in your name to a different Digital Token Account (whether in your name or in the name of another client of the Bank), you shall submit to the Bank a transfer-out Instruction ("Network Transfer-Out Instruction") by the applicable cut-off time prescribed by the Bank in such form and manner as may be prescribed by the Bank from time to time and together with such supporting documents as the Bank may require, including stating the reason for the transfer.
 - (ii) You undertake and represent that you shall have the sufficient quantity of freely available Digital Payment Tokens in your Digital Token Account for all sale and transfers (including any Network Transfer-Out). The Bank reserves the right to, in our absolute discretion, without assigning any reason whatsoever, reject the Network Transfer-Out Instruction.
 - (iii) You agree and acknowledge that Network Transfer-Out Instructions shall not be used to settle any bilateral purchase or sale of Digital Tokens between you and another client of the Bank.

(e) Transfers subject to the Bank's discretion

Transfers of Digital Payment Tokens to other payment service providers are subject to the Bank's sole discretion, taking into account various considerations such as the Bank's processes, and may not always be permitted. For example, transfers may not be permitted if it has reason to believe that any of such Digital Payment Tokens is or has become associated with unusual, illegal or illicit activities, circumstances or transactions, or if facilitating such transaction could cause the Bank to be in breach of applicable laws. In such circumstances, the Bank reserves the right to take such action as it seems appropriate, including returning the Digital Payment Tokens to the transferor, disposing of or otherwise dealing with such Digital Payment Tokens as may be required for compliance with applicable laws.

3.10 Supported Digital Tokens

The Bank reserves the right to at any time, in our absolute discretion and without your consent, without assigning any reason whatsoever, remove any Digital Token from the then-current list of Supported Digital Tokens ("Withdrawn Digital Tokens"), and cease to provide the Digital Token Custody Services or any other services (including the Digital Token Trading Services) in relation to such Withdrawn Digital Tokens.

3.11 Forks

- (a) You acknowledge and accept that blockchain networks may be subject to changes in protocol rules resulting in a fork of the relevant network, and any Digital Payment Tokens or other cryptocurrencies newly created as a result of such forks shall not be deemed to be a Supported Cryptocurrency unless the Bank determines otherwise.
- (b) You further acknowledge and agree that the Bank reserves the right to, in its absolute discretion, decide:
 - (i) whether to support the forked network, and the terms and conditions, including the associated costs, upon which the Bank will implement support of such forked network;
 - (ii) which cryptographic token of the branch(es) resulting from such fork to support; and
 - (iii) whether to detain, dispose of or however else deal with any Digital Payment Tokens or other cryptocurrencies newly created in connection with such forks,

and the Bank shall be under no obligation to (aa) notify you of any such forks and/or (bb) account to you, credit your Digital Token Account, or provide any of the Digital Token Services in relation to, any Digital Payment Tokens or other cryptocurrencies newly created in connection with such forks.

RISK WARNING ON DIGITAL PAYMENT TOKEN SERVICES

The Monetary Authority of Singapore (MAS) requires us to provide this risk warning to you as a customer of a digital payment token (DPT) service provider.

Before you:

- (a) pay or transfer your DPT service provider any money or DPT;
- (b) enter into an agreement with a third party to buy or sell DPTs upon inducement by your DPT service provider;
- (c) enter into an agreement with a third party under an arrangement by your DPT service provider for the transmission of DPTs;
- (d) pay a third party any money or DPT under an arrangement by your DPT service provider for the transmission of DPTs; or (e) transfer any digital payment token instrument to your DPT service provider;
- (e) transfer any digital payment token instrument to your DPT service provider,

you should be aware of the following.

- 1. Your DPT service provider is exempted by MAS from holding a licence to provide DPT services. Please note that you may not be able to recover all the money or DPTs you paid to your DPT service provider, or any other third party referred to above, if your DPT service provider's or the third party's business fails.
- 2. You should not transact in the DPT if you are not familiar with this DPT. This includes how the DPT is created, and how the DPT you intend to transact is transferred or held by your DPT service provider.
- 3. You should be aware that the value of DPTs may fluctuate greatly. You should buy DPTs only if you are prepared to accept the risk of losing all of the money you put into such tokens.
- 4. You should be aware that your DPT service provider, as part of its licence to provide DPT services, may offer services related to DPTs which are promoted as having a stable value, commonly known as "stablecoin".

IMPORTANT DISCLOSURES RELATING TO DIGITAL TOKENS

Disclosures Relating to the Digital Exchange

- 1. The Digital Exchange is part of the larger DBS Group, which is one of the largest banking group in Southeast Asia by total assets and is engaged in a range of commercial banking and financial services, principally in Asia.
- 2. The affiliates of the Digital Exchange from the DBS Group may, from time to time, be admitted as members of the Digital Exchange. Such DBS Group entities include DBS Private Bank and DBS Vickers Securities (Singapore) Pte. Ltd. Such DBS Group entities may only participate on the organised market operated by the Digital Exchange on behalf of their clients, and may not execute any trades on their own accounts.
- 3. The Bank, an intermediary member of the Digital Exchange, is able to trade on the Digital Exchange on your behalf because the Bank deals with you as an accredited investor (as defined under the Securities and Futures Act). The Digital Exchange is regulated by the MAS as a Recognised Market Operator under the Securities and Futures Act and is subject to less stringent regulations than approved exchanges.



DEPOSIT INSURANCE SCHEME

Singapore dollar deposits of non-bank depositors and monies and deposits denominated in Singapore dollars under the Supplementary Retirement Scheme are insured by the Singapore Deposit Insurance Corporation, for up to S\$100,000 in aggregate per depositor per Scheme member by law. Monies and deposits denominated in Singapore dollars under the CPF Investment Scheme and CPF Retirement Sum Scheme are aggregated and separately insured up to S\$100,000 for each depositor per Scheme member. Foreign currency deposits, dual currency investments, structured deposits and other investment products are not insured.





